ADMISSIONS AND CONTINUED OCCUPANCY POLICY

FOR THE

PUBLIC HOUSING PROGRAM

May 1, 2005

Revision Date		
February 1, 2006	June 1, 2013	
October 1, 2006	July 1, 2014	
June 1, 2007	June 1, 2015	
September 1, 2008	March 1, 2016	
September 1, 2009	March 1, 2017	
August 1, 2010	July 1, 2017	
June 1, 2011	July 1, 2018	
June 1, 2012		

Approved by the PHA Board of Commissioners:

Submitted to HUD:

Copyright 2018 by Nan McKay & Associates, Inc.
All rights reserved
Permission to reprint granted only to the Public Housing Authority that has purchased this plan from Nan McKay & Associates, Inc. This document may not be reprinted or distributed to any other person or entity other than the purchasing agency without the express written permission of Nan McKay & Associates, Inc.



Nan McKay and Associates Product Order Form

FOR MORE INFORMATION CONTACT US AT: 800.783.3100 sales@nanmckay.com

\$1779

and Your Rights Under Fair Housing on one DVD.

□ 406-120 DVD

UBLICATIONS		POLICIES & PLANS	
FORMS AND TOOLS		MODEL FAMILY SELF-SUFFICIENCY ACTION PLAN	
310-015 UPCS Quick Check Guide	\$30	☐ 308-017 FSS Action Plan	\$4
☐ 310-027 HQS Quick Check Guide	\$30	MODEL ADMISSIONS & CONTINUED OCCUPANCY PO	JUCV
☐ 310-003 PH Occupancy Tracking Tool	\$249	□ 301-001 ACOP Only	\$8
rogram information products and services		309-001 ACOP and Annual Revision Service	
☐ 303-001 PIH Alert	\$349	AAODEL A DAAINIISTDATIVE DLANI	
□ 308-552 Housing Regulations on CD	\$249	MODEL ADMINISTRATIVE PLAN	\$8
309-552 Housing Regulations CD and Annual Revision	Service \$448	☐ 301-002 Model Admin Plan Only ☐ 309-002 Model Admin Plan and Annual Rev	
OURSE BOOKS \$129 EACH		a so / soz //tode/ / tallilli Hall dila / tillibar key	ν,
□ 040-800 HCV HUD Reference Coursebook		MODEL EIV USE and SECURITY POLICIES	
040-900 Public Housing HUD Reference Courseb	ook	□ 308-018	\$3
□ 040-720 HUD References Rent Calculation	1	MODEL TENIANT CELECTION DIANI LO : I	
f you have attended an NMA training in the last two years and need to p placement materials, please email sales@nanmckay.com.	ourchase	MODEL TENANT SELECTION PLAN and Guide	.
,		301-016 Model TSP Only	
ASTER BOOKS		309-016 Model TSP and Annual Revision Se	•
ave 10% when you order the Master Book and Annual Revision Service	e together.	*If you would like to purchase the paper copy of the Model Administr Admissions & Continued Occupancy Policy, please email sales@nann	
AMILY SELF-SUFFICIENCY		HANDBOOKS A	sk about quantity disco
□ 308-003 CD	\$229		sk about quality also
☐ 309-003 CD and Annual Revision Service	\$428 \$385.20	FORMS AND TOOLS \$20 EACH	
JBLIC HOUSING	ψ003.20	☐ 304-001 Essentials for Commissioners ☐ 304-002 Assisted Housing 101	
□ 308-004 CD	\$229	□ 304-010 Customer Service for PHAs	
☐ 309-004 CD and Annual Revision Service	\$428	☐ 304-003 Public Housing Family Handbook (E	in aliah)
OJECT-BASED ACCOUNTING FOR PHAS	\$385.20	304-003 Public Housing Family Handbook (S	
□ 308-005 CD	\$229	□ 304-005 HCV Family Handbook (English)	
☐ 309-005 CD and Annual Revision Service	\$428	304-006 HCV Family Handbook (English)	
has public housing assessment system	\$385.20	☐ 304-007 Owner's Handbook	
□ 308-012 CD	\$229		
UBLIC HOUSING MANAGEMENT		HANDROOK OLIANITIV DISCOUR	JTC
□ 308-015 CD	\$229	HANDBOOK QUANTITY DISCOUN	
309-015 CD and Annual Revision Service	\$428	1-99\$20 each 100-199	·
	\$385.20	200+\$7.50 each 1000+frr Quantity discounts cannot be combined with coupons, other o	J
NIFORM PHYSICAL CONDITION STANDARDS	4000	Quantity discounts cannot be combined with coupons, other o	itters, or discounts.
☐ 308-006 CD ☐ 309-006 CD and Annual Revision Service	\$229 \$428		
	\$385.20	VIDEO & DVD	
OUSING CHOICE VOUCHER	****		
☐ 308-007 CD ☐ 309-007 CD and Annual Revision Service	\$229 \$428	HOUSEKEEPING BASICS	
	\$385.20	407 100 BYB (F. 1.1)	\$249
CV FINANCIAL MANAGEMENT & REPORTING		406-109 DVD (English) 406-159 DVD (Spanish)	\$249 \$249
308-008 CD	\$229	_	,,
☐ 309-008 CD and Annual Revision Service	\$428 \$385.20	YOUR RIGHTS UNDER FAIR HOUSING	A 150
CV EXECUTIVE MANAGEMENT		☐ 406-118 DVD	\$450
□ 308-010 CD	\$229	PUBLIC HOUSING ORIENTATION	
☐ 309-010 CD and Annual Revision Service	\$428 \$385.20	□ 406-103 DVD	\$520
CV HOUSING QUALITY STANDARDS	,	VOUCHER FAMILY BRIEFING	
□ 308-009 CD	\$229	☐ 406-102 DVD (English)	\$620
☐ 309-009 CD and Annual Revision Service	\$428	☐ 406-152 DVD (Spanish)	\$620
you've taken a seminar and would like to purchase the Annual Revision S	\$385.20 ervice only, please	THE VIDEO BRIEFING SERIES DVD	
ntact sales@nanmckay.com		This series includes Public Housing Orientation, Voucher Family Briefin	11 1 · D ·

Network Licenses are available for all applicable Master Books on CD and their revisions. Please call for availability. Prices are subject to change without notice. Please see next page for quantity discounts. Quantity discounts cannot be combined with coupons, other special offers or discounts.

*All Master Books and Revision Services on CD are comprised of searchable PDF files. If you are interested in purchasing the paper copy of a Master Book, please email sales@nanmckay.com.



Nan McKay and Associates Product Order Form

FOR MORE INFORMATION CONTACT US AT: 800.783.3100 sales@nanmckay.com

DIRECTIONS:

(example) 100-001

Please enter the products you would like to order below, selecting from the options on page 1. List the product(s) by 6-digit product number, price, and quantity.

- $\ensuremath{\text{1}}.$ Calculate the subtotal by adding all the product totals
- 2. Apply discounts if applicable
- 3. Add sales tax if you live in California or New Mexico

PRODUCT # PRICE QUANTITY TOTAL

\$299.00 2 \$598.00

4. Calculate shipping and handling cost.

CALCULATE YOUR PURCHASE TOTAL SUBTOTAL 1. DISCOUNT 2. *SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each \$8.50 Master Books on CD Each \$8.50 Master Books on Paper Each \$17.25 Revision Services on Paper Each \$14.00 Policy / Plan Each \$17.25 Workbooks / Course Books Each \$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10 \$8.50 H1 - 25 \$14.00 26 - 50 \$24.00 51 - 75 \$36.00	
*SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
*SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
SUBTOTAL 1. DISCOUNT 2. *SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
*SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
*SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
SUBTOTAL 1. DISCOUNT 2. *SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
*SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on Paper Each\$17.25 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	CALCULATE YOUR PURCHASE TOTAL
*SALES TAX 3. *California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on Paper Each\$17.25 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	SUBTOTAL 1.
*California and New Mexico SHIPPING & HANDLING 4. PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$14.00 Policy / Plan Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	DISCOUNT 2.
## PURCHASE TOTAL 5. HANDBOOK QTY DISCOUNTS	
HANDBOOK QTY DISCOUNTS 1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each \$8.50 Master Books on CD Each \$8.50 Master Books on Paper Each \$17.25 Revision Services on CD Each \$8.50 Revision Services on Paper Each \$14.00 Policy / Plan Each \$17.25 Workbooks / Course Books Each \$14.00 Handbooks / Quick Check Guides By Quantity \$1 - 10 \$8.50 11 - 25 \$14.00 26 - 50 \$24.00 51 - 75 \$36.00	SHIPPING & HANDLING 4.
1-99\$20 each 100-199\$9.50 each 200+\$7.50 each 1000+free logo added! SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each \$8.50 Master Books on CD Each \$8.50 Master Books on Paper Each \$17.25 Revision Services on CD Each \$8.50 Revision Services on Paper Each \$14.00 Policy / Plan Each \$17.25 Workbooks / Course Books Each \$14.00 Handbooks / Quick Check Guides By Quantity \$1-10 \$8.50 11 - 25 \$14.00 26 - 50 \$24.00 51 - 75 \$36.00	PURCHASE TOTAL 5.
SHIPPING & HANDLING 3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each \$8.50 Master Books on CD Each \$8.50 Master Books on Paper Each \$17.25 Revision Services on CD Each \$8.50 Revision Services on Paper Each \$17.25 Workbooks / Course Books Each \$14.00 Handbooks / Quick Check Guides By Quantity \$1.10 \$1.1 - 25 \$14.00 \$26 - 50 \$24.00 \$1 - 75 \$36.00	HANDBOOK QTY DISCOUNTS
3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	
3 days turnaround time, plus about 5 days for ground shipping. Overnight shipping is also available; please contact sales@nanmckay.com for pricing. CD / DVD Each\$8.50 Master Books on CD Each\$8.50 Master Books on Paper Each\$17.25 Revision Services on CD Each\$8.50 Revision Services on Paper Each\$17.25 Workbooks / Course Books Each\$14.00 Handbooks / Quick Check Guides By Quantity 1 - 10\$8.50 11 - 25\$14.00 26 - 50\$24.00 51 - 75\$36.00	SHIPPING & HANDLING
76 - 100\$47.00 101+Add \$8.25 per 25 handbooks ordered.	

Please include a shipping address that	is not a P.O. Box.	
full NAME:		
TITLE:		
AGENCY:		
EMAIL:		
PHONE:	_ FAX:	
BILLING ADDRESS:		
CITY:	STATE:	_ ZIP:
SHIPPING ADDRESS (NO P.O. BOXES):		
CITY:	_ STATE:	_ ZIP:
PAYMENT OPTIONS		
To order using a credit card or purchas 619.258.5791. Make checks payable order form(s) with your check to 1810 Payment must be received before order following payment options.	e to Nan McKay and Gillespie Way, Suite can be completed.	I Associates, Inc. and mail the 202, El Cajon, CA 92020. Please select one of the
ORDER TOTAL:		
PURCHASE ORDER NUMBER:		
CHECK PAYABLE TO NMA ENCLOSE	D: Check #	:
CHARGE MY CREDIT CARD: VISA	Master Card	American Express
NAME ON CREDIT CARD:		
CREDIT CARD #:		
SECURITY CODE:	EXPIRATION DATE	:/
TODAY'S DATE: / / / /		
AUTHORIZED SIGNATURE (Please sig	yn within the bounda	ries of the box below)
*QUANTITY DISCOUNTS Handbooks. NMA offers quantity discounts on	handbooks, Mixina and	matchina of titles is allowed for orders

Handbooks. NMA offers quantity discounts on handbooks. Mixing and matching of titles is allowed for orders totaling 200 or more books. Customized covers with HA name & logo available with 1000 or more. Please call for pricing and customization details. Quantity discounts cannot be combined with coupons, other offers, or discounts.

POLICIES

Network Licenses are available for all applicable Master Books on CD and their revisions. Please call for availability. Electronic media purchases may not be returned.

© Nan McKay & Associates, Inc. All rights reserved. For domestic use only. Unauthorized copying, reproduction, hiring, lending, public performance and broadcasting prohibited.

NMA reserves the right to change any policies or pricing at any time and in its sole discretion.

Introduction ABOUT THE MODEL ACOP

HOW	TO USE THE MODEL ACOP	Intro-i
ABOU	JT THE MODEL ACOP AND THE PUBLIC HOUSING LEASE	Intro-ii
REFE	RENCES CITED IN THE MODEL ACOP	Intro-ii
	URCES CITED IN THE MODEL ACOP	
	Chapter 1	
	OVERVIEW OF THE PROGRAM AND PLAN	
INTR	ODUCTION	1-1
11 (111		
	PART I: THE PHA	
I.A.	OVERVIEW	1-1
I.B.	ORGANIZATION AND STRUCTURE OF THE PHA	1-2
I.C.	PHA MISSION	1-3
I.D.	THE PHA'S COMMITMENT TO ETHICS AND SERVICE	1-4
	PART II: THE PUBLIC HOUSING PROGRAM	
II.A.	OVERVIEW AND HISTORY OF THE PROGRAM	1-5
II.B.	PUBLIC HOUSING PROGRAM BASICS	
II.C.	PUBLIC HOUSING PARTNERSHIPS	1-6
II.D.	APPLICABLE REGULATIONS	1-10
	PART III: THE ADMISSIONS AND CONTINUED OCCUPANCY P	OLICIES
III.A.	OVERVIEW AND PURPOSE OF THE POLICY	1-11
III.B.	CONTENTS OF THE POLICY	1-11
III.C.	UPDATING AND REVISING THE POLICY	1-12

Chapter 2 FAIR HOUSING AND EQUAL OPPORTUNITY

INTRO	ODUCTION	2-1
	PART I: NONDISCRIMINATION	
I.A.	OVERVIEW	2-3
I.B.	NONDISCRIMINATION	
	PART II: POLICIES RELATED TO PERSONS WITH DISABILITIES	
II.A.	OVERVIEW	2-7
II.B.	DEFINITION OF REASONABLE ACCOMMODATION	
II.C.	REQUEST FOR AN ACCOMMODATION	2-9
II.D.	VERIFICATION OF DISABILITY	2-10
II.E.	APPROVAL/DENIAL OF A REQUESTED ACCOMMODATION	2-11
II.F.	PROGRAM ACCESSIBILITY FOR PERSONS WITH HEARING OR	
	VISION IMPAIRMENTS	2-12
II.G.	PHYSICAL ACCESSIBILITY	2-13
II.H.	DENIAL OR TERMINATION OF ASSISTANCE	2-14
	PART III: IMPROVING ACCESS TO SERVICES FOR PERSONS WITH LIMITED ENGLISH PROFICIENCY (LEP)	
ПΙΔ	OVERVIEW	2-15
	ORAL INTERPRETATION	
	WRITTEN TRANSLATION	
	IMPLEMENTATION PLAN	
	EXHIBITS	
2-1:	DEFINITION OF A PERSON WITH A DISABILITY UNDER FEDERAL CIVIL RIGHTS LAWS	2-19

Chapter 3 ELIGIBILITY

INTR	ODUCTION	3-1
	PART I: DEFINITIONS OF FAMILY AND HOUSEHOLD MEMBERS	
I.A.	OVERVIEW	3-3
I.B.	FAMILY AND HOUSEHOLD	3-3
I.C.	FAMILY BREAK-UP AND REMAINING MEMBER OF TENANT FAMILY	3-4
I.D.	HEAD OF HOUSEHOLD	3-5
I.E.	SPOUSE, COHEAD, AND OTHER ADULT	3-5
I.F.	DEPENDENT	
I.G.	FULL-TIME STUDENT	
I.H.	ELDERLY AND NEAR-ELDERLY PERSONS, AND ELDERLY FAMILY	
I.I.	PERSONS WITH DISABILITIES AND DISABLED FAMILY	3-7
I.J.	GUESTS	
I.K.	FOSTER CHILDREN AND FOSTER ADULTS	
I.L.	ABSENT FAMILY MEMBERS	
I.M.	LIVE-IN AIDE	3-12
	PART II: BASIC ELIGIBILITY CRITERIA	
II.A.	INCOME ELIGIBILITY AND TARGETING	3-13
II.B.	CITIZENSHIP OR ELIGIBLE IMMIGRATION STATUS	3-15
II.C.	SOCIAL SECURITY NUMBERS	
II.D.	FAMILY CONSENT TO RELEASE OF INFORMATION	3-18
	PART III: DENIAL OF ADMISSION	
III.A.	OVERVIEW	3-19
III.B.	REQUIRED DENIAL OF ADMISSION	
III.C.	OTHER PERMITTED REASONS FOR DENIAL OF ADMISSION	3-21
III.D.	SCREENING	
III.E.	CRITERIA FOR DECIDING TO DENY ADMISSION	3-28
III.F.	PROHIBITION AGAINST DENIAL OF ASSISTANCE TO VICTIMS	
	OF DOMESTIC VIOLENCE, DATING VIOLENCE, SEXUAL ASSAULT,	
	OR STALKING	
III.G.	NOTICE OF ELIGIBILITY OR DENIAL	3-32
	EXHIBITS	
3-1:	DETAILED DEFINITIONS RELATED TO DISABILITIES	3-33

Chapter 4 APPLICATIONS, WAITING LIST AND TENANT SELECTION

INTRO	ODUCTION	4-1
	PART I: THE APPLICATION PROCESS	
I.A.	OVERVIEW	4-3
I.B.	APPLYING FOR ASSISTANCE	
I.C.	ACCESSIBILITY OF THE APPLICATION PROCESS	
I.D.	PLACEMENT ON THE WAITING LIST	
	PART II: MANAGING THE WAITING LIST	
II.A.	OVERVIEW	1-7
II.B.	ORGANIZATION OF THE WAITING LIST	
II.C.	OPENING AND CLOSING THE WAITING LIST	
II.D.	FAMILY OUTREACH	
II.E.	REPORTING CHANGES IN FAMILY CIRCUMSTANCES	
II.F.	UPDATING THE WAITING LIST	
	PART III: TENANT SELECTION	
III A	OVERVIEW	4-15
III.B.		
III.C.		
III.D.	THE APPLICATION INTERVIEW	
	FINAL ELIGIBILITY DETERMINATION	
	Chapter 5	
	OCCUPANCY STANDARDS AND UNIT OFFERS	
INTRO	ODUCTION	5-1
	PART I: OCCUPANCY STANDARDS	
I.A.	OVERVIEW	5-1
I.B.	DETERMINING UNIT SIZE	
I.C.	EXCEPTIONS TO OCCUPANCY STANDARDS	5-4
	PART II: UNIT OFFERS	
II.A.	OVERVIEW	5-5
II.B.	NUMBER OF OFFERS	5-5
II.C.	TIME LIMIT FOR UNIT OFFER ACCEPTANCE OR REFUSAL	5-6
II.D.	REFUSALS OF UNIT OFFERS	5-6
II.E.	ACCESSIBLE UNITS	5-8
II.F.	DESIGNATED HOUSING	5-8

Chapter 6 INCOME AND RENT DETERMINATIONS

INTRO	ODUCTION	6-1
	PART I: ANNUAL INCOME	
I.A.	OVERVIEW	6-3
I.B.	HOUSEHOLD COMPOSITION AND INCOME	
I.C.	ANTICIPATING ANNUAL INCOME	
I.D.	EARNED INCOME	6-9
I.E.	EARNED INCOME DISALLOWANCE	6-13
I.F.	BUSINESS INCOME	6-17
I.G.	ASSETS	
I.H.	PERIODIC PAYMENTS	
I.I.	PAYMENTS IN LIEU OF EARNINGS	6-29
I.J.	WELFARE ASSISTANCE	
I.K.	PERIODIC AND DETERMINABLE ALLOWANCES	
I.L.	ADDITIONAL EXCLUSIONS FROM ANNUAL INCOME	6-32
	PART II: ADJUSTED INCOME	
II.A.	INTRODUCTION	
II.B.	DEPENDENT DEDUCTION	
II.C.	ELDERLY OR DISABLED FAMILY DEDUCTION	
II.D.	MEDICAL EXPENSES DEDUCTION DISABILITY ASSISTANCE EXPENSES DEDUCTION	
II.E.	CHILD CARE EXPENSE DEDUCTION	
II.F. II.G.	PERMISSIVE DEDUCTIONS	
II.G.	PERMISSIVE DEDUCTIONS	0-40
	PART III: CALCULATING RENT	
III.A.	OVERVIEW OF INCOME-BASED RENT CALCULATIONS	6-47
	FINANCIAL HARDSHIPS AFFECTING MINIMUM RENT	
III.C.	UTILITY ALLOWANCES	
III.D.	PRORATED RENT FOR MIXED FAMILIES	6-56
III.E.		
	EXHIBITS	
6-1:	ANNUAL INCOME INCLUSIONS	
6-2:	ANNUAL INCOME EXCLUSIONS	
6-3:	TREATMENT OF FAMILY ASSETS	
6-4:	EARNED INCOME DISALLOWANCE	
6-5:	THE EFFECT OF WELFARE BENEFIT REDUCTION	6-69

Chapter 7 VERIFICATION

INTRO	ODUCTION	7-1
	PART I: GENERAL VERIFICATION REQUIREMENTS	
I.A.	FAMILY CONSENT TO RELEASE OF INFORMATION	7-1
I.B.	OVERVIEW OF VERIFICATION REQUIREMENTS	
I.C.	UP-FRONT INCOME VERIFICATION (UIV)	
I.D.	THIRD-PARTY WRITTEN AND ORAL VERIFICATION	7-6
I.E.	SELF-CERTIFICATION	
	PART II: VERIFYING FAMILY INFORMATION	
II.A.	VERIFICATION OF LEGAL IDENTITY	7-11
II.B.	SOCIAL SECURITY NUMBERS	7-12
II.C.	DOCUMENTATION OF AGE	7-13
II.D.	FAMILY RELATIONSHIPS	7-14
II.E.	VERIFICATION OF STUDENT STATUS	7-15
II.F.	DOCUMENTATION OF DISABILITY	7-16
II.G.	CITIZENSHIP OR ELIGIBLE IMMIGRATION STATUS	7-17
II.H.	VERIFICATION OF PREFERENCE STATUS	7-18
	PART III: VERIFYING INCOME AND ASSETS	
III.A.	EARNED INCOME	7-19
III.B.	BUSINESS AND SELF EMPLOYMENT INCOME	7-19
III.C.	PERIODIC PAYMENTS AND PAYMENTS IN LIEU OF EARNINGS	
III.D.	ALIMONY OR CHILD SUPPORT	
III.E.	ASSETS AND INCOME FROM ASSETS	7-22
III.F.	NET INCOME FROM RENTAL PROPERTY	
III.G.	RETIREMENT ACCOUNTS	7-23
III.H.	INCOME FROM EXCLUDED SOURCES	7-24
III.I.	ZERO ANNUAL INCOME STATUS	7-24
	PART IV: VERIFYING MANDATORY DEDUCTIONS	
IV.A.	DEPENDENT AND ELDERLY/DISABLED HOUSEHOLD DEDUCTIONS	7-25
IV.B.	MEDICAL EXPENSE DEDUCTION	7-26
	DISABILITY ASSISTANCE EXPENSES	
IV.D.	CHILD CARE EXPENSES	7-30
	EXHIBITS	
7-1:	SUMMARY OF DOCUMENTATION REQUIREMENTS	
	FOR NONCITIZENS	7-33

Chapter 8 LEASING AND INSPECTIONS

INTR	RODUCTION	8-1
	PART I: LEASING	
I.A.	OVERVIEW	8-1
I.B.	LEASE ORIENTATION	8-2
I.C.	EXECUTION OF LEASE	8-3
I.D.	MODIFICATIONS TO THE LEASE	8-4
I.E.	SECURITY DEPOSITS	
I.F.	PAYMENTS UNDER THE LEASE	8-7
	PART II: INSPECTIONS	
II.A.	OVERVIEW	8-11
II.B.	TYPES OF INSPECTIONS	8-11
II.C.		
II.D.	INSPECTION RESULTS	8-14
	EXHIBITS	
8-1:	SMOKE-FREE POLICY	8-17

Chapter 9 REEXAMINATIONS

INTR	ODUCTION	9-1
	PART I: ANNUAL REEXAMINATIONS FOR FAMILIES PAYING INCOME-BASED RENTS	
I.A.	OVERVIEW	9-3
I.B.	STREAMLINED ANNUAL REEXAMINATIONS	9-4
I.C.	SCHEDULING ANNUAL REEXAMINATIONS	9-5
I.D.	CONDUCTING ANNUAL REEXAMINATIONS	9-6
I.E.	EFFECTIVE DATES	9-8
	PART II: REEXAMINATIONS FOR FAMILIES PAYING FLAT RENTS	
II.A.	OVERVIEW	9-9
II.B.	FULL REEXAMINATION OF FAMILY INCOME AND COMPOSITION	9-9
II.C.	REEXAMINATION OF FAMILY COMPOSITION ("ANNUAL UPDATE")	9-10
	PART III: INTERIM REEXAMINATIONS	
III.A.	OVERVIEW	9-13
	CHANGES IN FAMILY AND HOUSEHOLD COMPOSITION	
III.C.	CHANGES AFFECTING INCOME OR EXPENSES	9-15
III.D.	PROCESSING THE INTERIM REEXAMINATION	9-17
	PART IV: RECALCULATING TENANT RENT	
	OVERVIEW	
IV.B.	CHANGES IN UTILITY ALLOWANCES	9-19
IV.C.	NOTIFICATION OF NEW TENANT RENT	9-19
IV.D.	DISCREPANCIES	9-20

Chapter 10 PETS

INTR	ODUCTION	10-1
	PART I: SERVICE ANIMALS AND ASSISTANCE ANIM	MALS
I.A.	OVERVIEW	10-3
I.B.	APPROVAL OF ASSISTANCE ANIMALS	
I.C.	CARE AND HANDLING	
	PART II: PET POLICIES FOR ALL DEVELOPMEN	TS
II.A.	OVERVIEW	10-7
II.B.	MANAGEMENT APPROVAL OF PETS	
II.C.	STANDARDS FOR PETS	10-9
II.D.	PET RULES	10-11
	PART III: PET DEPOSITS AND FEES IN	
	ELDERLY/DISABLED DEVELOPMENTS	
III.A.	OVERVIEW	10-17
	PET DEPOSITS	
III.C.	OTHER CHARGES	10-18
	PART IV: PET DEPOSITS AND FEES IN GENERA	L
	OCCUPANCY DEVELOPMENTS	
IV.A.	OVERVIEW	10-19
IV.B.	PET DEPOSITS	10-19
IV.C.	NON-REFUNDABLE NOMINAL PET FEE	10-20
IV.D.	OTHER CHARGES	10-21

Chapter 11 COMMUNITY SERVICE

INTR	ODUCTION	11-1
	PART I: COMMUNITY SERVICE REQUIREMENT	
I.A.	OVERVIEW	11-1
I.B.	REQUIREMENTS	11-2
I.C.	DETERMINATION OF EXEMPTION STATUS AND COMPLIANCE	11-7
I.D.	DOCUMENTATION AND VERIFICATION	11-11
I.E.	NONCOMPLIANCE	11-13
II.A.	PART II: IMPLEMENTATION OF COMMUNITY SERVICE OVERVIEW	11-17
	EXHIBITS	
11-1:	COMMUNITY SERVICE AND SELF-SUFFICIENCY POLICY	11-19
11-2:	DEFINITION OF A PERSON WITH A DISABILITY UNDER SOCIAL	
	SECURITY ACTS 216(i)(l) and Section 1416(excerpt) FOR PURPOSES	
	OF EXEMPTION FROM COMMUNITY SERVICE	11-25
11-3:	PHA DETERMINATION OF EXEMPTION FOR COMMUNITY SERVICE	CE 11-27
11-4	CSSR WORK-OUT AGREEMENT	11-29

Chapter 12 TRANSFER POLICY

INTR	ODUCTION	12-1
	PART I: EMERGENCY TRANSFERS	
I.A.	OVERVIEW	12-1
I.B.	EMERGENCY TRANSFERS	12-2
I.C.	EMERGENCY TRANSFER PROCEDURES	
I.D.	COSTS OF TRANSFER	12-3
	PART II: PHA REQUIRED TRANSFERS	
II.A.	OVERVIEW	12-5
II.B.	TYPES OF PHA REQUIRED TRANSFERS	12-5
II.C.	ADVERSE ACTION	12-8
II.D.	COST OF TRANSFER	12-8
	PART III: TRANSFERS REQUESTED BY TENANT	S
III.A.	OVERVIEW	12-9
III.B.	TYPES OF RESIDENT REQUESTED TRANSFERS	12-9
III.C.	ELIGIBILITY FOR TRANSFER	12-10
III.D.	SECURITY DEPOSITS	
III.E.	COST OF TRANSFER	12-11
III.F.	HANDLING OF REQUESTS	
	PART IV: TRANSFER PROCESSING	
IV.A.	OVERVIEW	12-13
IV.B.	110 11 (21 212 212 1	
IV.C.	TRANSFER OFFER POLICY	
	GOOD CAUSE FOR UNIT REFUSAL	
IV.E.	DECONCENTRATION	
IV.F.	REEXAMINATION POLICIES FOR TRANSFERS	12-15

Chapter 13 LEASE TERMINATIONS

INTR	ODUCTION	13-1
	PART I: TERMINATION BY TENANT	
I.A.	TENANT CHOOSES TO TERMINATE THE LEASE	13-3
	PART II: TERMINATION BY PHA – MANDATORY	
II.A.	OVERVIEW	13-5
II.B.	FAILURE TO PROVIDE CONSENT	13-5
II.C.	FAILURE TO DOCUMENT CITIZENSHIP	
II.D.	FAILURE TO DISCLOSE AND DOCUMENT SOCIAL SECURITY NUMBERS	13-6
II.E.	FAILURE TO ACCEPT THE PHA'S OFFER OF A LEASE REVISION	13-6
II.F.	METHAMPHETAMINE CONVICTION	13-7
II.G	LIFETIME REGISTERED SEX OFFENDERS	
II.H.	NONCOMPLIANCE WITH COMMUNITY SERVICE REQUIREMENTS	
II.I.	DEATH OF A SOLE FAMILY MEMBER	
	PART III: TERMINATION BY PHA – OTHER AUTHORIZED REASONS	
III.A.	OVERVIEW	13-9
III.B.		
III.C.		
III.D.	ALTERNATIVES TO TERMINATION OF TENANCY	13-19
III.E.	CRITERIA FOR DECIDING TO TERMINATE TENANCY	13-20
III.F.	TERMINATIONS RELATED TO DOMESTIC VIOLENCE,	
	DATING VIOLENCE, SEXUAL ASSAULT, OR STALKING	13-24
	PART IV: NOTIFICATION REQUIREMENTS, EVICTION	
	PROCEDURES AND RECORD KEEPING	
IV.A.	OVERVIEW	13-27
IV.B.	CONDUCTING CRIMINAL RECORDS CHECKS	13-27
	DISCLOSURE OF CRIMINAL RECORDS TO FAMILY	
	LEASE TERMINATION NOTICE	
	EVICTION	
	NOTIFICATION TO POST OFFICE	
	RECORD KEEPING	

Chapter 14 GRIEVANCES AND APPEALS

INTR	RODUCTION	14-1
	PART I: INFORMAL HEARINGS FOR PUBLIC HOUSING APPLICA	ANTS
I.A.	OVERVIEW	14-1
I.B.		
	PART II: INFORMAL HEARINGS WITH REGARD TO NONCITIZ	ENS
II.A.	HEARING AND APPEAL PROVISIONS FOR NONCITIZENS	14-5
P	PART III: GRIEVANCE PROCEDURES FOR PUBLIC HOUSING RESI	IDENTS
III.A.	REQUIREMENTS	14-9
III.B.	DEFINITIONS	14-10
III.C.		
III.D.	INFORMAL SETTLEMENT OF GRIEVANCE	14-12
	PROCEDURES TO OBTAIN A HEARING	
III.F.		
	PROCEDURES GOVERNING THE HEARING	
III.H.	DECISION OF THE HEARING OFFICER/PANEL	14-20
	Chapter 15 PROGRAM INTEGRITY	
INTR	RODUCTION	15-1
	PART I: PREVENTING, DETECTING, AND INVESTIGATING ERRORS AND PROGRAM ABUSE	
I.A.	PREVENTING ERRORS AND PROGRAM ABUSE	15-3
I.B.	DETECTING ERRORS AND PROGRAM ABUSE	
I.C.	INVESTIGATING ERRORS AND PROGRAM ABUSE	15-5
	PART II: CORRECTIVE MEASURES AND PENALTIES	
II.A.	UNDER- OR OVERPAYMENT	
II.B.	FAMILY-CAUSED ERRORS AND PROGRAM ABUSE	15-8
II.C.	PHA-CAUSED ERRORS OR PROGRAM ABUSE	15-10
II.D.	CRIMINAL PROSECUTION	
II.E.	FRAUD AND PROGRAM ABUSE RECOVERIES	15-12

Chapter 16 PROGRAM ADMINISTRATION

INTR	ODUCTION	16-1
	PART I: SETTING UTILITY ALLOWANCES	
I.A.	OVERVIEW	16-3
I.B	UTILITY ALLOWANCES	16-3
I.C.	SURCHARGES FOR PHA-FURNISHED UTILITIES	16-5
I.D.	NOTICE REQUIREMENTS	16-5
I.E.	REASONABLE ACCOMMODATION	16-6
	PART II: ESTABLISHING FLAT RENTS	
II.A.	OVERVIEW	16-7
II.B.	FLAT RENTS	16-7
	PART III: FAMILY DEBTS TO THE PHA	
III.A.	OVERVIEW	16-9
III.B.	REPAYMENT POLICY	16-10
	PART IV: PUBLIC HOUSING ASSESSMENT SYSTEM (PHAS)	
IV.A.	OVERVIEW	16-13
IV.B.	PHAS INDICATORS	16-13
IV.C.	PHAS SCORING	16-15
	PART V: RECORD KEEPING	
V.A.	OVERVIEW	16-17
V.B.	RECORD RETENTION	16-17
V.C.	RECORDS MANAGEMENT	16-18
	PART VI: REPORTING REQUIREMENTS FOR CHILDREN WITH ELEVATED BLOOD LEAD LEVEL	
VI.A.	REPORTING REQUIREMENTS	16-21

PART VII: VIOLENCE AGAINST WOMEN ACT (VAWA): NOTIFICATION, DOCUMENTATION, AND CONFIDENTIALITY

. 16-23 . 16-24 . 16-26 . 16-28
. 16-26
. 16-26
. 16-28
. 16-29
. 16-35
. 16-37
. 16-41

Glossary

Notice PIH 2007-27 (HA), Disallowed Costs and Sanctions Resulting from On-Site Monitoring Reviews

http://www.hud.gov/offices/pih/publications/notices/07/pih2007-27.pdf

Notice PIH 2012-10, Verification of Social Security Numbers (SSNs) and Supplemental Security Income (SSI) Benefits; and Effective Use of the Enterprise Income Verification (EIV) System's Identity Verification Report http://portal.hud.gov/huddoc/pih2012-10.pdf

Notice PIH 2017-12, Administrative Guidance for Effective and Mandated Use of the Enterprise Income Verification (EIV) System

https://www.hud.gov/sites/documents/PIH2017-12EIVNOTICE.PDF

Notice PIH 2010-26 (HA), Nondiscrimination and Accessibility Notice http://www.hud.gov/offices/pih/publications/notices/10/pih2010-26.pdf

OMB Circular A-133

http://www.whitehouse.gov/omb/circulars/a133_compliance_supplement_2010

Public Housing Occupancy Guidebook, June 2003

http://www.hud.gov/offices/pih/programs/ph/rhiip/phguidebook.cfm

Rental Housing Integrity Improvement Program (RHIIP) Frequently Asked Questions http://www.hud.gov/offices/pih/programs/ph/rhiip/faq.cfm

VAWA Reauthorization Act of 2013

http://www.gpo.gov/fdsys/pkg/FR-2013-08-06/pdf/2013-18920.pdf

Verification FAQs

http://www.hud.gov/offices/pih/programs/ph/rhiip/faq_verif.cfm

Verification Guidance, March 2004 (attachment to Notice PIH 2004-1) http://www.hud.gov/offices/pih/publications/notices/04/verifguidance.pdf

The HUD Web site is http://portal.hud.gov/hudportal/HUD.

Guidebooks, handbooks, and other HUD resources may be found at the HUDClips Web site: http://portal.hud.gov/hudportal/HUD?src=/program_offices/administration/hudclips.

Absent Head, Spouse, or Cohead

PHA Policy

An employed head, spouse, or cohead absent from the unit more than 180 consecutive days due to employment will continue to be considered a family member.

Individuals Confined for Medical Reasons

PHA Policy

An individual confined to a nursing home or hospital on a permanent basis is not considered a family member.

If there is a question about the status of a family member, the PHA will request verification from a responsible medical professional and will use this determination. If the responsible medical professional cannot provide a determination, the person generally will be considered temporarily absent. The family may present evidence that the family member is confined on a permanent basis and request that the person not be considered a family member.

Return of Permanently Absent Family Members

PHA Policy

The family must request PHA approval for the return of any adult family members that the PHA has determined to be permanently absent. The individual is subject to the eligibility and screening requirements discussed in this chapter.

3-I.M. LIVE-IN AIDE

Live-in aide means a person who resides with one or more elderly persons, or near-elderly persons, or persons with disabilities, and who: (1) is determined to be essential to the care and well-being of the person(s), (2) is not obligated for the support of the person(s), and (3) would not be living in the unit except to provide the necessary supportive services [24 CFR 5.403].

The PHA must approve a live-in aide if needed as a reasonable accommodation for a person with disabilities in accordance with 24 CFR 8.

A live-in aide is considered a household member but not a family member. The income of the live-in aide is not counted in determining the annual income of the family [24 CFR 5.609(c)(5)]. Relatives may be approved as live-in aides if they meet all of the criteria defining a live-in aide. However, a relative who serves as a live-in aide is not considered a family member and would not be considered a remaining member of a tenant family.

PHA Policy

A family's request for a live-in aide may be made either orally or in writing. The PHA will verify the need for a live-in aide, if necessary, with a reliable, knowledgeable professional as provided by the family, such as a doctor, social worker, or case worker. For continued approval, the family may be required to submit a new, written request—subject to PHA verification—at each annual reexamination.

In addition, the family and live-in aide will be required to submit a certification stating that the live-in aide is (1) not obligated for the support of the person(s) needing the care, and (2) would not be living in the unit except to provide the necessary supportive services.

The PHA has the discretion not to approve a particular person as a live-in aide, and may withdraw such approval, if [24 CFR 966.4(d)(3)(i)]:

The person commits fraud, bribery or any other corrupt or criminal act in connection with any federal housing program;

The person has a history of drug-related criminal activity or violent criminal activity; or

The person currently owes rent or other amounts to the PHA or to another PHA in connection with Section 8 or public housing assistance under the 1937 Act.

Within 10 business days of receiving a request for a live-in aide, including all required documentation related to the request, the PHA will notify the family of its decision in writing.

PART II: BASIC ELIGIBILITY CRITERIA

3-II.A. INCOME ELIGIBILITY AND TARGETING

Income Limits

HUD is required by law to establish income limits that determine the income eligibility of applicants for HUD's assisted housing programs, including the public housing program. The income limits are published annually and are based on HUD estimates of the median incomes for families of different sizes in a particular area or county.

Types of Low-Income Families [24 CFR 5.603(b)]

Low-income family. A family whose annual income does not exceed 80 percent of the median income for the area, adjusted for family size.

Very low-income family. A family whose annual income does not exceed 50 percent of the median income for the area, adjusted for family size.

Extremely low-income family. A family whose annual income does not exceed the federal poverty level or 30 percent of the median income for the area, whichever number is higher.

Area median income is determined by HUD, with adjustments for smaller and larger families. HUD may establish income ceilings higher or lower than 30, 50, or 80 percent of the median income for an area if HUD finds that such variations are necessary because of unusually high or low family incomes.

HUD also publishes over-income limits annually, but these are not used at admission. Over-income limits will be discussed in Chapter 13.

Using Income Limits for Eligibility [24 CFR 960.201]

Income limits are used to determine eligibility at admission. Eligibility is established by comparing a family's annual income with HUD's published income limits. To be income-eligible, a family must be a *low-income* family.

Using Income Limits for Targeting [24 CFR 960.202(b)]

At least 40 percent of the families admitted from the PHA waiting list to the public housing program during a PHA fiscal year must be *extremely low-income* families. This is called the "basic targeting requirement."

If admissions of extremely low-income families to the PHA's housing choice voucher program during a PHA fiscal year exceed the 75 percent minimum targeting requirement for that program, such excess shall be credited against the PHA's public housing basic targeting requirement for the same fiscal year.

The fiscal year credit for housing choice voucher program admissions that exceed the minimum voucher program targeting requirement must not exceed the lower of:

- Ten percent of public housing waiting list admissions during the PHA fiscal year
- Ten percent of waiting list admission to the PHA's housing choice voucher program during the PHA fiscal year
- The number of qualifying low-income families who commence occupancy during the fiscal year of public housing units located in census tracts with a poverty rate of 30 percent or more. For this purpose, qualifying low-income family means a low-income family other than an extremely low-income family.

For discussion of how income targeting is used in tenant selection, see Chapter 4.

Phasing In Flat Rents [Notice PIH 2017-23; 24 CFR 960.253(b)]

When new flat rents requirements were implemented in 2014, HUD limited the increase for existing residents paying flat rent at that time to no more than 35 percent of the current tenant rent per year. In some cases, this meant that some residents had or will have their flat rents phased-in at the time of their annual recertification. To do this, PHAs conduct a flat rent impact analysis to determine whether a phase-in is or was necessary. For families whose flat rent is being phased-in, the PHA must multiply the family's current rent amount by 1.35 and compare the result to the flat rent under the PHA's policies. Families who have subsequently been admitted to the program or have subsequently selected flat rent will not experience a phase-in.

Notice PIH 2017-23 requires that flat rents must be phased in at the full 35 percent per year. PHAs do not have the option of phasing in flat rent increases at less than 35 percent per year.

Example: A family was paying a flat rent of \$500 per month. At their annual recertification, the PHA has increased the flat rent for their unit size to comply with the new requirements to \$700. The PHA conducted a flat rent impact analysis as follows:

$$$500 \times 1.35 = $675$$

Since the PHA's increased flat rent of \$700 resulted in a rent increase of more than 35 percent, the PHA offered the family the choice to pay either \$675 per month or an income-based rent. The flat rent increase was phased in. At their next annual recertification in November 2015, the PHA will again multiply the family's current flat rent by 1.35 and compare the results to the PHA's current flat rent.

Flat Rents and Earned Income Disallowance [A&O FAQs]

Because the EID is a function of income-based rents, a family paying flat rent cannot qualify for the EID even if a family member experiences an event that would qualify the family for the EID. If the family later chooses to pay income-based rent, they would only qualify for the EID if a new qualifying event occurred.

Under the EID original calculation method, a family currently paying flat rent that previously qualified for the EID while paying income-based rent and is currently within their exclusion period would have the exclusion period continue while paying flat rent as long as the employment that is the subject of the exclusion continues. A family paying flat rent could therefore see a family member's exclusion period expire while the family is paying flat rent.

Under the EID revised calculation method, a family currently paying flat rent that previously qualified for the EID while paying income-based rent and is currently within their exclusion period would have the exclusion period continue while paying flat rent regardless whether the employment that is the subject of the exclusion continues. A family paying flat rent could therefore see a family member's exclusion period expire while the family is paying flat rent.

Chapter 7

VERIFICATION

[24 CFR 960.259, 24 CFR 5.230, Notice PIH 2017-12]

INTRODUCTION

The PHA must verify all information that is used to establish the family's eligibility and level of assistance and is required to obtain written authorization from the family in order to collect the information. Applicants and program participants must cooperate with the verification process as a condition of receiving assistance. The PHA must not pass on the cost of verification to the family.

The PHA will follow the verification guidance provided by HUD in Notice PIH 2017-12 and any subsequent guidance issued by HUD. This chapter summarizes those requirements and provides supplementary PHA policies.

Part I describes the general verification process. Part II provides more detailed requirements related to family information. Part III provides information on income and assets, and Part IV covers mandatory deductions.

Verification policies, rules and procedures will be modified as needed to accommodate persons with disabilities. All information obtained through the verification process will be handled in accordance with the records management policies established by the PHA.

PART I: GENERAL VERIFICATION REQUIREMENTS

7-I.A. FAMILY CONSENT TO RELEASE OF INFORMATION [24 CFR 960.259, 24 CFR 5.230]

The family must supply any information that the PHA or HUD determines is necessary to the administration of the program and must consent to PHA verification of that information [24 CFR 960.259(a)(1)].

Consent Forms

It is required that all adult applicants and tenants sign form HUD-9886, Authorization for Release of Information. The purpose of form HUD-9886 is to facilitate automated data collection and computer matching from specific sources and provides the family's consent only for the specific purposes listed on the form. HUD and the PHA may collect information from State Wage Information Collection Agencies (SWICAs) and current and former employers of adult family members. Only HUD is authorized to collect information directly from the Internal Revenue Service (IRS) and the Social Security Administration (SSA). Adult family members must sign other consent forms as needed to collect information relevant to the family's eligibility and level of assistance.

Penalties for Failing to Consent [24 CFR 5.232]

If any family member who is required to sign a consent form fails to do so, the PHA will deny admission to applicants and terminate the lease of tenants. The family may request a hearing in accordance with the PHA's grievance procedures.

7-I.B. OVERVIEW OF VERIFICATION REQUIREMENTS

HUD's Verification Hierarchy [Notice PIH 2017-12]

HUD mandates the use of the EIV system and offers administrative guidance on the use of other methods to verify family information and specifies the circumstances in which each method will be used. In general, HUD requires the PHA to use the most reliable form of verification that is available and to document the reasons when the PHA uses a lesser form of verification.

In order of priority, the forms of verification that the PHA will use are:

- Up-front Income Verification (UIV) using HUD's Enterprise Income Verification (EIV) system
- Up-front Income Verification (UIV) using a non-HUD system
- Written Third Party Verification (may be provided by applicant or resident)
- Written Third-party Verification Form
- Oral Third-party Verification
- Self-Certification

Each of the verification methods is discussed in subsequent sections below.

Requirements for Acceptable Documents

PHA Policy

Any documents used for verification must be the original (not photocopies) and generally must be dated within 60 days of the PHA request. The documents must not be damaged, altered or in any way illegible.

Print-outs from web pages are considered original documents.

The PHA staff member who views the original document must make a photocopy, annotate the copy with the name of the person who provided the document and the date the original was viewed, and sign the copy.

Any family self-certifications must be made in a format acceptable to the PHA and must be signed in the presence of a PHA representative or PHA notary public.

File Documentation

The PHA must document in the file how the figures used in income and rent calculations were determined. All verification attempts, information obtained, and decisions reached during the verification process will be recorded in the family's file in sufficient detail to demonstrate that the PHA has followed all of the verification policies set forth in this ACOP. The record should be sufficient to enable a staff member or HUD reviewer to understand the process followed and conclusions reached.

PHA Policy

The PHA will document, in the family file, the following:

Reported family annual income

Value of assets

Expenses related to deductions from annual income

Other factors influencing the adjusted income or income-based rent determination

When the PHA is unable to obtain third-party verification, the PHA will document in the family file the reason that third-party verification was not available [24 CFR 960.259(c)(1); Notice PIH 2017-12].

7-I.C. UP-FRONT INCOME VERIFICATION (UIV)

Up-front income verification (UIV) refers to the PHA's use of the verification tools available from independent sources that maintain computerized information about earnings and benefits. UIV will be used to the extent that these systems are available to the PHA.

There may be legitimate differences between the information provided by the family and UIV-generated information. If the family disputes the accuracy of UIV data, no adverse action can be taken until the PHA has independently verified the UIV information and the family has been granted the opportunity to contest any adverse findings through the PHA's informal review/hearing processes. (For more on UIV and income projection, see section 6-I.C.)

Upfront Income Verification Using HUD's Enterprise Income Verification (EIV) System (Mandatory)

PHAs must use HUD's EIV system in its entirety as a third-party source to verify tenant employment and income information during mandatory reexaminations or recertifications of family composition and income in accordance with 24 CFR 5.236 and administrative guidance issued by HUD. HUD's EIV system contains data showing earned income, unemployment benefits, social security benefits, and SSI benefits for participant families. The following policies apply to the use of HUD's EIV system.

EIV Income Reports

The data shown on income reports is updated quarterly. Data may be between three and six months old at the time reports are generated.

PHA Policy

The PHA will obtain income reports for annual reexaminations on a monthly basis. Reports will be generated as part of the regular reexamination process.

Income reports will be compared to family-provided information as part of the annual reexamination process. Income reports may be used in the calculation of annual income, as described in Chapter 6.I.C. Income reports may also be used to meet the regulatory requirement for third party verification, as described above. Policies for resolving discrepancies between income reports and family-provided information will be resolved as described in Chapter 6.I.C. and in this chapter.

Income reports will be used in interim reexaminations to identify any discrepancies between reported income and income shown in the EIV system, and as necessary to verify earned income, and to verify and calculate unemployment benefits, Social Security and/or SSI benefits. EIV will also be used to verify that families claiming zero income are not receiving income from any of these sources.

Income reports will be retained in resident files with the applicable annual or interim reexamination documents.

When the PHA determines through income reports and third-party verification that a family has concealed or under-reported income, corrective action will be taken pursuant to the policies in Chapter 15, Program Integrity.

EIV Identity Verification

The EIV system verifies resident identities against Social Security Administration (SSA) records. These records are compared to Public and Indian Housing Information Center (PIC) data for a match on social security number, name, and date of birth.

PHAs are required to use EIV's *Identity Verification Report* on a monthly basis to improve the availability of income information in EIV [Notice PIH 2017-12].

When identity verification for a resident fails, a message will be displayed within the EIV system and no income information will be displayed.

PHA Policy

The PHA will identify residents whose identity verification has failed by reviewing EIV's *Identity Verification Report* on a monthly basis. The PHA will attempt to resolve PIC/SSA discrepancies by obtaining appropriate documentation from the tenant. When the PHA determines that discrepancies exist as a result of PHA errors, such as spelling errors or incorrect birth dates, it will correct the errors promptly.

Upfront Income Verification Using Non-HUD Systems (Optional)

In addition to mandatory use of the EIV system, HUD encourages PHAs to utilize other upfront verification sources.

PHA Policy

The PHA will inform all applicants and residents of its use of the following UIV resources during the admission and reexamination process:

HUD's EIV system

[Insert any additional UIV sources used by the PHA]

7-I.D. THIRD-PARTY WRITTEN AND ORAL VERIFICATION

HUD's current verification hierarchy defines two types of written third-party verification. The more preferable form, "written third-party verification," consists of an original document generated by a third-party source, which may be received directly from a third-party source or provided to the PHA by the family. If written third-party verification is not available, the PHA must attempt to obtain a "written third-party verification form." This is a standardized form used to collect information from a third party.

Written Third-Party Verification [Notice PIH 2017-12]

Written third-party verification documents must be original and authentic and may be supplied by the family or received from a third-party source.

Examples of acceptable tenant-provided documents include, but are not limited to: pay stubs, payroll summary reports, employer notice or letters of hire and termination, SSA benefit verification letters, bank statements, child support payment stubs, welfare benefit letters and/or printouts, and unemployment monetary benefit notices.

The PHA is required to obtain, at minimum, two current and consecutive pay stubs for determining annual income from wages.

The PHA may reject documentation provided by the family if the document is not an original, if the document appears to be forged, or if the document is altered, mutilated, or illegible.

PHA Policy

Third-party documents provided by the family must be dated within 60 days of the PHA request date.

If the PHA determines that third-party documents provided by the family are not acceptable, the PHA will explain the reason to the family and request additional documentation.

As verification of earned income, the PHA will require the family to provide the two most current, consecutive pay stubs.

Written Third-Party Verification Form

When upfront verification is not available and the family is unable to provide written third-party documents, the PHA must request a written third-party verification form. HUD's position is that this traditional third-party verification method presents administrative burdens and risks which may be reduced through the use of family-provided third-party documents.

PHAs may mail, fax, or e-mail third-party written verification form requests to third-party sources.

PHA Policy

The PHA will send third-party verification forms directly to the third party.

Third-party verification forms will be sent when third-party verification documents are unavailable or are rejected by the PHA.

Oral Third-Party Verification [Notice PIH 2017-12]

For third-party oral verification, PHAs contact sources, identified by UIV techniques or by the family, by telephone or in person.

Oral third-party verification is mandatory if neither form of written third-party verification is available.

Third-party oral verification may be used when requests for written third-party verification forms have not been returned within a reasonable time—e.g., 10 business days.

PHAs should document in the file the date and time of the telephone call or visit, the name of the person contacted, the telephone number, as well as the information confirmed.

PHA Policy

In collecting third-party oral verification, PHA staff will record in the family's file the name and title of the person contacted, the date and time of the conversation (or attempt), the telephone number used, and the facts provided.

When any source responds verbally to the initial written request for verification the PHA will accept the verbal response as oral verification but will also request that the source complete and return any verification forms that were provided.

When Third-Party Verification is Not Required [Notice PIH 2017-12]

Third-party verification may not be available in all situations. HUD has acknowledged that it may not be cost-effective or reasonable to obtain third-party verification of income, assets, or expenses when these items would have a minimal impact on the family's total tenant payment.

PHA Policy

If the family cannot provide original documents, the PHA will pay the service charge required to obtain third-party verification, unless it is not cost effective in which case a self-certification will be acceptable as the only means of verification. The cost of verification will not be passed on to the family.

The cost of postage and envelopes to obtain third-party verification of income, assets, and expenses is not an unreasonable cost [VG, p. 18].

Primary Documents

Third-party verification is not required when legal documents are the primary source, such as a birth certificate or other legal documentation of birth.

Imputed Assets

The PHA may accept a self-certification from the family as verification of assets disposed of for less than fair market value [HCV GB, p. 5-28].

PHA Policy

The PHA will accept a self-certification from a family as verification of assets disposed of for less than fair market value [HCV GB, p. 5-28].

Value of Assets and Asset Income [24 CFR 960.259]

For families with net assets totaling \$5,000 or less, the PHA may accept the family's declaration of asset value and anticipated asset income. However, the PHA is required to obtain third-party verification of all assets regardless of the amount during the intake process, whenever a family member is added, and at least every three years thereafter.

PHA Policy

For families with net assets totaling \$5,000 or less, the PHA will accept the family's self-certification of the value of family assets and anticipated asset income when applicable. The family's declaration must show each asset and the amount of income expected from that asset. All family members 18 years of age and older must sign the family's declaration.

The PHA will use third-party documentation for assets as part of the intake process, whenever a family member is added to verify the individual's assets, and every three years thereafter.

Chapter 8

LEASING AND INSPECTIONS

[24 CFR 5, Subpart G; 24 CFR 966, Subpart A]

INTRODUCTION

Public housing leases are the contractual basis of the legal relationship between the PHA and the tenant. All units must be occupied pursuant to a dwelling lease agreement that complies with HUD regulations.

HUD regulations require the PHA to inspect each dwelling unit prior to move-in, at move-out, and annually during the period of occupancy. In addition, the PHA may conduct additional inspections in accordance with PHA policy.

This chapter is divided into two parts as follows:

<u>Part I: Leasing</u>. This part describes pre-leasing activities and the PHA's policies pertaining to lease execution, lease modification, and payments under the lease.

Part II: Inspections. This part describes the PHA's policies for inspecting dwelling units.

PART I: LEASING

8-I.A. OVERVIEW

An eligible family may occupy a public housing dwelling unit under the terms of a lease. The lease must meet all regulatory requirements, and must also comply with applicable state and local laws and codes.

The term of the lease must be for a period of 12 months. The lease must be renewed automatically for another 12-month term, except that the PHA may not renew the lease if the family has violated the community service requirement [24 CFR 966.4(a)(2)].

PHAs must adopt smoke-free policies, which must be implemented no later than July 30, 2018. The policy is attached as Exhibit 8-1.

Part I of this chapter contains regulatory information on leasing, where applicable, as well as the PHA's leasing policies.

8-I.B. LEASE ORIENTATION

PHA Policy

After unit acceptance but prior to occupancy, a PHA representative will conduct a lease orientation with the family. The head of household or spouse is required to attend.

Orientation Agenda

PHA Policy

When families attend the lease orientation, they will be provided with:

A copy of the lease

A copy of the PHA's grievance procedure

A copy of the house rules

A copy of the PHA's schedule of maintenance charges

A copy of "Is Fraud Worth It?" (form HUD-1141-OIG), which explains the types of actions a family must avoid and the penalties for program abuse

A copy of "What You Should Know about EIV," a guide to the Enterprise Income Verification (EIV) system published by HUD as an attachment to Notice PIH 2017-12

A copy of the form HUD-5380, VAWA Notice of Occupancy Rights

A copy of form HUD-5382, Certification of Domestic Violence, Dating Violence, Sexual Assault, or Stalking

A copy of the PHA's smoke free policy

A notice that includes the procedures for requesting relief and the PHA's criteria for granting requests for relief for excess utility surcharges

Topics to be discussed and explained to all families include:

Applicable deposits and all other charges

Review and explanation of lease provisions

Unit maintenance requests and work orders

The PHA's interim reporting requirements

Review and explanation of occupancy forms

Community service requirements

Family choice of rent

VAWA protections

Smoke-free policies

PART I: ANNUAL REEXAMINATIONS FOR FAMILIES PAYING INCOME-BASED RENTS [24 CFR 960.257]

9-I.A. OVERVIEW

For those families who choose to pay income-based rent, the PHA must conduct a reexamination of income and family composition at least annually [24 CFR 960.257(a)(1)]. For families who choose flat rents, the PHA must conduct a reexamination of family composition at least annually, and must conduct a reexamination of family income at least once every 3 years [24 CFR 960.257(a)(2)]. Policies related to the reexamination process for families paying flat rent are located in Part II of this chapter.

For all residents of public housing, whether those residents are paying income-based or flat rents, the PHA must conduct an annual review of community service requirement compliance. This annual reexamination is also a good time to have residents sign consent forms for criminal background checks in case the criminal history of a resident is needed at some point for the purposes of lease enforcement or eviction.

The PHA is required to obtain all of the information necessary to conduct reexaminations. How that information will be collected is left to the discretion of the PHA. Families are required to provide current and accurate information on income, assets, allowances and deductions, family composition and community service compliance as part of the reexamination process [24 CFR 960.259].

This part contains the PHA's policies for conducting annual reexaminations.

9-I.B STREAMLINED ANNUAL REEXAMINATIONS [24 CFR 960.257]

HUD permits PHAs to streamline the income determination process for family members with fixed sources of income. While third-party verification of all income sources must be obtained during the intake process and every three years thereafter, in the intervening years the PHA may determine income from fixed sources by applying a verified cost of living adjustment (COLA) or rate of interest. The PHA may, however, obtain third-party verification of all income, regardless of the source. Further, upon request of the family, the PHA must perform third-party verification of all income sources.

Fixed sources of income include Social Security and SSI benefits, pensions, annuities, disability or death benefits, and other sources of income subject to a COLA or rate of interest. The determination of fixed income may be streamlined even if the family also receives income from other non-fixed sources.

Two streamlining options are available, depending upon the percentage of the family's income that is received from fixed sources. If at least 90 percent of the family's income is from fixed sources, the PHA may streamline the verification of fixed income and may choose whether to verify non-fixed income amounts in years where no fixed-income review is required. If the family receives less than 90 percent of its income from fixed sources, the PHA may streamline the verification of fixed income and must verify non-fixed income annually.

PHA Policy

The PHA will streamline the annual reexamination process by applying the verified COLA or interest rate to fixed-income sources. The PHA will document in the file how the determination that a source of income was fixed was made.

If a family member with a fixed source of income is added, the PHA will use third-party verification of all income amounts for that family member.

If verification of the COLA or rate of interest is not available, the PHA will obtain third-party verification of income amounts.

Third-party verification of fixed sources of income will be obtained during the intake process and at least once every three years thereafter.

Third-party verification of non-fixed income will be obtained annually regardless of the percentage of family income received from fixed sources.

9-I.C. SCHEDULING ANNUAL REEXAMINATIONS

The PHA must establish a policy to ensure that the annual reexamination for each family paying an income-based rent is completed within a 12 month period [24 CFR 960.257(a)(1)].

PHA Policy

Generally, the PHA will schedule annual reexaminations to coincide with the family's anniversary date. The PHA will begin the annual reexamination process approximately 120 days in advance of the scheduled effective date.

Anniversary date is defined as 12 months from the effective date of the family's last annual reexamination or, during a family's first year in the program, from the effective date of the family's initial examination (admission).

If the family transfers to a new unit, the PHA will perform a new annual reexamination, and the anniversary date will be changed.

The PHA may also schedule an annual reexamination for completion prior to the anniversary date for administrative purposes.

Notification of and Participation in the Annual Reexamination Process

The PHA is required to obtain information needed to conduct annual reexaminations. How that information will be collected is left to the discretion of the PHA. However, PHAs should give tenants who were not provided the opportunity to provide contact information at the time of admission the option to complete Form HUD-92006 at this time. The PHA should provide the family with the opportunity to update, change, or remove information from the HUD-92006 at the time of the annual reexamination [Notice PIH 2009-36].

PHA Policy

Families generally are required to participate in an annual reexamination interview, which must be attended by the head of household, spouse, or cohead. If participation in an in-person interview poses a hardship because of a family member's disability, the family should contact the PHA to request a reasonable accommodation.

Notification of annual reexamination interviews will be sent by first-class mail and will contain the date, time, and location of the interview. In addition, it will inform the family of the information and documentation that must be brought to the interview.

If the family is unable to attend a scheduled interview, the family should contact the PHA in advance of the interview to schedule a new appointment. In all circumstances, if a family does not attend the scheduled interview the PHA will send a second notification with a new interview appointment time.

If a family fails to attend two scheduled interviews without PHA approval, the family will be in violation of their lease and may be terminated in accordance with the policies in Chapter 13.

An advocate, interpreter, or other assistant may assist the family in the interview process.

9-I.D. CONDUCTING ANNUAL REEXAMINATIONS

The terms of the public housing lease require the family to furnish information regarding income and family composition as may be necessary for the redetermination of rent, eligibility, and the appropriateness of the housing unit [24 CFR 966.4(c)(2)].

PHA Policy

Families will be asked to bring all required information (as described in the reexamination notice) to the reexamination appointment. The required information will include a PHA-designated reexamination form, an Authorization for the Release of Information/Privacy Act Notice, as well as supporting documentation related to the family's income, expenses, and family composition.

Any required documents or information that the family is unable to provide at the time of the interview must be provided within 10 business days of the interview. If the family is unable to obtain the information or materials within the required time frame, the family may request an extension.

If the family does not provide the required documents or information within the required time frame (plus any extensions), the family will be in violation of their lease and may be terminated in accordance with the policies in Chapter 13.

The information provided by the family generally must be verified in accordance with the policies in Chapter 7. Unless the family reports a change, or the agency has reason to believe a change has occurred in information previously reported by the family, certain types of information that are verified at admission typically do not need to be re-verified on an annual basis. These include:

- Legal identity
- Age
- Social security numbers
- A person's disability status
- Citizenship or immigration status

10-II.C. STANDARDS FOR PETS [24 CFR 5.318; 960.707(b)]

PHAs may establish reasonable requirements related to pet ownership including, but not limited to:

- Limitations on the number of animals in a unit, based on unit size
- Prohibitions on types of animals that the PHA classifies as dangerous, provided that such classifications are consistent with applicable state and local law
- Prohibitions on individual animals, based on certain factors, including the size and weight of the animal
- Requiring pet owners to have their pets spayed or neutered

PHAs may not require pet owners to have any pet's vocal cords removed.

PHAs may not require pet owners to obtain or carry liability insurance.

PHAs may not require that cats be declawed.

Definition of "Common Household Pet"

There is no regulatory definition of common household pet for public housing programs, although the regulations for pet ownership in both elderly/disabled and general occupancy developments use the term. The regulations for pet ownership in elderly/disabled developments expressly authorize PHAs to define the term [24 CFR 5.306(2)].

PHA Policy

Common household pet means a domesticated animal, such as a dog, cat, bird, or fish that is traditionally recognized as a companion animal and is kept in the home for pleasure rather than commercial purposes.

The following animals are not considered common household pets:

Reptiles

Rodents

Insects

Arachnids

Wild animals or feral animals

Pot-bellied pigs

Animals used for commercial breeding

Pet Restrictions

PHA Policy

The following animals are not permitted:

Any animal whose adult weight will exceed 25 pounds

Dogs of the pit bull, rottweiler, chow, or boxer breeds

Ferrets or other animals whose natural protective mechanisms pose a risk to small children of serious bites or lacerations

Any animal not permitted under state or local law or code

Number of Pets

PHA Policy

Residents may own a maximum of 2 pets, only 1 of which may be a dog.

In the case of fish, residents may keep no more than can be maintained in a safe and healthy manner in a tank holding up to 10 gallons. Such a tank or aquarium will be counted as 1 pet.

Other Requirements

PHA Policy

Dogs and cats must be spayed or neutered at the time of registration or, in the case of underage animals, within 30 days of the pet reaching 6 months of age. Exceptions may be made upon veterinary certification that subjecting this particular pet to the procedure would be temporarily or permanently medically unsafe or unnecessary.

Pets must be licensed in accordance with state or local law. Residents must provide proof of licensing at the time of registration and annually, in conjunction with the resident's annual reexamination.

PART III: TERMINATION BY PHA – OTHER AUTHORIZED REASONS

13-III.A. OVERVIEW

Besides requiring PHAs to terminate the lease under the circumstances described in Part II, HUD requires the PHA to establish provisions in the lease for termination pertaining to certain criminal activity, alcohol abuse, and certain household obligations stated in the regulations. While these provisions for lease termination must be in the lease agreement, HUD does not require PHAs to terminate for such violations in all cases. The PHA has the discretion to consider circumstances surrounding the violation or, in applicable situations, whether the offending household member has entered or completed rehabilitation, and the PHA may, as an alternative to termination, require the exclusion of the culpable household member. The PHA must adopt policies concerning the use of these options.

In addition, HUD authorizes PHAs to terminate the lease for other grounds, but for only those grounds that constitute serious or repeated violations of material terms of the lease or for other good cause. The PHA must develop policies pertaining to what constitutes serious or repeated lease violations, and other good cause, based upon the content of the PHA lease. In the development of the terms of the lease, the PHA must consider the limitations imposed by state and local landlord-tenant law, as well as HUD regulations and federal statutes. Because of variations in state and local landlord-tenant law, and because HUD affords PHAs wide discretion in some areas, a broad range of policies could be acceptable.

The PHA also has the option to terminate the tenancies of certain over-income families.

The PHA may consider alternatives to termination and must establish policies describing the criteria the PHA will use when deciding what action to take, the types of evidence that will be acceptable, and the steps the PHA must take when terminating a family's lease.

13-III.B. MANDATORY LEASE PROVISIONS [24 CFR 966.4(1)(5)]

This section addresses provisions for lease termination that must be included in the lease agreement according to HUD regulations. Although the provisions are required, HUD does not require PHAs to terminate for such violations in all cases, therefore PHA policies are needed.

Definitions [24 CFR 5.100]

The following definitions will be used for this and other parts of this chapter:

Affiliated individual is defined in section 16-VII.B.

Covered person means a tenant, any member of the tenant's household, a guest, or another person under the tenant's control.

Dating violence is defined in section 16-VII.B.

Domestic violence is defined in section 16-VII.B.

Drug means a controlled substance as defined in section 102 of the Controlled Substances Act [21 U.S.C. 802].

Drug-related criminal activity means the illegal manufacture, sale, distribution, or use of a drug, or the possession of a drug with the intent to manufacture, sell, distribute, or use the drug.

Guest means a person temporarily staying in the unit with the consent of a tenant or other member of the household who has express or implied authority to so consent on behalf of the tenant.

Household means the family and PHA-approved live-in aide. The term household also includes foster children and/or foster adults that have been approved to reside in the unit [HUD-50058, Instruction Booklet, p. 65].

Other person under the tenant's control means that the person, although not staying as a guest in the unit, is, or was at the time of the activity in question, on the premises because of an invitation from the tenant or other member of the household who has express or implied authority to so consent on behalf of the tenant. Absent evidence to the contrary, a person temporarily and infrequently on the premises solely for legitimate commercial purposes is not *under the tenant's control*.

Premises means the building or complex or development in which the public or assisted housing dwelling unit is located, including common areas and grounds.

Sexual assault is defined in section 16-VII.B.

Stalking is defined in section 16-VII.B.

Violent criminal activity means any criminal activity that has as one of its elements the use, attempted use, or threatened use of physical force substantial enough to cause, or be reasonably likely to cause, serious bodily injury or property damage.

If a family member has violated federal, state, or local law that imposes obligations in connection with the occupancy or use of the premises.

If a household member has engaged in or threatened violent or abusive behavior toward PHA personnel.

Abusive or violent behavior towards PHA personnel includes verbal as well as physical abuse or violence. Use of racial epithets, or other language, written or oral, that is customarily used to intimidate may be considered abusive or violent behavior.

Threatening refers to oral or written threats or physical gestures that communicate intent to abuse or commit violence.

In making its decision to terminate the lease, the PHA will consider alternatives as described in Section 13-III.D and other factors described in Sections 13-III.E and 13-III.F. Upon consideration of such alternatives and factors, the PHA may, on a case-by-case basis, choose not to terminate the lease.

Family Absence from Unit [24 CFR 982.551(i)]

It is reasonable that the family may be absent from the public housing unit for brief periods. However, the PHA needs a policy on how long the family may be absent from the unit. Absence in this context means that no member of the family is residing in the unit.

PHA Policy

The family must supply any information or certification requested by the PHA to verify that the family is living in the unit, or relating to family absence from the unit, including any PHA-requested information or certification on the purposes of family absences. The family must cooperate with the PHA for this purpose.

The family must promptly notify the PHA when all family members will be absent from the unit for an extended period. An extended period is defined as any period greater than 30 calendar days. In such a case promptly means within 10 business days of the start of the extended absence.

If a family is absent from the public housing unit for more than 180 consecutive days, and the family does not adequately verify that they are living in the unit, the PHA will terminate the lease for other good cause.

Abandonment of the unit. If the family appears to have vacated the unit without giving proper notice, the PHA will follow state and local landlord-tenant law pertaining to abandonment before taking possession of the unit. If necessary, the PHA will secure the unit immediately to prevent vandalism and other criminal activity.

Over-Income Families [24 CFR 960.261; FR Notice 7/26/18]

The Housing Opportunity Through Modernization Act (HOTMA) of 2016 placed an income limitation on public housing tenancies. The over-income requirement states that after a family's income has exceeded 120 percent of area median income (AMI) (or a different limitation established by the secretary) for two consecutive years, the PHA must either terminate the family's tenancy within six months of the determination, or charge the family a monthly rent that is the higher of the applicable fair market rent (FMR) or the amount of monthly subsidy for the unit, including amounts from the operating and capital funds, as determined by regulations.

PHAs also have discretion, under 24 CFR 960.261, to adopt policies allowing termination of tenancy for families whose income exceeds the limit for program eligibility. Such policies would exempt families participating in the Family Self-Sufficiency (FSS) program or currently receiving the earned income disallowance.

PHA Policy

At annual or interim reexamination, if a family's income exceeds the applicable over-income limit, the PHA will document the family file and begin tracking the family's over-income status.

If one year after the applicable annual or interim reexamination the family's income continues to exceed the applicable over-income limit, the PHA will notify the family in writing that their income has exceeded the over-income limit for one year, and that if the family continues to be over-income for 12 consecutive months, the family will be subject to the PHA's over-income policies.

If two years after the applicable annual or interim reexamination the family's income continues to exceed the applicable over-income limit, the PHA will charge the family a rent that is the higher of the applicable fair market rent (FMR) or the amount of monthly subsidy for the unit. The PHA will notify the family in writing of their new rent amount. The new rent amount will be effective 30 days after the PHA's written notice to the family.

If, at any time, an over-income family experiences a decrease in income, the family may request an interim redetermination of rent in accordance with PHA policy. If, as a result, the previously over-income family is now below the over-income limit, the family is no longer subject to over-income provisions as of the effective date of the recertification. The PHA will notify the family in writing that over-income policies no longer apply to them. If the family's income later exceeds the over-income limit again, the family is entitled to a new two-year grace period.

The PHA will begin tracking over-income families once these policies have been adopted, but no later than March 24, 2019.

The PHA will not evict or terminate the tenancies of families whose income exceeds the income limit for program eligibility as described at 24 CFR 960.261.

13-III.D. ALTERNATIVES TO TERMINATION OF TENANCY

Exclusion of Culpable Household Member [24 CFR 966.4(1)(5)(vii)(C)]

As an alternative to termination of the lease for criminal activity or alcohol abuse HUD provides that the PHA may consider exclusion of the culpable household member. Such an alternative can be used for any other reason where such a solution appears viable in accordance with PHA policy.

Additionally, under the Violence against Women Reauthorization Act of 2013, the PHA may bifurcate a lease in order to terminate the tenancy of an individual who is a tenant or lawful occupant of a unit and engages in criminal activity directly related to domestic violence, dating violence, sexual assault, or stalking.

PHA Policy

The PHA will consider requiring the tenant to exclude a household member in order to continue to reside in the assisted unit, where that household member has participated in or been culpable for action or failure to act that warrants termination.

As a condition of the family's continued occupancy, the head of household must certify that the culpable household member has vacated the unit and will not be permitted to visit or to stay as a guest in the assisted unit. The family must present evidence of the former household member's current address upon PHA request.

Repayment of Family Debts

PHA Policy

If a family owes amounts to the PHA, as a condition of continued occupancy, the PHA will require the family to repay the full amount or to enter into a repayment agreement, within 30 days of receiving notice from the PHA of the amount owed. See Chapter 16 for policies on repayment agreements.

13-III.E. CRITERIA FOR DECIDING TO TERMINATE TENANCY

A PHA that has grounds to terminate a tenancy is not required to do so, except as explained in Part II of this chapter, and may consider all of the circumstances relevant to a particular case before making a decision.

Evidence [24 CFR 982.553(c)]

For criminal activity, HUD permits the PHA to terminate the lease if a *preponderance of the evidence* indicates that a household member has engaged in the activity, regardless of whether the household member has been arrested or convicted, and without satisfying the standard of proof used for a criminal conviction.

PHA Policy

The PHA will use the preponderance of the evidence as the standard for making all termination decisions.

Preponderance of the evidence is defined as evidence which is of greater weight or more convincing than the evidence which is offered in opposition to it; that is, evidence which as a whole shows that the fact sought to be proved is more probable than not.

Preponderance of the evidence may not be determined by the number of witnesses, but by the greater weight of all evidence.

Consideration of Circumstances [24 CFR 966.4(l)(5)(vii)(B)]

Although it is required that certain lease provisions exist for criminal activity and alcohol abuse, HUD provides that the PHA may consider all circumstances relevant to a particular case in order to determine whether or not to terminate the lease.

Such relevant circumstances can also be considered when terminating the lease for any other reason.

PHA Policy

The PHA will consider the following facts and circumstances before deciding whether to terminate the lease for any of the HUD required lease provisions or for any other reasons:

The seriousness of the offending action, especially with respect to how it would affect other residents' safety or property

The extent of participation or culpability of the leaseholder, or other household members, in the offending action, including whether the culpable member is a minor, a person with disabilities, or (as discussed further in section 13-III.F) a victim of domestic violence, dating violence, sexual assault, or stalking

The effects that the eviction will have on other family members who were not involved in the action or failure to act

The effect on the community of the termination, or of the PHA's failure to terminate the tenancy

The effect of the PHA's decision on the integrity of the public housing program

The demand for housing by eligible families who will adhere to lease responsibilities

The extent to which the leaseholder has shown personal responsibility and whether they have taken all reasonable steps to prevent or mitigate the offending action

The length of time since the violation occurred, including the age of the individual at the time of the conduct, as well as the family's recent history, and the likelihood of favorable conduct in the future

While a record of arrest(s) will not be used as the basis for termination, an arrest may, however, trigger an investigation to determine whether the participant actually engaged in disqualifying criminal activity. As part of its investigation, the PHA may obtain the police report associated with the arrest and consider the reported circumstances of the arrest. The PHA may also consider:

Any statements made by witnesses or the participant not included in the police report

Whether criminal charges were filed

Whether, if filed, criminal charges were abandoned, dismissed, not prosecuted, or ultimately resulted in an acquittal

Any other evidence relevant to determining whether or not the participant engaged in disqualifying activity

Evidence of criminal conduct will be considered if it indicates a demonstrable risk to safety and/or property.

In the case of program abuse, the dollar amount of the underpaid rent and whether or not a false certification was signed by the family

Consideration of Rehabilitation [24 CFR 966.4(l)(5)(vii)(D)]

HUD authorizes PHAs to take into consideration whether a household member who had used illegal drugs or abused alcohol and is no longer engaging in such use or abuse is participating in or has successfully completed a supervised drug or alcohol rehabilitation program.

PHA Policy

In determining whether to terminate the lease for illegal drug use or a pattern of illegal drug use, or for abuse or a pattern of abuse of alcohol, by a household member who is no longer engaging in such use or abuse, the PHA will consider whether such household member has successfully completed a supervised drug or alcohol rehabilitation program.

For this purpose the PHA will require the tenant to submit evidence of the household member's successful completion of a supervised drug or alcohol rehabilitation program.

Reasonable Accommodation [24 CFR 966.7]

If the family includes a person with disabilities, the PHA's decision to terminate the family's lease is subject to consideration of reasonable accommodation in accordance with 24 CFR Part 8.

PHA Policy

If a family indicates that the behavior of a family member with a disability is the reason for a proposed termination of lease, the PHA will determine whether the behavior is related to the disability. If so, upon the family's request, the PHA will determine whether alternative measures are appropriate as a reasonable accommodation. The PHA will only consider accommodations that can reasonably be expected to address the behavior that is the basis of the proposed lease termination. See Chapter 2 for a discussion of reasonable accommodation.

Nondiscrimination Limitation [24 CFR 966.4(1)(5)(vii)(F)]

The PHA's eviction actions must be consistent with fair housing and equal opportunity provisions of 24 CFR 5.105.

13-III.F. TERMINATIONS RELATED TO DOMESTIC VIOLENCE, DATING VIOLENCE, SEXUAL ASSAULT, OR STALKING

This section addresses the protections against termination of tenancy that the Violence against Women Act of 2013 (VAWA) provides for public housing residents who are victims of domestic violence, dating violence, sexual assault, or stalking. For general VAWA requirements and PHA policies pertaining to notification, documentation, and confidentiality, see section 16-VII of this ACOP, where definitions of key VAWA terms are also located.

VAWA Protections against Termination [24 CFR 5.2005(c)]

VAWA provides that no person may deny assistance, tenancy, or occupancy rights to public housing to a tenant on the basis or as a direct result of criminal activity directly relating to domestic violence, dating violence, sexual assault, or stalking that is engaged in by a member of the household of the tenant or any guest or other person under the control of the tenant, if the tenant or affiliated individual is the victim or threatened victim of such domestic violence, dating violence, sexual assault, or stalking [FR Notice 8/6/13].

VAWA further provides that incidents of actual or threatened domestic violence, dating violence, sexual assault, or stalking may not be construed either as serious or repeated violations of the lease by the victim or threatened victim of such violence or as good cause for terminating the tenancy or occupancy rights of the victim of such violence [24 CFR 5.2005(c)(1), FR Notice 8/6/13].

Limits on VAWA Protections [24 CFR 5.2005(d) and (e), FR Notice 8/6/13]

While VAWA prohibits a PHA from using domestic violence, dating violence, sexual assault, or stalking as the cause for a termination or eviction action against a public housing tenant who is the victim of the abuse, the protections it provides are not absolute. Specifically:

- VAWA does not limit a PHA's otherwise available authority to terminate assistance to or
 evict a victim for lease violations not premised on an act of domestic violence, dating
 violence, sexual assault, or stalking providing that the PHA does not subject the victim to a
 more demanding standard than the standard to which it holds other tenants.
- VAWA does not limit a PHA's authority to terminate the tenancy of any public housing tenant if the PHA can demonstrate an actual and imminent threat to other tenants or those employed at or providing service to the property if that tenant's tenancy is not terminated.

HUD regulations define *actual and imminent threat* to mean words, gestures, actions, or other indicators of a physical threat that (a) is real, (b) would occur within an immediate time frame, and (c) could result in death or serious bodily harm [24 CFR 5.2005(d)(2) and (e)]. In determining whether an individual would pose an actual and imminent threat, the factors to be considered include:

- The duration of the risk
- The nature and severity of the potential harm
- The likelihood that the potential harm will occur
- The length of time before the potential harm would occur [24 CFR 5.2005(e)]

In order to demonstrate an actual and imminent threat, the PHA must have objective evidence of words, gestures, actions, or other indicators. Even when a victim poses an actual and imminent threat, however, HUD regulations authorize a PHA to terminate the victim's assistance "only when there are no other actions that could be taken to reduce or eliminate the threat, including but not limited to transferring the victim to a different unit, barring the perpetrator from the property, contacting law enforcement to increase police presence or develop other plans to keep the property safe, or seeking other legal remedies to prevent the perpetrator from acting on a threat" [24 CFR 5.2005(d)(3)]. Additionally, HUD regulations state that restrictions "predicated on public safety cannot be based on stereotypes, but must be tailored to particularized concerns about individual residents" [24 CFR 5.2005(d)(3)].

PHA Policy

In determining whether a public housing tenant who is a victim of domestic violence, dating violence, sexual assault, or stalking is an actual and imminent threat to other tenants or those employed at or providing service to a property, the PHA will consider the following, and any other relevant, factors:

Whether the threat is toward an employee or tenant other than the victim of domestic violence, dating violence, sexual assault, or stalking

Whether the threat is a physical danger beyond a speculative threat

Whether the threat is likely to happen within an immediate time frame

Whether the threat to other tenants or employees can be eliminated in some other way, such as by helping the victim relocate to a confidential location, transferring the victim to another unit, or seeking a legal remedy to prevent the perpetrator from acting on the threat

If the tenant wishes to contest the PHA's determination that he or she is an actual and imminent threat to other tenants or employees, the tenant may do so as part of the grievance hearing or in a court proceeding.

Documentation of Abuse [24 CFR 5.2007]

PHA Policy

When an individual facing termination of tenancy for reasons related to domestic violence, dating violence, sexual assault, or stalking claims protection under VAWA, the PHA will request in writing that the individual provide documentation supporting the claim in accordance with the policies in section 16-VII.D of this ACOP.

The PHA reserves the right to waive the documentation requirement if it determines that a statement or other corroborating evidence from the individual will suffice. In such cases the PHA will document the waiver in the individual's file.

Terminating or Evicting a Perpetrator of Domestic Violence

Although VAWA provides protection from termination for victims of domestic violence, it does not provide such protection for perpetrators. In fact, VAWA gives the PHA the explicit authority to bifurcate a lease, or remove a household member from a lease, "in order to evict, remove, or terminate assistance to any individual who is a tenant or lawful occupant of the housing and who engages in criminal activity directly relating to domestic violence, dating violence, sexual assault, or stalking against an affiliated individual or other individual, without evicting, removing, terminating assistance to, or otherwise penalizing a victim of such criminal activity who is also a tenant or lawful occupant of the housing" [FR Notice 8/6/13]. Moreover, HUD regulations impose on the PHA the obligation to consider lease bifurcation in any circumstances involving domestic violence, dating violence, or stalking [24 CFR 966.4(e)(9)].

Specific lease language affirming the PHA's authority to bifurcate a lease is not necessary, and the authority supersedes any local, state, or federal law to the contrary. However, if the PHA chooses to exercise its authority to bifurcate a lease, it must follow any procedures prescribed by HUD or by applicable local, state, or federal law for eviction, lease termination, or termination of assistance. This means that the PHA must follow the same rules when terminating or evicting an individual as it would when terminating or evicting an entire family [FR Notice 3/16/07]. However, perpetrators should be given no more than 30 days' notice of termination in most cases [Notice PIH 2017-08].

PHA Policy

The PHA will bifurcate a family's lease and terminate the tenancy of a family member if the PHA determines that the family member has committed criminal acts of physical violence against other family members or others. This action will not affect the tenancy or program assistance of the remaining, nonculpable family members.

In making its decision, the PHA will consider all credible evidence, including, but not limited to, a signed certification (form HUD-5382) or other documentation of abuse submitted to the PHA by the victim in accordance with this section and section 16-VII.D. The PHA will also consider the factors in section 13.III.E. Upon such consideration, the PHA may, on a case-by-case basis, choose not to bifurcate the lease and terminate the tenancy of the culpable family member.

If the PHA does bifurcate the lease and terminate the tenancy of the culpable family member, it will do so in accordance with the lease, applicable law, and the policies in this ACOP. If the person removed from the lease was the only tenant eligible to receive assistance, the PHA must provide any remaining tenant a chance to establish eligibility for the unit. If the remaining tenant cannot do so, the PHA must provide the tenant reasonable time to find new housing or to establish eligibility for another housing program covered by VAWA 2013.

14-III.C. APPLICABILITY [24 CFR 966.51]

Grievances could potentially address most aspects of a PHA's operation. However, there are some situations for which the grievance procedure is not applicable.

The grievance procedure is applicable only to individual tenant issues relating to the PHA. It is not applicable to disputes between tenants not involving the PHA. Class grievances are not subject to the grievance procedure and the grievance procedure is not to be used as a forum for initiating or negotiating policy changes of the PHA.

If HUD has issued a due process determination, a PHA may exclude from the PHA grievance procedure any grievance concerning a termination of tenancy or eviction that involves:

- Any criminal activity that threatens the health, safety or right to peaceful enjoyment of the premises of other residents or employees of the PHA;
- Any violent or drug-related criminal activity on or off such premises; or
- Any criminal activity that resulted in felony conviction of a household member

In states without due process determinations, PHAs must grant opportunity for grievance hearings for all lease terminations, regardless of cause, with the following exception: PHAs may use expedited grievance procedures for the excluded categories listed above. These expedited grievance procedures are described in Section 14-III.E. below.

If HUD has issued a due process determination, the PHA may evict through the state/local judicial eviction procedures. In this case, the PHA is not required to provide the opportunity for a hearing under the PHA's grievance procedure as described above.

PHA Policy

The PHA is located in a HUD-declared due process state. Therefore, the PHA will not offer grievance hearings for lease terminations involving criminal activity that threatens the health, safety, or right to peaceful enjoyment of the premises of other residents or employees of the PHA, for violent or drug-related criminal activity on or off the premises, or for any criminal activity that resulted in felony conviction of a household member.

See Chapter 13 for related policies on the content of termination notices.

14-III.D. INFORMAL SETTLEMENT OF GRIEVANCE [24 CFR 966.54]

HUD regulations state that any grievance must be personally presented, either orally or in writing, to the PHA office or to the office of the housing development in which the complainant resides so that the grievance may be discussed informally and settled without a hearing.

PHA Policy

The PHA will accept requests for an informal settlement of a grievance either orally or in writing, to the PHA office within 10 business days of the grievable event. Within 10 business days of receipt of the request the PHA will arrange a meeting with the tenant at a mutually agreeable time and confirm such meeting in writing to the tenant.

If a tenant fails to attend the scheduled meeting without prior notice, the PHA will reschedule the appointment only if the tenant can show good cause for failing to appear, or if it is needed as a reasonable accommodation for a person with disabilities.

Good cause is defined as an unavoidable conflict which seriously affects the health, safety or welfare of the family.

HUD regulations require that a summary of such discussion will be prepared within a reasonable time and one copy will be given to the tenant and one retained in the PHA's tenant file.

The summary must specify the names of the participants, dates of meeting, the nature of the proposed disposition of the complaint and the specific reasons therefore, and will specify the procedures by which a hearing may be obtained if the complainant is not satisfied.

PHA Policy

The PHA will prepare a summary of the informal settlement within 5 business days; one copy to be given to the tenant and one copy to be retained in the PHA's tenant file.

For PHAs who have the option to establish an expedited grievance procedure, and who exercise this option, the informal settlement of grievances is not applicable to those grievances for which the expedited grievance procedure applies.

14-III.E. PROCEDURES TO OBTAIN A HEARING

Requests for Hearing and Failure to Request

PHA Policy

The resident must submit a written request for a grievance hearing to the PHA within 5 business days of the tenant's receipt of the summary of the informal settlement.

If the complainant does not request a hearing, the PHA's disposition of the grievance under the informal settlement process will become final. However, failure to request a hearing does not constitute a waiver by the complainant of the right to contest the PHA's action in disposing of the complaint in an appropriate judicial proceeding.

Scheduling of Hearings [24 CFR 966.56(a)]

If the complainant has complied with all requirements for requesting a hearing as described above, a hearing must be scheduled by the hearing officer/panel promptly for a time and place reasonably convenient to both the complainant and the PHA. A written notification specifying the time, place and the procedures governing the hearing must be delivered to the complainant and the appropriate PHA official.

PHA Policy

Within 10 business days of receiving a written request for a hearing, the hearing officer will schedule and send written notice of the hearing to both the complainant and the PHA.

The PHA may wish to permit the tenant to request to reschedule a hearing for good cause.

PHA Policy

The tenant may request to reschedule a hearing for good cause, or if it is needed as a reasonable accommodation for a person with disabilities. Good cause is defined as an unavoidable conflict which seriously affects the health, safety, or welfare of the family. Requests to reschedule a hearing must be made orally or in writing prior to the hearing date. At its discretion, the PHA may request documentation of the "good cause" prior to rescheduling the hearing.

Expedited Grievance Procedure [24 CFR 966.52(a)]

The PHA may establish an expedited grievance procedure for any grievance concerning a termination of tenancy or eviction that involves:

- Any criminal activity that threatens the health, safety, or right to peaceful enjoyment of the premises by other residents or employees of the PHA;
- Any drug-related criminal activity on or near such premises; or
- Any criminal activity that resulted in felony conviction of a household member.

In such expedited grievances, the informal settlement of grievances as discussed in 14-III.D is not applicable.

The PHA may adopt special procedures concerning expedited hearings, including provisions for expedited notice or scheduling, or provisions for expedited decision on the grievance.

PHA Policy

The PHA will not offer expedited grievance procedures.

PART I: PREVENTING, DETECTING, AND INVESTIGATING ERRORS AND PROGRAM ABUSE

15-I.A. PREVENTING ERRORS AND PROGRAM ABUSE

HUD created the Enterprise Income Verification (EIV) system to provide PHAs with a powerful tool for preventing errors and program abuse. PHAs are required to use the EIV system in its entirety in accordance with HUD administrative guidance [24 CFR 5.233]. PHAs are further required to:

- Provide applicants and residents with form HUD-52675, "Debts Owed to PHAs and Terminations"
- Require all adult members of an applicant or participant family to acknowledge receipt of form HUD-52675 by signing a copy of the form for retention in the family file

PHA Policy

The PHA anticipates that the vast majority of families and PHA employees intend to and will comply with program requirements and make reasonable efforts to avoid errors.

To ensure that the PHA's program is administered effectively and according to the highest ethical and legal standards, the PHA will employ a variety of techniques to ensure that both errors and intentional program abuse are rare.

The PHA will provide each applicant and resident with a copy of "Is Fraud Worth It?" (form HUD-1141-OIG), which explains the types of actions a family must avoid and the penalties for program abuse.

The PHA will provide each applicant and resident with a copy of "What You Should Know about EIV," a guide to the Enterprise Income Verification (EIV) system published by HUD as an attachment to Notice PIH 2017-12. In addition, the PHA will require the head of each household to acknowledge receipt of the guide by signing a copy for retention in the family file.

The PHA will require mandatory orientation sessions for all prospective residents either prior to or upon execution of the lease. The PHA will discuss program compliance and integrity issues. At the conclusion of all program orientation sessions, the family representative will be required to sign a program briefing certificate to confirm that all rules and pertinent regulations were explained to them.

The PHA will routinely provide resident counseling as part of every reexamination interview in order to clarify any confusion pertaining to program rules and requirements.

PHA staff will be required to review and explain the contents of all HUD- and PHA-required forms prior to requesting family member signatures.

The PHA will place a warning statement about the penalties for fraud (as described in 18 U.S.C. 1001 and 1010) on key PHA forms and form letters that request information from a family member.

The PHA will provide each PHA employee with the necessary training on program rules and the organization's standards of conduct and ethics.

At every regular reexamination the PHA staff will explain any changes in HUD regulations or PHA policy that affect residents.

For purposes of this chapter the term *error* refers to an unintentional error or omission. Program abuse or fraud refers to a single act or pattern of actions that constitute a false statement, omission, or concealment of a substantial fact, made with the intent to deceive or mislead.

15-I.B. DETECTING ERRORS AND PROGRAM ABUSE

In addition to taking steps to prevent errors and program abuse, the PHA will use a variety of activities to detect errors and program abuse.

Quality Control and Analysis of Data

PHA Policy

The PHA will employ a variety of methods to detect errors and program abuse, including:

The PHA routinely will use EIV and other non-HUD sources of up-front income verification. This includes the Work Number and any other private or public databases available to the PHA.

At each annual reexamination, current information provided by the family will be compared to information provided at the last annual reexamination to identify inconsistencies and incomplete information.

The PHA will compare family-reported income and expenditures to detect possible unreported income.

Independent Audits and HUD Monitoring

Notice PIH 2015-16 requires all PHAs that expend \$750,000 or more in federal awards annually to have an independent audit (IPA). In addition, HUD conducts periodic on-site and automated monitoring of PHA activities and notifies the PHA of errors and potential cases of program abuse.

PHA Policy

The PHA will use the results reported in any IPA or HUD monitoring reports to identify potential program abuses as well as to assess the effectiveness of the PHA's error detection and abuse prevention efforts.

Individual Reporting of Possible Errors and Program Abuse

PHA Policy

The PHA will encourage staff, residents, and the public to report possible program abuse.

Chapter 16

PROGRAM ADMINISTRATION

INTRODUCTION

This chapter discusses administrative policies and practices that are relevant to the activities covered in this ACOP. The policies are discussed in seven parts as described below:

<u>Part I: Setting Utility Allowances</u>. This part describes how utility allowances are established and revised. Also discussed are the requirements to establish surcharges for excess consumption of PHA-furnished utilities.

<u>Part II: Establishing Flat Rents</u>. This part describes the requirements and policies related to establishing and updating flat rent amounts.

<u>Part III:</u> Repayment of Family <u>Debts</u>. This part contains policies for recovery of monies that have been underpaid by families, and describes the circumstances under which the PHA will offer repayment agreements to families. Also discussed are the consequences for failure to make payments in accordance with a repayment agreement.

<u>Part IV: Public Housing Assessment System (PHAS)</u>. This part describes the PHAS indicators, how PHAs are scored under PHAS, and how those scores affect a PHA.

<u>Part V: Record Keeping</u>. All aspects of the program involve certain types of record-keeping. This part outlines the privacy rights of applicants and participants and record retention policies the PHA will follow.

<u>Part VI: Reporting and Record Keeping for Children with Elevated Blood Lead Level</u>. This part describes the PHA's reporting responsibilities related to children with elevated blood lead levels that are living in public housing.

Part VII: Violence against Women Act (VAWA): Notification, Documentation, and Confidentiality. This part contains key terms used in VAWA and describes requirements related to notifying families about their rights and responsibilities under VAWA; requesting documentation from victims of domestic violence, dating violence, sexual assault, and stalking; and maintaining the confidentiality of information obtained from victims.

PART II: ESTABLISHING FLAT RENTS

16-II.A. OVERVIEW

Flat rents are designed to encourage self-sufficiency and to avoid creating disincentives for continued residency by families who are attempting to become economically self-sufficient.

Flat rents are also used to prorate assistance for a mixed family. A mixed family is one whose members include those with citizenship or eligible immigration status, and those without citizenship or eligible immigrations status [24 CFR 5.504].

This part discusses how the PHA establishes and updates flat rents. Policies related to the use of flat rents, family choice of rent, flat rent hardships, and proration of rent for a mixed family are discussed in Chapter 6.

16-II.B. FLAT RENTS [24 CFR 960.253(b) and Notice PIH 2017-23]

Establishing Flat Rents

The 2015 Appropriations Act requires that flat rents must be set at no less than 80 percent of the applicable fair market rent (FMR). Alternatively, the PHA may set flat rents at no less than 80 percent of the applicable small area FMR(SAFMR) for metropolitan areas, or 80 percent of the applicable unadjusted rents for nonmetropolitan areas.

For areas where HUD has not determined a SAFMR or an unadjusted rent, PHAs must set flat rents at no less than 80 percent of the FMR or apply for an exception flat rent.

The 2015 Appropriations Act permits PHAs to request an exception flat rent that is lower than either 80 percent of the FMR or SAFMR/unadjusted rent if the PHA can demonstrate that these FMRs do not reflect the market value of a particular property or unit.

In order to demonstrate the need for an exception flat rent, PHAs are required to submit a market analysis methodology that demonstrates the value of the unit. The PHA must use HUD's rent reasonableness methodology to determine flat rents. In determining flat rents, PHAs must consider the following:

- Location
- Quality
- Unit size
- Unit type
- Age of the unit
- Amenities at the property and in immediate neighborhood
- Housing services provided
- Maintenance provided by the PHA
- Utilities provided by the PHA and/or landlord for (comparable units in the market study)

The PHA must provide a corresponding key explaining the calculations used for determining the valuation for each factor.

PHAs must receive written HUD approval before implementing exception flat rents. PHAs with a previously approved flat rent exception request may submit a written request to extend the approved flat rents for up to two additional years, provided local market conditions remain unchanged. Detailed information on how to request exception flat rents can be found in Notice PIH 2017-23.

PHAs are now required to apply a utility allowance to flat rents as necessary. Flat rents set at 80 percent of the FMR must be reduced by the amount of the unit's utility allowance, if any.

Review of Flat Rents

No later than 90 days after the effective date of the new annual FMRs/SAFMRs/unadjusted rent, PHAs must implement new flat rents as necessary based changes to the FMR/SAFMR/unadjusted rent or request an exception.

If the FMR falls from year to year, the PHA may, but is not required to, lower the flat rent to 80 percent of the current FMR/SAFMR/unadjusted rent.

PHA Policy

If the FMR/SAFMR/unadjusted rent is lower than the previous year, the PHA will reduce flat rents to 80 percent of the current FMR/SAFMR.

Posting of Flat Rents

PHA Policy

The PHA will publicly post the schedule of flat rents in a conspicuous manner in the applicable PHA or project office.

Documentation of Flat Rents [24 CFR 960.253(b)(5)]

The PHA must maintain records that document the method used to determine flat rents, and that show how flat rents were determined by the PHA in accordance with this method.

PART III: FAMILY DEBTS TO THE PHA

16-III.A. OVERVIEW

This part describes the PHA's policies for recovery of monies owed to the PHA by families.

PHA Policy

When an action or inaction of a resident family results in the underpayment of rent or other amounts, the PHA holds the family liable to return any underpayments to the PHA.

The PHA will enter into repayment agreements in accordance with the policies contained in this part as a means to recover overpayments.

When a family refuses to repay monies owed to the PHA, the PHA will utilize other available collection alternatives including, but not limited to, the following:

Collection agencies

Small claims court

Civil law suit

State income tax set-off program

16-III.B. REPAYMENT POLICY

Family Debts to the PHA

PHA Policy

Any amount owed to the PHA by a public housing family must be repaid. If the family is unable to repay the debt within 30 days, the PHA will offer to enter into a repayment agreement in accordance with the policies below.

If the family refuses to repay the debt, does not enter into a repayment agreement, or breaches a repayment agreement, the PHA will terminate the family's tenancy in accordance with the policies in Chapter 13. The PHA will also pursue other modes of collection.

General Repayment Agreement Guidelines

Down Payment Requirement

PHA Policy

Before executing a repayment agreement with a family, the PHA will generally require a down payment of 10 percent of the total amount owed. If the family can provide evidence satisfactory to the PHA that a down payment of 10 percent would impose an undue hardship, the PHA may, in its sole discretion, require a lesser percentage or waive the requirement.

Payment Thresholds

Notice PIH 2017-12 recommends that the total amount that a family must pay each month—the family's monthly share of rent plus the monthly debt repayment amount—should not exceed 40 percent of the family's monthly adjusted income, which is considered "affordable." Moreover, Notice PIH 2017-12 acknowledges that PHAs have the discretion to establish "thresholds and policies" for repayment agreements with families [24 CFR 982.552(c)(1)(vii)].

PHA Policy

If a family is paying less than 40 percent of its monthly adjusted income (MAI) in rent, the minimum monthly payment amount will be the greater of the following two amounts:

The difference between 40 percent of the family's MAI and the TTP at the time the agreement is executed

\$25

If a family can provide evidence satisfactory to the PHA that a monthly payment amount of \$25 would impose an undue hardship, the PHA may, in its sole discretion, require a lower monthly payment amount.

If the family's income increases or decreases during the term of a repayment agreement, either the PHA or the family may request that the monthly payment amount be adjusted accordingly.

Execution of the Agreement

PHA Policy

Any repayment agreement between the PHA and a family must be signed and dated by the PHA and by the head of household and spouse/cohead (if applicable).

Due Dates

PHA Policy

All payments are due by the close of business on the 15th day of the month. If the 15th does not fall on a business day, the due date is the close of business on the first business day after the 15th.

Late or Missed Payments

PHA Policy

If a payment is not received by the end of the business day on the date due, and prior approval for the missed payment has not been given by the PHA, the PHA will send the family a delinquency notice giving the family 10 business days to make the late payment. If the payment is not received by the due date of the delinquency notice, it will be considered a breach of the agreement and the PHA will terminate tenancy in accordance with the policies in Chapter 13.

If a family receives three delinquency notices for unexcused late payments in a 12-month period, the repayment agreement will be considered in default, and the PHA will terminate tenancy in accordance with the policies in Chapter 13.

No Offer of Repayment Agreement

PHA Policy

The PHA generally will not enter into a repayment agreement with a family if there is already a repayment agreement in place with the family, or if the amount owed by the family exceeds the federal or state threshold for criminal prosecution.

Repayment Agreements Involving Improper Payments

Notice PIH 2017-12 requires certain provisions to be included in any repayment agreement involving amounts owed by a family because it underreported or failed to report income:

- A reference to the items in the public housing lease that state the family's obligation to provide true and complete information at every reexamination and the grounds on which the PHA may terminate assistance because of a family's action or failure to act
- A statement clarifying that each month the family not only must pay to the PHA the monthly payment amount specified in the agreement but must also pay to the PHA the monthly tenant rent
- A statement that the terms of the repayment agreement may be renegotiated if the family's income decreases or increases
- A statement that late or missed payments constitute default of the repayment agreement and may result in termination of tenancy

PART VI: REPORTING REQUIREMENTS FOR CHILDREN WITH ELEVATED BLOOD LEAD LEVEL

16-VI.A. REPORTING REQUIREMENTS [24 CFR 35.1130(e); Notice PIH 2017-13]

The PHA has certain responsibilities relative to children with elevated blood lead levels that are living in public housing.

The PHA must report the name and address of a child identified as having an elevated blood lead level (EBLL) to the public health department within five business days of being so notified by any other medical health care professional. The PHA must also report each known case of a child with an EBLL to the HUD field office.

PHA Policy

The PHA will provide the public health department written notice of the name and address of any child identified as having an elevated blood lead level.

The PHA will provide written notice of each known case of a child with an EBLL to the HUD field office, and to HUD's Office of Lead Hazard Control (OLHCHH), within five business days of receiving the information.

ADMISSIONS AND CONTINUED OCCUPANCY POLICY

FOR THE

PUBLIC HOUSING PROGRAM

May 1, 2005

Revision Date	
February 1, 2006	June 1, 2013
October 1, 2006	July 1, 2014
June 1, 2007	June 1, 2015
September 1, 2008	March 1, 2016
September 1, 2009	March 1, 2017
August 1, 2010	July 1, 2017
June 1, 2011	July 1, 2018
June 1, 2012	June 1, 2019

Approved by the PHA Board of Commissioners:

Submitted to HUD:

Copyright 2019 by Nan McKay & Associates, Inc.
All rights reserved
Permission to reprint granted only to the Public Housing Authority that has purchased this plan from Nan McKay & Associates, Inc. This document may not be reprinted or distributed to any other person or entity other than the purchasing agency without the express written permission of Nan McKay & Associates, Inc.

Introduction ABOUT THE MODEL ACOP

HOW	TO USE THE MODEL ACOP	Intro-i
ABOU	UT THE MODEL ACOP AND THE PUBLIC HOUSING LEASE	Intro-ii
REFE	RENCES CITED IN THE MODEL ACOP	Intro-ii
RESO	URCES CITED IN THE MODEL ACOP	Intro-iii
	Chapter 1	
	OVERVIEW OF THE PROGRAM AND PLAN	
INTR	ODUCTION	1-1
	PART I: THE PHA	
I.A.	OVERVIEW	1-1
I.B.	ORGANIZATION AND STRUCTURE OF THE PHA	
I.C.	PHA MISSION	1-3
I.D.	THE PHA'S COMMITMENT TO ETHICS AND SERVICE	1-4
	PART II: THE PUBLIC HOUSING PROGRAM	
II.A.	OVERVIEW AND HISTORY OF THE PROGRAM	1-5
II.B.	PUBLIC HOUSING PROGRAM BASICS	1-6
II.C.	PUBLIC HOUSING PARTNERSHIPS	1-6
II.D.	APPLICABLE REGULATIONS	1-10
	PART III: THE ADMISSIONS AND CONTINUED OCCUPANCY PO	OLICIES
III.A.	OVERVIEW AND PURPOSE OF THE POLICY	1-11
	CONTENTS OF THE POLICY	
III.C.	UPDATING AND REVISING THE POLICY	1-12

Chapter 2 FAIR HOUSING AND EQUAL OPPORTUNITY

INTRO	ODUCTION	2-1
	PART I: NONDISCRIMINATION	
I.A.	OVERVIEW	2-3
I.B.	NONDISCRIMINATION	
	PART II: POLICIES RELATED TO PERSONS WITH DISABILITIES	
II.A.	OVERVIEW	2-7
II.B.	DEFINITION OF REASONABLE ACCOMMODATION	
II.C.	REQUEST FOR AN ACCOMMODATION	2-9
II.D.	VERIFICATION OF DISABILITY	
II.E.	APPROVAL/DENIAL OF A REQUESTED ACCOMMODATION	2-11
II.F.	PROGRAM ACCESSIBILITY FOR PERSONS WITH HEARING OR	
	VISION IMPAIRMENTS	2-12
II.G.	PHYSICAL ACCESSIBILITY	2-13
II.H.	DENIAL OR TERMINATION OF ASSISTANCE	2-14
	PART III: IMPROVING ACCESS TO SERVICES FOR PERSONS WITH LIMITED ENGLISH PROFICIENCY (LEP)	
III.A.	OVERVIEW	2-15
III.B.	ORAL INTERPRETATION	2-16
III.C.	WRITTEN TRANSLATION	2-16
III.D.	IMPLEMENTATION PLAN	2-17
	EXHIBITS	
2-1:	DEFINITION OF A PERSON WITH A DISABILITY UNDER FEDERAL CIVIL RIGHTS LAWS	2-19

Chapter 3 ELIGIBILITY

INTR	ODUCTION	3-1
	PART I: DEFINITIONS OF FAMILY AND HOUSEHOLD MEMBERS	
I.A.	OVERVIEW	3-3
I.B.	FAMILY AND HOUSEHOLD	
I.C.	FAMILY BREAK-UP AND REMAINING MEMBER OF TENANT FAMILY	
I.D.	HEAD OF HOUSEHOLD	3-5
I.E.	SPOUSE, COHEAD, AND OTHER ADULT	3-5
I.F.	DEPENDENT	3-6
I.G.	FULL-TIME STUDENT	3-6
I.H.	ELDERLY AND NEAR-ELDERLY PERSONS, AND ELDERLY FAMILY	3-7
I.I.	PERSONS WITH DISABILITIES AND DISABLED FAMILY	3-7
I.J.	GUESTS	3-8
I.K.	FOSTER CHILDREN AND FOSTER ADULTS	3-9
I.L.	ABSENT FAMILY MEMBERS	3-10
I.M.	LIVE-IN AIDE	3-12
	PART II: BASIC ELIGIBILITY CRITERIA	
II.A.	INCOME ELIGIBILITY AND TARGETING	3-13
II.B.	CITIZENSHIP OR ELIGIBLE IMMIGRATION STATUS	3-15
II.C.	SOCIAL SECURITY NUMBERS	3-17
II.D.	FAMILY CONSENT TO RELEASE OF INFORMATION	3-18
	PART III: DENIAL OF ADMISSION	
III.A.	OVERVIEW	3-19
III.B.	REQUIRED DENIAL OF ADMISSION	3-19
III.C.	OTHER PERMITTED REASONS FOR DENIAL OF ADMISSION	3-21
III.D.	SCREENING	3-23
III.E.		3-28
III.F.	PROHIBITION AGAINST DENIAL OF ASSISTANCE TO VICTIMS	
	OF DOMESTIC VIOLENCE, DATING VIOLENCE, SEXUAL ASSAULT,	
	OR STALKING	
III.G.	NOTICE OF ELIGIBILITY OR DENIAL	3-32
	EXHIBITS	
3-1:	DETAILED DEFINITIONS RELATED TO DISABILITIES	3-33

Chapter 4 APPLICATIONS, WAITING LIST AND TENANT SELECTION

INTRODUCTION		4-1
	PART I: THE APPLICATION PROCESS	
I.A.	OVERVIEW	4-3
I.B.	APPLYING FOR ASSISTANCE	
I.C.	ACCESSIBILITY OF THE APPLICATION PROCESS	
I.D.	PLACEMENT ON THE WAITING LIST	
	PART II: MANAGING THE WAITING LIST	
II.A.	OVERVIEW	<i>1</i> ₋ 7
II.B.	ORGANIZATION OF THE WAITING LIST	4-7 4 ₋ 7
II.C.	OPENING AND CLOSING THE WAITING LIST	
II.D.	FAMILY OUTREACH	
II.E.	REPORTING CHANGES IN FAMILY CIRCUMSTANCES	
II.F.	UPDATING THE WAITING LIST	
	PART III: TENANT SELECTION	
ΙΙΙ Δ	OVERVIEW	<i>A</i> ₋ 15
III.B.		-
	NOTIFICATION OF SELECTION	
	THE APPLICATION INTERVIEW	
	FINAL ELIGIBILITY DETERMINATION	
	Chapter 5	
	OCCUPANCY STANDARDS AND UNIT OFFERS	
INTR	ODUCTION	5-1
	PART I: OCCUPANCY STANDARDS	
I.A.	OVERVIEW	5-1
I.B.	DETERMINING UNIT SIZE	5-2
I.C.	EXCEPTIONS TO OCCUPANCY STANDARDS	5-4
	PART II: UNIT OFFERS	
II.A.	OVERVIEW	5-5
II.B.	NUMBER OF OFFERS	5-5
II.C.	TIME LIMIT FOR UNIT OFFER ACCEPTANCE OR REFUSAL	5-6
II.D.	REFUSALS OF UNIT OFFERS	5-6
II.E.	ACCESSIBLE UNITS	5-8
II.F.	DESIGNATED HOUSING	5-8

Chapter 6 INCOME AND RENT DETERMINATIONS

INTRO	ODUCTION	6-1
	PART I: ANNUAL INCOME	
I.A.	OVERVIEW	6-3
I.B.	HOUSEHOLD COMPOSITION AND INCOME	
I.C.	ANTICIPATING ANNUAL INCOME	
I.D.	EARNED INCOME	6-9
I.E.	EARNED INCOME DISALLOWANCE	6-13
I.F.	BUSINESS INCOME	6-17
I.G.	ASSETS	
I.H.	PERIODIC PAYMENTS	
I.I.	PAYMENTS IN LIEU OF EARNINGS	6-29
I.J.	WELFARE ASSISTANCE	
I.K.	PERIODIC AND DETERMINABLE ALLOWANCES	
I.L.	ADDITIONAL EXCLUSIONS FROM ANNUAL INCOME	6-32
	PART II: ADJUSTED INCOME	
II.A.	INTRODUCTION	
II.B.	DEPENDENT DEDUCTION	
II.C.	ELDERLY OR DISABLED FAMILY DEDUCTION	
II.D.	MEDICAL EXPENSES DEDUCTION DISABILITY ASSISTANCE EXPENSES DEDUCTION	
II.E.	CHILD CARE EXPENSE DEDUCTION	
II.F. II.G.	PERMISSIVE DEDUCTIONS	
II.G.	PERMISSIVE DEDUCTIONS	0-40
	PART III: CALCULATING RENT	
III.A.	OVERVIEW OF INCOME-BASED RENT CALCULATIONS	6-47
	FINANCIAL HARDSHIPS AFFECTING MINIMUM RENT	
III.C.	UTILITY ALLOWANCES	
III.D.	PRORATED RENT FOR MIXED FAMILIES	6-56
III.E.		
	EXHIBITS	
6-1:	ANNUAL INCOME INCLUSIONS	
6-2:	ANNUAL INCOME EXCLUSIONS	
6-3:	TREATMENT OF FAMILY ASSETS	
6-4:	EARNED INCOME DISALLOWANCE	
6-5:	THE EFFECT OF WELFARE BENEFIT REDUCTION	6-69

Chapter 7 VERIFICATION

INTRO	ODUCTION	7-1
	PART I: GENERAL VERIFICATION REQUIREMENTS	
I.A.	FAMILY CONSENT TO RELEASE OF INFORMATION	7-1
I.B.	OVERVIEW OF VERIFICATION REQUIREMENTS	
I.C.	UP-FRONT INCOME VERIFICATION (UIV)	
I.D.	THIRD-PARTY WRITTEN AND ORAL VERIFICATION	7-6
I.E.	SELF-CERTIFICATION	
	PART II: VERIFYING FAMILY INFORMATION	
II.A.	VERIFICATION OF LEGAL IDENTITY	7-11
II.B.	SOCIAL SECURITY NUMBERS	7-12
II.C.	DOCUMENTATION OF AGE	7-13
II.D.	FAMILY RELATIONSHIPS	7-14
II.E.	VERIFICATION OF STUDENT STATUS	7-15
II.F.	DOCUMENTATION OF DISABILITY	7-16
II.G.	CITIZENSHIP OR ELIGIBLE IMMIGRATION STATUS	7-17
II.H.	VERIFICATION OF PREFERENCE STATUS	7-18
	PART III: VERIFYING INCOME AND ASSETS	
III.A.	EARNED INCOME	7-19
III.B.	BUSINESS AND SELF EMPLOYMENT INCOME	7-19
III.C.	PERIODIC PAYMENTS AND PAYMENTS IN LIEU OF EARNINGS	
III.D.	ALIMONY OR CHILD SUPPORT	
III.E.	ASSETS AND INCOME FROM ASSETS	7-22
III.F.	NET INCOME FROM RENTAL PROPERTY	
III.G.	RETIREMENT ACCOUNTS	7-23
III.H.	INCOME FROM EXCLUDED SOURCES	7-24
III.I.	ZERO ANNUAL INCOME STATUS	7-24
	PART IV: VERIFYING MANDATORY DEDUCTIONS	
IV.A.	DEPENDENT AND ELDERLY/DISABLED HOUSEHOLD DEDUCTIONS	7-25
IV.B.	MEDICAL EXPENSE DEDUCTION	7-26
	DISABILITY ASSISTANCE EXPENSES	
IV.D.	CHILD CARE EXPENSES	7-30
	EXHIBITS	
7-1:	SUMMARY OF DOCUMENTATION REQUIREMENTS	
	FOR NONCITIZENS	7-33

Chapter 8 LEASING AND INSPECTIONS

INTR	ODUCTION	8-1
	PART I: LEASING	
I.A.	OVERVIEW	8-1
I.B.	LEASE ORIENTATION	8-2
I.C.	EXECUTION OF LEASE	8-3
I.D.	MODIFICATIONS TO THE LEASE	8-4
I.E.	SECURITY DEPOSITS	8-6
I.F.	PAYMENTS UNDER THE LEASE	8-7
I.G	MINIMUM HEATING STANDARDS	8-10
	PART II: INSPECTIONS	
II.A.	OVERVIEW	8-13
II.B.		8-13
II.C.	NOTICE AND SCHEDULING OF INSPECTIONS	8-15
II.D.	INSPECTION RESULTS	8-16
	EXHIBITS	
8-1:	SMOKE-FREE POLICY	8-19

Chapter 9 REEXAMINATIONS

INTRO	ODUCTION	9-1
	PART I: ANNUAL REEXAMINATIONS FOR FAMILIES PAYING	
	INCOME-BASED RENTS	
I.A.	OVERVIEW	9-3
I.B.	STREAMLINED ANNUAL REEXAMINATIONS	9-4
I.C.	SCHEDULING ANNUAL REEXAMINATIONS	9-5
I.D.	CONDUCTING ANNUAL REEXAMINATIONS	9-6
I.E.	EFFECTIVE DATES	9-8
	PART II: REEXAMINATIONS FOR FAMILIES PAYING FLAT RENTS	
II.A.	OVERVIEW	9-9
II.B.	FULL REEXAMINATION OF FAMILY INCOME AND COMPOSITION	9-9
II.C.	REEXAMINATION OF FAMILY COMPOSITION ("ANNUAL UPDATE")	9-10
	PART III: INTERIM REEXAMINATIONS	
III.A.	OVERVIEW	9-13
III.B.	CHANGES IN FAMILY AND HOUSEHOLD COMPOSITION	9-13
III.C.	CHANGES AFFECTING INCOME OR EXPENSES	9-15
III.D.	PROCESSING THE INTERIM REEXAMINATION	9-17
	PART IV: RECALCULATING TENANT RENT	
	OVERVIEW	
IV.B.	CHANGES IN UTILITY ALLOWANCES	9-19
IV.C.	NOTIFICATION OF NEW TENANT RENT	9-19
IV.D.	DISCREPANCIES	9-20

Chapter 10 PETS

INTRO	ODUCTION	10-1
	PART I: SERVICE ANIMALS AND ASSISTANCE AN	IMALS
I.A.	OVERVIEW	10-3
I.B.	APPROVAL OF ASSISTANCE ANIMALS	
I.C.	CARE AND HANDLING	
	PART II: PET POLICIES FOR ALL DEVELOPME	NTS
II.A.	OVERVIEW	10-7
II.B.	MANAGEMENT APPROVAL OF PETS	
II.C.	STANDARDS FOR PETS	10-9
II.D.	PET RULES.	10-11
	PART III: PET DEPOSITS AND FEES IN	
	ELDERLY/DISABLED DEVELOPMENTS	
III.A.	OVERVIEW	10-17
III.B.	PET DEPOSITS	10-17
III.C.	OTHER CHARGES	10-18
	PART IV: PET DEPOSITS AND FEES IN GENER	AL
	OCCUPANCY DEVELOPMENTS	
	OVERVIEW	
	PET DEPOSITS	
	NON-REFUNDABLE NOMINAL PET FEE	
IV.D.	OTHER CHARGES	10-21

Chapter 11 COMMUNITY SERVICE

INTR	ODUCTION	11-1
	PART I: COMMUNITY SERVICE REQUIREMENT	
I.A.	OVERVIEW	11-1
I.B.	REQUIREMENTS	11-2
I.C.	DETERMINATION OF EXEMPTION STATUS AND COMPLIANCE	11-7
I.D.	DOCUMENTATION AND VERIFICATION	11-11
I.E.	NONCOMPLIANCE	11-13
II.A.	PART II: IMPLEMENTATION OF COMMUNITY SERVICE OVERVIEW	11-17
	EXHIBITS	
11-1:	COMMUNITY SERVICE AND SELF-SUFFICIENCY POLICY	11-19
11-2:	DEFINITION OF A PERSON WITH A DISABILITY UNDER SOCIAL	
	SECURITY ACTS 216(i)(l) and Section 1416(excerpt) FOR PURPOSES	
	OF EXEMPTION FROM COMMUNITY SERVICE	11-25
11-3:	PHA DETERMINATION OF EXEMPTION FOR COMMUNITY SERVIO	CE 11-27
11-4	CSSR WORK-OUT AGREEMENT	11-29

Chapter 12 TRANSFER POLICY

INTR	INTRODUCTION	
	PART I: EMERGENCY TRANSFERS	
I.A.	OVERVIEW	12-1
I.B.	EMERGENCY TRANSFERS	12-2
I.C.	EMERGENCY TRANSFER PROCEDURES	12-3
I.D.	COSTS OF TRANSFER	12-3
	PART II: PHA REQUIRED TRANSFERS	
II.A.	OVERVIEW	12-5
II.B.	TYPES OF PHA REQUIRED TRANSFERS	12-5
II.C.	ADVERSE ACTION	12-8
II.D.	COST OF TRANSFER	12-8
	PART III: TRANSFERS REQUESTED BY TENANTS	5
III.A.	OVERVIEW	12-9
III.B.	TYPES OF RESIDENT REQUESTED TRANSFERS	12-9
III.C.	ELIGIBILITY FOR TRANSFER	12-10
III.D.	SECURITY DEPOSITS	12-11
III.E.	COST OF TRANSFER	12-11
III.F.	HANDLING OF REQUESTS	12-12
	PART IV: TRANSFER PROCESSING	
IV.A.	OVERVIEW	12-13
IV.B.	TRANSFER LIST	
IV.C.	TRANSFER OFFER POLICY	
IV.D.	GOOD CAUSE FOR UNIT REFUSAL	12-14
IV.E.	DECONCENTRATION	
IV.F.	REEXAMINATION POLICIES FOR TRANSFERS	12-15

Chapter 13 LEASE TERMINATIONS

INTR	INTRODUCTION	
	PART I: TERMINATION BY TENANT	
I.A.	TENANT CHOOSES TO TERMINATE THE LEASE	13-3
	PART II: TERMINATION BY PHA – MANDATORY	
II.A.	OVERVIEW	13-5
II.B.	FAILURE TO PROVIDE CONSENT	13-5
II.C.	FAILURE TO DOCUMENT CITIZENSHIP	
II.D.	FAILURE TO DISCLOSE AND DOCUMENT SOCIAL SECURITY NUMBERS	13-6
II.E.	FAILURE TO ACCEPT THE PHA'S OFFER OF A LEASE REVISION	13-6
II.F.	METHAMPHETAMINE CONVICTION	13-7
II.G	LIFETIME REGISTERED SEX OFFENDERS	13-7
II.H.	NONCOMPLIANCE WITH COMMUNITY SERVICE REQUIREMENTS	13-7
II.I.	DEATH OF A SOLE FAMILY MEMBER	13-7
	PART III: TERMINATION BY PHA – OTHER AUTHORIZED REASONS	
III.A.	OVERVIEW	13-9
III.B.		
III.C.		
III.D.	ALTERNATIVES TO TERMINATION OF TENANCY	13-20
III.E.	CRITERIA FOR DECIDING TO TERMINATE TENANCY	13-21
III.F.	TERMINATIONS RELATED TO DOMESTIC VIOLENCE,	
	DATING VIOLENCE, SEXUAL ASSAULT, OR STALKING	13-25
	PART IV: NOTIFICATION REQUIREMENTS, EVICTION	
	PROCEDURES AND RECORD KEEPING	
IV.A.	OVERVIEW	13-29
IV.B.	CONDUCTING CRIMINAL RECORDS CHECKS	13-29
	DISCLOSURE OF CRIMINAL RECORDS TO FAMILY	
IV.D.	LEASE TERMINATION NOTICE	13-31
	EVICTION	
	NOTIFICATION TO POST OFFICE	
	RECORD KEEPING	

Chapter 14 GRIEVANCES AND APPEALS

INTR	ODUCTION	14-1
	PART I: INFORMAL HEARINGS FOR PUBLIC HOUSING APPLIC	CANTS
I.A.	OVERVIEW	14-1
I.B.	INFORMAL HEARING PROCESS	14-2
	PART II: INFORMAL HEARINGS WITH REGARD TO NONCITIZ	ZENS
II.A.	HEARING AND APPEAL PROVISIONS FOR NONCITIZENS	14-5
P	ART III: GRIEVANCE PROCEDURES FOR PUBLIC HOUSING RES	SIDENTS
III.A.	REQUIREMENTS	14-9
III.B.		
	APPLICABILITY	
	INFORMAL SETTLEMENT OF GRIEVANCE	
III.E.		
III.F.		
	PROCEDURES GOVERNING THE HEARING	
III.H.	DECISION OF THE HEARING OFFICER	14-20
	EXHIBITS	
14-1:	SAMPLE GRIEVANCE PROCEDURE	14-23
	Chapter 15	
	PROGRAM INTEGRITY	
INTR	ODUCTION	15-1
	PART I: PREVENTING, DETECTING, AND	
	INVESTIGATING ERRORS AND PROGRAM ABUSE	
I.A.	PREVENTING ERRORS AND PROGRAM ABUSE	15-3
I.B.	DETECTING ERRORS AND PROGRAM ABUSE	15-4
I.C.	INVESTIGATING ERRORS AND PROGRAM ABUSE	15-5
	PART II: CORRECTIVE MEASURES AND PENALTIES	
II.A.	UNDER- OR OVERPAYMENT	15-7
II.B.	FAMILY-CAUSED ERRORS AND PROGRAM ABUSE	
II.C.	PHA-CAUSED ERRORS OR PROGRAM ABUSE	
II.D.	CRIMINAL PROSECUTION	15-12
II.E.	FRAUD AND PROGRAM ABUSE RECOVERIES	15-12

Chapter 16 PROGRAM ADMINISTRATION

INTR	INTRODUCTION	
	PART I: SETTING UTILITY ALLOWANCES	
I.A.	OVERVIEW	16-3
I.B	UTILITY ALLOWANCES	16-3
I.C.	SURCHARGES FOR PHA-FURNISHED UTILITIES	16-5
I.D.	NOTICE REQUIREMENTS	16-5
I.E.	REASONABLE ACCOMMODATION	16-6
	PART II: ESTABLISHING FLAT RENTS	
II.A.	OVERVIEW	16-7
II.B.	FLAT RENTS	16-7
	PART III: FAMILY DEBTS TO THE PHA	
III.A.	OVERVIEW	16-9
III.B.	REPAYMENT POLICY	16-10
	PART IV: PUBLIC HOUSING ASSESSMENT SYSTEM (PHAS)	
IV.A.	OVERVIEW	16-13
IV.B.	PHAS INDICATORS	16-13
IV.C.	PHAS SCORING	16-15
	PART V: RECORD KEEPING	
V.A.	OVERVIEW	16-17
V.B.	RECORD RETENTION	16-17
V.C.	RECORDS MANAGEMENT	16-18
	PART VI: REPORTING REQUIREMENTS FOR CHILDREN WITH ELEVATED BLOOD LEAD LEVEL	
VI.A.	REPORTING REQUIREMENTS	16-21

PART VII: VIOLENCE AGAINST WOMEN ACT (VAWA): NOTIFICATION, DOCUMENTATION, AND CONFIDENTIALITY

VII.A.	OVERVIEW	16-23
VII.B.	DEFINITIONS	16-23
VII.C.	NOTIFICATION	16-24
	DOCUMENTATION	
VII.E.	CONFIDENTIALITY	16-28
	TW/HYDIRG	
	EXHIBITS	
16-1:	SAMPLE NOTICE OF OCCUPANCY RIGHTS UNDER THE	
	VIOLENCE AGAINST WOMEN ACT, FORM HUD-5380	16-29
16-2:	CERTIFICATION OF DOMESTIC VIOLENCE, DATING	
	VIOLENCE, SEXUAL ASSAULT, OR STALKING AND	
	ALTERNATE DOCUMENTATION, FORM HUD-5382	16-35
16-3	EMERGENCY TRANSFER PLAN FOR VICTIMS OF	
	DOMESTIC VIOLENCE, DATING VIOLENCE, SEXUAL ASSAULT,	
	OR STALKING	16-37
16-4	EMERGENCY TRANSFER REQUEST FOR CERTAIN VICTIMS	
	OF DOMESTIC VIOLENCE, DATING VIOLENCE, SEXUAL ASSAULT,	
	OR STALKING, FORM HUD-5383	16-41

Glossary

State Law

Where there is no mandatory federal guidance, PHAs must comply with state law, if it exists. Where state law is more restrictive than federal law, but does not conflict with it, the PHA should follow the state law.

Industry Practice

Where no law or HUD authority exists on a particular subject, industry practice may support PHA policy. An industry practice is a way of doing things that is followed by most housing authorities.

RESOURCES CITED IN THE MODEL ACOP

The model ACOP cites several documents. Where a document or resource is cited frequently, it may be abbreviated. Where it is cited only once or twice, the model ACOP may contain the entire name of the document or resource. Following is a key to abbreviations used for various sources that are frequently cited in the ACOP, and a list of references and document locations that are referenced in the model ACOP or that may be helpful to you.

Abbreviations

Throughout the model ACOP, abbreviations are used to designate certain documents in citations. The following is a table of abbreviations of documents cited by the model ACOP.

Abbreviation	Document
CFR	Code of Federal Regulations
HCV GB	Housing Choice Voucher Program Guidebook (7420.10G), April 2001.
HUD-50058 IB	HUD-50058 Instruction Booklet
PH OCC GB	Public Housing Occupancy Guidebook, June 2003
RHIIP FAQs	Rental Housing Integrity Improvement Program (RHIIP) Frequently Asked Questions
VG	Verification Guidance, March 2004 (attachment to PIH Notice 2004-1)

Resources and Where to Find Them

Following is a list of resources helpful to the PHA or referenced in the model ACOP, and the online location of each.

Document and Location

Code of Federal Regulations

http://www.ecfr.gov

Earned Income Disregard FAQs

http://www.hud.gov/offices/pih/phr/about/ao_faq_eid.cfm

Enterprise Income Verification (EIV) System PHA Security Procedures, Version 1.2, issued January 2005

http://www.hud.gov/offices/pih/programs/ph/rhiip/docs/eivsecguidepha.pdf

Executive Order 11063

http://www.hud.gov/offices/fheo/FHLaws/EXO11063.cfm

Federal Register

http://www.gpo.gov/fdsys/search/getfrtoc.action

General Income and Rent Determination FAQs

http://www.hud.gov/offices/pih/programs/ph/rhiip/faq_gird.cfm

Housing Choice Voucher Program Guidebook (7420.10G), April 2001 http://www.hud.gov/offices/pih/programs/hcv/forms/guidebook.cfm

HUD-50058 Instruction Booklet

https://www.hud.gov/sites/documents/FORM50058INSTRUCTBOOKLET.PDF

Joint Statement of the Department of Housing and Urban Development and the Department of Justice, issued May 17, 2004

http://www.hud.gov/offices/fheo/library/huddojstatement.pdf

Final Guidance to Federal Financial Assistance Recipients Regarding Title VI Prohibition Against National Origin Discrimination Affecting Limited English Proficient Persons, published January 22, 2007

http://www.hud.gov/offices/fheo/promotingfh/FederalRegistepublishedguidance.pdf

Notice PIH 2007-27 (HA), Disallowed Costs and Sanctions Resulting from On-Site Monitoring Reviews

http://www.hud.gov/offices/pih/publications/notices/07/pih2007-27.pdf

Notice PIH 2018-24, Verification of Social Security Numbers (SSNs) Social Security (SS) and Supplemental Security Income (SSI) Benefits; and Effective Use of the Enterprise Income Verification (EIV) System's Identity Verification Report https://www.hud.gov/sites/dfiles/PIH/documents/PIH-2018-

24_EIV_SSN_Notice_FINAL.pdf

Notice PIH 2017-12, Administrative Guidance for Effective and Mandated Use of the Enterprise Income Verification (EIV) System

https://www.hud.gov/sites/documents/PIH2017-12EIVNOTICE.PDF

Notice PIH 2010-26 (HA), Nondiscrimination and Accessibility Notice http://www.hud.gov/offices/pih/publications/notices/10/pih2010-26.pdf

OMB Circular A-133

http://www.whitehouse.gov/omb/circulars/a133_compliance_supplement_2010

Public Housing Occupancy Guidebook, June 2003

http://www.hud.gov/offices/pih/programs/ph/rhiip/phguidebook.cfm

Rental Housing Integrity Improvement Program (RHIIP) Frequently Asked Questions http://www.hud.gov/offices/pih/programs/ph/rhiip/faq.cfm

VAWA Reauthorization Act of 2013

http://www.gpo.gov/fdsys/pkg/FR-2013-08-06/pdf/2013-18920.pdf

Verification FAQs

http://www.hud.gov/offices/pih/programs/ph/rhiip/faq_verif.cfm

Verification Guidance, March 2004 (attachment to Notice PIH 2004-1) http://www.hud.gov/offices/pih/publications/notices/04/verifguidance.pdf

The HUD Web site is https://www.hud.gov/.

Guidebooks, handbooks, and other HUD resources may be found at the HUDClips Web site: https://www.hud.gov/program_offices/administration/hudclips.

PART I: NONDISCRIMINATION

2-I.A. OVERVIEW

Federal laws require PHAs to treat all applicants and tenant families equally, providing the same quality of service, regardless of family characteristics and background. Federal law prohibits discrimination in housing on the basis of race, color, religion, sex, national origin, age, familial status, and disability. In addition, HUD regulations provide for additional protections regarding sexual orientation, gender identity, and marital status. The PHA will comply fully with all federal, state, and local nondiscrimination laws, and with rules and regulations governing fair housing and equal opportunity in housing and employment, including:

- Title VI of the Civil Rights Act of 1964
- Title VIII of the Civil Rights Act of 1968 (as amended by the Community Development Act of 1974 and the Fair Housing Amendments Act of 1988)
- Executive Order 11063
- Section 504 of the Rehabilitation Act of 1973
- The Age Discrimination Act of 1975
- Title II of the Americans with Disabilities Act (to the extent that it applies, otherwise Section 504 and the Fair Housing Amendments govern)
- The Equal Access to Housing in HUD Programs Regardless of Sexual Orientation or Gender Identity Final Rule, published in the *Federal Register* February 3, 2012 and further clarified in Notice PIH 2014-20
- The Violence against Women Act of 2013 (VAWA)
- Any applicable state laws or local ordinances and any legislation protecting individual rights of tenants, applicants, or staff that may subsequently be enacted

When more than one civil rights law applies to a situation, the laws will be read and applied together.

PHA Policy

No state or local nondiscrimination laws or ordinances apply.

2-I.B. NONDISCRIMINATION

Federal regulations prohibit discrimination against certain protected classes and other groups of people. State and local requirements, as well as PHA policies, can prohibit discrimination against additional classes of people.

The PHA shall not discriminate because of race, color, sex, religion, familial status, age, disability or national origin (called "protected classes").

Familial status includes children under the age of 18 living with parents or legal custodians, pregnant women, and people securing custody of children under the age of 18.

The PHA will not discriminate on the basis of marital status, gender identity, or sexual orientation [FR Notice 02/03/12].

PHA Policy

The PHA does not identify any additional protected classes.

The PHA will not use any of these factors to:

- Deny to any family the opportunity to apply for housing, nor deny to any qualified applicant the opportunity to participate in the public housing program
- Provide housing that is different from that provided to others
- Subject anyone to segregation or disparate treatment
- Subject anyone to sexual harassment
- Restrict anyone's access to any benefit enjoyed by others in connection with the housing program
- Treat a person differently in determining eligibility or other requirements for admission
- Steer an applicant or tenant toward or away from a particular area based on any of these factors
- Deny anyone access to the same level of services
- Deny anyone the opportunity to participate in a planning or advisory group that is an integral part of the housing program
- Discriminate in the provision of residential real estate transactions
- Discriminate against someone because they are related to or associated with a member of a protected class
- Publish or cause to be published an advertisement or notice indicating the availability of housing that prefers or excludes persons who are members of a protected class

Providing Information to Families

The PHA must take steps to ensure that families are fully aware of all applicable civil rights laws. As part of the public housing orientation process, the PHA will provide information to public housing applicant families about civil rights requirements.

Discrimination Complaints

If an applicant or tenant family believes that any family member has been discriminated against by the PHA, the family should advise the PHA. The PHA should make every reasonable attempt to determine whether the applicant or tenant family's assertions have merit and take any warranted corrective action.

In all cases, the PHA may advise the family to file a fair housing complaint if the family feels they have been discriminated against under the Fair Housing Act.

Upon receipt of a housing discrimination complaint, the PHA is required to:

- Provide written notice of the complaint to those alleged and inform the complainant that such notice was made
- Investigate the allegations and provide the complainant and those alleged with findings and either a proposed corrective action or an explanation of why corrective action is not warranted
- Keep records of all complaints, investigations, notices, and corrective actions [Notice PIH 2014-20]

PHA Policy

Applicants or tenant families who believe that they have been subject to unlawful discrimination may notify the PHA either orally or in writing.

Within 10 business days of receiving the complaint, the PHA will provide a written notice to those alleged to have violated the rule. The PHA will also send a written notice to the complainant informing them that notice was sent to those alleged to have violated the rule, as well as information on how to complete and submit a housing discrimination complaint form to HUD's Office of Fair Housing and Equal Opportunity (FHEO).

The PHA will attempt to remedy discrimination complaints made against the PHA and will conduct an investigation into all allegations of discrimination.

Within 10 business days following the conclusion of the PHA's investigation, the PHA will provide the complainant and those alleged to have violated the rule with findings and either a proposed corrective action plan or an explanation of why corrective action is not warranted.

The PHA will keep a record of all complaints, investigations, notices, and corrective actions. (See Chapter 16.)

Time Frame for Determination of Citizenship Status [24 CFR 5.508(g)]

For new occupants joining the resident family the PHA must verify status at the first interim or regular reexamination following the person's occupancy, whichever comes first.

If an individual qualifies for a time extension for the submission of required documents, the PHA must grant such an extension for no more than 30 days [24 CFR 5.508(h)].

Each family member is required to submit evidence of eligible status only one time during continuous occupancy.

PHA Policy

The PHA will verify the status of applicants at the time other eligibility factors are determined.

3-II.C. SOCIAL SECURITY NUMBERS [24 CFR 5.216 and 5.218, Notice PIH 2018-24]

The applicant and all members of the applicant's household must disclose the complete and accurate social security number (SSN) assigned to each household member, and the documentation necessary to verify each SSN. If a child under age six has been added to an applicant family within the six months prior to program admission, an otherwise eligible family may be admitted to the program and must disclose and document the child's SSN within 90 days of admission. A detailed discussion of acceptable documentation is provided in Chapter 7.

Note: These requirements do not apply to noncitizens who do not contend eligible immigration status.

In addition, each participant who has not previously disclosed an SSN, has previously disclosed an SSN that HUD or the SSA determined was invalid, or has been issued a new SSN must submit their complete and accurate SSN and the documentation required to verify the SSN at the time of the next interim or annual reexamination or recertification. Participants age 62 or older as of January 31, 2010, whose determination of eligibility was begun before January 31, 2010, are exempt from this requirement and remain exempt even if they move to a new assisted unit.

The PHA must deny assistance to an applicant family if they do not meet the SSN disclosure and documentation requirements contained in 24 CFR 5.216.

3-II.D. FAMILY CONSENT TO RELEASE OF INFORMATION [24 CFR 5.230]

HUD requires each adult family member, and the head of household, spouse, or cohead, regardless of age, to sign form HUD-9886, Authorization for the Release of Information Privacy Act Notice, and other consent forms as needed to collect information relevant to the family's eligibility and level of assistance. Chapter 7 provides detailed information concerning the consent forms and verification requirements.

The PHA must deny admission to the program if any member of the applicant family fails to sign and submit consent forms which allow the PHA to obtain information that the PHA has determined is necessary in administration of the public housing program [24 CFR 960.259(a) and (b)].

PART III: DENIAL OF ADMISSION

3-III.A. OVERVIEW

A family that does not meet the eligibility criteria discussed in Parts I and II must be denied admission.

In addition, HUD requires or permits the PHA to deny admission based on certain types of current or past behaviors of family members as discussed in this part. The PHA's authority in this area is limited by the Violence against Women Act of 2013 (VAWA), which expressly prohibits the denial of admission to an otherwise qualified applicant on the basis or as a direct result of the fact that the applicant is or has been the victim of domestic violence, dating violence, sexual assault, or stalking [24 CFR 5.2005(b)].

This part covers the following topics:

- Required denial of admission
- Other permitted reasons for denial of admission
- Screening
- Criteria for deciding to deny admission
- Prohibition against denial of admission to victims of domestic violence, dating violence, sexual assault, or stalking
- Notice of eligibility or denial

3-III.B. REQUIRED DENIAL OF ADMISSION [24 CFR 960.204]

PHAs are required to establish standards that prohibit admission of an applicant to the public housing program if they have engaged in certain criminal activity or if the PHA has reasonable cause to believe that a household member's current use or pattern of use of illegal drugs, or current abuse or pattern of abuse of alcohol may threaten the health, safety, or right to peaceful enjoyment of the premises by other residents.

Where the statute requires that the PHA prohibit admission for a prescribed period of time after some disqualifying behavior or event, the PHA may choose to continue that prohibition for a longer period of time [24 CFR 960.203(c)(3)(ii)].

HUD requires the PHA to deny assistance in the following cases:

Any member of the household has been evicted from federally-assisted housing in the last
three years for drug-related criminal activity. HUD permits but does not require the PHA to
admit an otherwise-eligible family if the household member has completed a PHA-approved
drug rehabilitation program or the circumstances which led to eviction no longer exist (e.g.
the person involved in the criminal activity no longer lives in the household).

PHA Policy

The PHA will admit an otherwise-eligible family who was evicted from federally-assisted housing within the past three years for drug-related criminal activity, if the PHA is able to verify that the household member who engaged in the criminal activity has completed a supervised drug rehabilitation program approved by the PHA, or the person who committed the crime is no longer living in the household.

• The PHA determines that any household member is currently engaged in the use of illegal drugs. *Drug* means a controlled substance as defined in section 102 of the Controlled Substances Act [21 U.S.C. 802]. *Currently engaged in the illegal use of a drug* means a person has engaged in the behavior recently enough to justify a reasonable belief that there is continuing illegal drug use by a household member [24 CFR 960.205(b)(1)].

PHA Policy

Currently engaged in is defined as any use of illegal drugs during the previous six months.

• The PHA has reasonable cause to believe that any household member's current use or pattern of use of illegal drugs, or current abuse or pattern of abuse of alcohol, may threaten the health, safety, or right to peaceful enjoyment of the premises by other residents.

PHA Policy

In determining reasonable cause, the PHA will consider all credible evidence, including but not limited to, any record of convictions, arrests, or evictions of household members related to the use of illegal drugs or the abuse of alcohol. A conviction will be given more weight than an arrest. A record of arrest(s) will not be used as the basis for the denial or proof that the applicant engaged in disqualifying criminal activity. The PHA will also consider evidence from treatment providers or community-based organizations providing services to household members.

- Any household member has ever been convicted of drug-related criminal activity for the production or manufacture of methamphetamine on the premises of federally assisted housing.
- Any household member is subject to a lifetime registration requirement under a state lifetime sex offender registration program.

PART I: THE APPLICATION PROCESS

4-I.A. OVERVIEW

This part describes the policies that guide the PHA's efforts to distribute and accept applications, and to make preliminary determinations of applicant family eligibility that affect placement of the family on the waiting list. This part also describes the PHA's obligation to ensure the accessibility of the application process.

4-I.B. APPLYING FOR ASSISTANCE

Any family that wishes to reside in public housing must apply for admission to the program [24 CFR 1.4(b)(2)(ii), 24 CFR 960.202(a)(2)(iv), and PH Occ GB, p. 68]. HUD permits the PHA to determine the format and content of its applications, as well how such applications will be made available to interested families and how applications will be accepted by the PHA. However, the PHA must include Form HUD-92006, Supplement to Application for Federally Assisted Housing, as part of the PHA's application [Notice PIH 2009-36].

PHA Policy

Depending upon the length of time between the date of application and the availability of housing, the PHA may use a one- or two-step application process.

A one-step process will be used when it is expected that a family will be selected from the waiting list within 60 days of the date of application. At application, the family must provide all of the information necessary to establish family eligibility and the amount of rent the family will pay.

A two-step process will be used when it is expected that a family will not be selected from the waiting list for at least 60 days from the date of application. Under the two-step application process, the PHA initially will require families to provide only the information needed to make an initial assessment of the family's eligibility, and to determine the family's placement on the waiting list. The family will be required to provide all of the information necessary to establish family eligibility and the amount of rent the family will pay when selected from the waiting list.

Families may obtain application forms from the PHA's office during normal business hours. Families may also request – by telephone or by mail – that an application form be sent to the family via first class mail.

Completed applications must be returned to the PHA by mail, electronically, by fax, or submitted in person during normal business hours. Applications must be filled out completely in order to be accepted by the PHA for processing. If an application is incomplete, the PHA will notify the family of the additional information required.

Page 4-3

4-I.C. ACCESSIBILITY OF THE APPLICATION PROCESS

The PHA must take a variety of steps to ensure that the application process is accessible to those people who might have difficulty complying with the standard PHA application process.

Disabled Populations [24 CFR 8; PH Occ GB, p. 68]

The PHA must provide reasonable accommodation as needed for persons with disabilities to make the application process fully accessible. The facility where applications are accepted and the application process must be fully accessible, or the PHA must provide an alternate approach that provides equal access to the program. Chapter 2 provides a full discussion of the PHA's policies related to providing reasonable accommodations for people with disabilities.

Limited English Proficiency

PHAs are required to take reasonable steps to ensure meaningful access to their programs and activities by persons with limited English proficiency [24 CFR 1]. Chapter 2 provides a full discussion on the PHA's policies related to ensuring access to people with limited English proficiency (LEP).

4-II.E. REPORTING CHANGES IN FAMILY CIRCUMSTANCES

PHA Policy

While the family is on the waiting list, the family must inform the PHA, within 10 business days, of changes in family size or composition, preference status, or contact information, including current residence, mailing address, and phone number. The changes must be submitted in writing.

Changes in an applicant's circumstances while on the waiting list may affect the family's qualification for a particular bedroom size or entitlement to a preference. When an applicant reports a change that affects their placement on the waiting list, the waiting list will be updated accordingly.

4-II.F. UPDATING THE WAITING LIST

HUD requires the PHA to establish policies that describe the circumstances under which applicants will be removed from the waiting list [24 CFR 960.202(a)(2)(iv)].

Purging the Waiting List

The decision to remove an applicant family that includes a person with disabilities from the waiting list is subject to reasonable accommodation. If the applicant did not respond to the PHA's request for information or updates because of the family member's disability, the PHA must, upon the family's request, reinstate the applicant family to their former position on the waiting list as a reasonable accommodation [24 CFR 8.4(a), 24 CFR 100.204(a), and PH Occ GB, p. 39 and 40]. See Chapter 2 for further information regarding reasonable accommodations.

PHA Policy

The waiting list will be updated as needed to ensure that all applicant information is current and timely.

To update the waiting list, the PHA will send an update request via first class mail to each family on the waiting list to determine whether the family continues to be interested in, and to qualify for, the program. This update request will be sent to the last address that the PHA has on record for the family. The update request will provide a deadline by which the family must respond and will state that failure to respond will result in the applicant's name being removed from the waiting list.

The family's response must be in writing and may be delivered in person, by mail, by email, or by fax. Responses should be postmarked or received by the PHA not later than 15 business days from the date of the PHA letter.

If the family fails to respond within 15 business days, the family will be removed from the waiting list without further notice.

If the notice is returned by the post office with no forwarding address, the applicant will be removed from the waiting list without further notice.

If the notice is returned by the post office with a forwarding address, the notice will be resent to the address indicated. The family will have 15 business days to respond from the date the letter was re-sent. If the family fails to respond within this time frame, the family will be removed from the waiting list without further notice.

When a family is removed from the waiting list during the update process for failure to respond, no informal hearing will be offered. Such failures to act on the part of the applicant prevent the PHA from making an eligibility determination; therefore no informal hearing is required.

If a family is removed from the waiting list for failure to respond, the PHA may reinstate the family if the lack of response was due to PHA error, or to circumstances beyond the family's control.

4-III.C. NOTIFICATION OF SELECTION

When the family has been selected from the waiting list, the PHA must notify the family [24 CFR 960.208].

PHA Policy

The PHA will notify the family by first class mail when it is selected from the waiting list.

The notice will inform the family of the following:

Date, time, and location of the scheduled application interview, including any procedures for rescheduling the interview

Who is required to attend the interview

Documents that must be provided at the interview to document the legal identity of household members, including information about what constitutes acceptable documentation

Documents that must be provided at the interview to document eligibility for a preference, if applicable

Other documents and information that should be brought to the interview

If a notification letter is returned to the PHA with no forwarding address, the family will be removed from the waiting list without further notice. Such failure to act on the part of the applicant prevents the PHA from making an eligibility determination; therefore no informal hearing will be offered.

4-III.D. THE APPLICATION INTERVIEW

HUD recommends that the PHA obtain the information and documentation needed to make an eligibility determination through a private interview. Being invited to attend an interview does not constitute admission to the program.

Assistance cannot be provided to the family until all SSN documentation requirements are met. However, if the PHA determines that an applicant family is otherwise eligible to participate in the program, the family may retain its place on the waiting list for a period of time determined by the PHA [Notice PIH 2018-24].

Reasonable accommodation must be made for persons with disabilities who are unable to attend an interview due to their disability [24 CFR 8.4(a) and 24 CFR 100.204(a)].

PHA Policy

Families selected from the waiting list are required to participate in an eligibility interview.

The head of household and the spouse/cohead will be strongly encouraged to attend the interview together. However, either the head of household or the spouse/cohead may attend the interview on behalf of the family. Verification of information pertaining to adult members of the household not present at the interview will not begin until signed release forms are returned to the PHA.

The interview will be conducted only if the head of household or spouse/cohead provides appropriate documentation of legal identity (Chapter 7 provides a discussion of proper documentation of legal identity). If the family representative does not provide the required documentation, the appointment may be rescheduled when the proper documents have been obtained.

Pending disclosure and documentation of social security numbers, the PHA will allow the family to retain its place on the waiting list for *[insert amount of time reasonable for PHA]*. If not all household members have disclosed their SSNs at the next time a unit becomes available, the PHA will offer a unit to the next eligible applicant family on the waiting list.

If the family is claiming a waiting list preference, the family must provide documentation to verify their eligibility for a preference (see Chapter 7). If the family is verified as eligible for the preference, the PHA will proceed with the interview. If the PHA determines the family is not eligible for the preference, the interview will not proceed and the family will be placed back on the waiting list according to the date and time of their application.

The family must provide the information necessary to establish the family's eligibility, including suitability, and to determine the appropriate amount of rent the family will pay. The family must also complete required forms, provide required signatures, and submit required documentation. If any materials are missing, the PHA will provide the family with a written list of items that must be submitted.

6-I.H. PERIODIC PAYMENTS

Periodic payments are forms of income received on a regular basis. HUD regulations specify periodic payments that are and are not included in annual income.

Periodic Payments **Included** in Annual Income

- Periodic payments from sources such as <u>social security</u>, <u>unemployment and welfare</u> <u>assistance</u>, <u>annuities</u>, <u>insurance policies</u>, <u>retirement funds</u>, <u>and pensions</u>. However, periodic payments from retirement accounts, annuities, and similar forms of investments are counted only after they exceed the amount contributed by the family [24 CFR 5.609(b)(4) and (b)(3)].
- <u>Disability or death benefits and lottery receipts</u> paid periodically, rather than in a single lump sum [24 CFR 5.609(b)(4) and HCV, p. 5-14]

Lump-Sum Payments for the Delayed Start of a Periodic Payment

Most lump sums received as a result of delays in processing periodic payments, such as unemployment or welfare assistance, are counted as income. However, lump-sum receipts for the delayed start of periodic social security or supplemental security income (SSI) payments are not counted as income. Additionally, any deferred disability benefits that are received in a lump sum or in prospective monthly amounts from the Department of Veterans Affairs are to be excluded from annual income [24 CFR 5.609(c)(14)].

PHA Policy

When a delayed-start payment is received and reported during the period in which the PHA is processing an annual reexamination, the PHA will adjust the tenant rent retroactively for the period the payment was intended to cover. The family may pay in full any amount due or request to enter into a repayment agreement with the PHA.

See the chapter on reexaminations for information about a family's obligation to report lump-sum receipts between annual reexaminations.

Treatment of Overpayment Deductions from Social Security Benefits

The PHA must make a special calculation of annual income when the Social Security Administration (SSA) overpays an individual, resulting in a withholding or deduction from his or her benefit amount until the overpayment is paid in full. The amount and duration of the withholding will vary depending on the amount of the overpayment and the percent of the benefit rate withheld. Regardless of the amount withheld or the length of the withholding period, the PHA must use the reduced benefit amount after deducting only the amount of the overpayment withholding from the gross benefit amount [Notice PIH 2018-24].

Periodic Payments **Excluded** from Annual Income

• Payments received for the <u>care of foster children or foster adults</u> (usually persons with disabilities, unrelated to the tenant family, who are unable to live alone) [24 CFR 5.609(c)(2)]. Kinship care payments are considered equivalent to foster care payments and are also excluded from annual income [Notice PIH 2012-1].

PHA Policy

The PHA will exclude payments for the care of foster children and foster adults only if the care is provided through an official arrangement with a local welfare agency [HCV GB, p. 5-18].

- Amounts paid by a state agency to a family with a <u>member who has a developmental</u> <u>disability and is living at home</u> to offset the cost of services and equipment needed to keep the developmentally disabled family member at home [24 CFR 5.609(c)(16)]
- Amounts received under the <u>Low-Income Home Energy Assistance Program</u> (42 U.S.C. 1626(c)) [24 CFR 5.609(c)(17)]
- Amounts received under the Child Care and Development Block Grant Act of 1990 (42 U.S.C. 9858q) [24 CFR 5.609(c)(17)]
- Earned Income Tax Credit (EITC) refund payments (26 U.S.C. 32(j)) [24 CFR 5.609(c)(17)]. *Note:* EITC may be paid periodically if the family elects to receive the amount due as part of payroll payments from an employer.
- Lump sums received as a result of <u>delays in processing Social Security and SSI payments</u> (see section 6-I.H.) [24 CFR 5.609(c)(14)].
- Lump-sums or prospective monthly amounts received as deferred <u>disability benefits from the Department of Veterans Affairs (VA)</u> [24 CFR 5.609(c)(14)].

- (g) Payments received under programs funded in whole or in part under the Workforce Investment Act of 1998 (29 U.S.C. 2931)
- (h) Deferred disability benefits from the Department of Veterans Affairs, whether received as a lump sum or in monthly prospective amounts
- (i) Income derived from the disposition of funds to the Grand River Band of Ottawa Indians (Pub. L. 94-540, 90 Stat. 2503-04)
- (j) Payments, funds, or distributions authorized, established, or directed by the Seneca Nation Settlement Act of 1990 (25 U.S.C. 1774f(b))
- (k) A lump sum or periodic payment received by an individual Indian pursuant to the Class Action Settlement Agreement in the United States District Court case entitled *Elouise Cobell et al.* v. *Ken Salazar et al.*, for a period of one year from the time of receipt of that payment as provided in the Claims Resolution Act of 2010
- (1) The first \$2,000 of per capita shares received from judgment funds awarded by the Indian Claims Commission or the U. S. Claims Court, the interests of individual Indians in trust or restricted lands, including the first \$2,000 per year of income received by individual Indians from funds derived from interests held in such trust or restricted lands (25 U.S.C. 1407-1408)
- (m) Benefits under the Indian Veterans Housing Opportunity Act of 2010 (only applies to Native American housing programs)
- (n) Payments received from programs funded under Title V of the Older Americans Act of 1985 (42 U.S.C. 3056(f))
- (o) Payments received on or after January 1, 1989, from the Agent Orange Settlement Fund or any other fund established pursuant to the settlement in *In Re Agent Orange* product liability litigation, M.D.L. No. 381 (E.D.N.Y.)
- (p) Payments received under 38 U.S.C. 1833(c) to children of Vietnam veterans born with spinal bifida, children of women Vietnam veterans born with certain birth defects, and children of certain Korean service veterans born with spinal bifida
- (q) Payments received under the Maine Indian Claims Settlement Act of 1980 (25 U.S.C. 1721)
- (r) The value of any child care provided or arranged (or any amount received as payment for such care or reimbursement for costs incurred for such care) under the Child Care and Development Block Grant Act of 1990 (42 U.S.C. 9858q)
- (s) Earned income tax credit (EITC) refund payments received on or after January 1, 1991 (26 U.S.C. 32(j))
- (t) Payments by the Indian Claims Commission to the Confederated Tribes and Bands of Yakima Indian Nation or the Apache Tribe of Mescalero Reservation (Pub. L. 95-433)

- (u) Amounts of scholarships funded under Title IV of the Higher Education Act of 1965, including awards under federal work-study programs or under the Bureau of Indian Affairs student assistance programs (20 U.S.C. 1087uu). For Section 8 programs, the exception found in § 237 of Public Law 109–249 applies and requires that the amount of financial assistance in excess of tuition shall be considered income in accordance with the provisions codified at 24 CFR 5.609(b)(9), except for those persons with disabilities as defined by 42 U.S.C. 1437a(b)(3)(E) (Pub. L. 109–249)
- (v) Allowances, earnings and payments to AmeriCorps participants under the National and Community Service Act of 1990 (42 U.S.C. 12637(d))
- (w) Any amount of crime victim compensation (under the Victims of Crime Act) received through crime victim assistance (or payment or reimbursement of the cost of such assistance) as determined under the Victims of Crime Act because of the commission of a crime against the applicant under the Victims of Crime Act (42 U.S.C. 10602)
- (x) Any amounts in an "individual development account" as provided by the Assets for Independence Act, as amended in 2002
- (y) Payments made from the proceeds of Indian tribal trust cases as described in Notice PIH 2013–30, "Exclusion from Income of Payments under Recent Tribal Trust Settlements" (25 U.S.C. 117b(a))
- (z) Major disaster and emergency assistance received under the Robert T. Stafford Disaster Relief and Emergency Assistance Act and comparable disaster assistance provided by states, local governments, and disaster assistance organizations
- (aa) Distributions from an ABLE account, and actual or imputed interest on the ABLE account balance

Chapter 7

VERIFICATION

[24 CFR 960.259, 24 CFR 5.230, Notice PIH 2018-18]

INTRODUCTION

The PHA must verify all information that is used to establish the family's eligibility and level of assistance and is required to obtain written authorization from the family in order to collect the information. Applicants and program participants must cooperate with the verification process as a condition of receiving assistance. The PHA must not pass on the cost of verification to the family.

The PHA will follow the verification guidance provided by HUD in Notice PIH 2018-18 and any subsequent guidance issued by HUD. This chapter summarizes those requirements and provides supplementary PHA policies.

Part I describes the general verification process. Part II provides more detailed requirements related to family information. Part III provides information on income and assets, and Part IV covers mandatory deductions.

Verification policies, rules and procedures will be modified as needed to accommodate persons with disabilities. All information obtained through the verification process will be handled in accordance with the records management policies established by the PHA.

PART I: GENERAL VERIFICATION REQUIREMENTS

7-I.A. FAMILY CONSENT TO RELEASE OF INFORMATION [24 CFR 960.259, 24 CFR 5.230]

The family must supply any information that the PHA or HUD determines is necessary to the administration of the program and must consent to PHA verification of that information [24 CFR 960.259(a)(1)].

Consent Forms

It is required that all adult applicants and tenants sign form HUD-9886, Authorization for Release of Information. The purpose of form HUD-9886 is to facilitate automated data collection and computer matching from specific sources and provides the family's consent only for the specific purposes listed on the form. HUD and the PHA may collect information from State Wage Information Collection Agencies (SWICAs) and current and former employers of adult family members. Only HUD is authorized to collect information directly from the Internal Revenue Service (IRS) and the Social Security Administration (SSA). Adult family members must sign other consent forms as needed to collect information relevant to the family's eligibility and level of assistance.

Penalties for Failing to Consent [24 CFR 5.232]

If any family member who is required to sign a consent form fails to do so, the PHA will deny admission to applicants and terminate the lease of tenants. The family may request a hearing in accordance with the PHA's grievance procedures.

7-I.B. OVERVIEW OF VERIFICATION REQUIREMENTS

HUD's Verification Hierarchy [Notice PIH 2018-18]

HUD mandates the use of the EIV system and offers administrative guidance on the use of other methods to verify family information and specifies the circumstances in which each method will be used. In general, HUD requires the PHA to use the most reliable form of verification that is available and to document the reasons when the PHA uses a lesser form of verification.

In order of priority, the forms of verification that the PHA will use are:

- Up-front Income Verification (UIV) using HUD's Enterprise Income Verification (EIV) system
- Up-front Income Verification (UIV) using a non-HUD system
- Written Third Party Verification (may be provided by applicant or resident)
- Written Third-party Verification Form
- Oral Third-party Verification
- Self-Certification

Each of the verification methods is discussed in subsequent sections below.

Requirements for Acceptable Documents

PHA Policy

Any documents used for verification must be the original (not photocopies) and generally must be dated within 60 days of the PHA request. The documents must not be damaged, altered or in any way illegible.

Print-outs from web pages are considered original documents.

The PHA staff member who views the original document must make a photocopy, annotate the copy with the name of the person who provided the document and the date the original was viewed, and sign the copy.

Any family self-certifications must be made in a format acceptable to the PHA and must be signed in the presence of a PHA representative or PHA notary public.

File Documentation

The PHA must document in the file how the figures used in income and rent calculations were determined. All verification attempts, information obtained, and decisions reached during the verification process will be recorded in the family's file in sufficient detail to demonstrate that the PHA has followed all of the verification policies set forth in this ACOP. The record should be sufficient to enable a staff member or HUD reviewer to understand the process followed and conclusions reached.

PHA Policy

The PHA will document, in the family file, the following:

Reported family annual income

Value of assets

Expenses related to deductions from annual income

Other factors influencing the adjusted income or income-based rent determination

When the PHA is unable to obtain third-party verification, the PHA will document in the family file the reason that third-party verification was not available [24 CFR 960.259(c)(1); Notice PIH 2018-18].

7-I.C. UP-FRONT INCOME VERIFICATION (UIV)

Up-front income verification (UIV) refers to the PHA's use of the verification tools available from independent sources that maintain computerized information about earnings and benefits. UIV will be used to the extent that these systems are available to the PHA.

There may be legitimate differences between the information provided by the family and UIV-generated information. If the family disputes the accuracy of UIV data, no adverse action can be taken until the PHA has independently verified the UIV information and the family has been granted the opportunity to contest any adverse findings through the PHA's informal review/hearing processes. (For more on UIV and income projection, see section 6-I.C.)

Upfront Income Verification Using HUD's Enterprise Income Verification (EIV) System (Mandatory)

PHAs must use HUD's EIV system in its entirety as a third-party source to verify tenant employment and income information during mandatory reexaminations or recertifications of family composition and income in accordance with 24 CFR 5.236 and administrative guidance issued by HUD. HUD's EIV system contains data showing earned income, unemployment benefits, social security benefits, and SSI benefits for participant families. The following policies apply to the use of HUD's EIV system.

EIV Income and IVT Reports

The data shown on income and income validation tool (IVT) reports is updated quarterly. Data may be between three and six months old at the time reports are generated.

PHA Policy

The PHA will obtain income and IVT reports for annual reexaminations on a monthly basis. Reports will be generated as part of the regular reexamination process.

Income and IVT reports will be compared to family-provided information as part of the annual reexamination process. Income reports may be used in the calculation of annual income, as described in Chapter 6.I.C. Income reports may also be used to meet the regulatory requirement for third party verification, as described above. Policies for resolving discrepancies between income and IVT reports and family-provided information will be resolved as described in Chapter 6.I.C. and in this chapter.

Income and IVT reports will be used in interim reexaminations to identify any discrepancies between reported income and income shown in the EIV system, and as necessary to verify earned income, and to verify and calculate unemployment benefits, Social Security and/or SSI benefits. EIV will also be used to verify that families claiming zero income are not receiving income from any of these sources.

Income and IVT reports will be retained in resident files with the applicable annual or interim reexamination documents.

When the PHA determines through EIV reports and third-party verification that a family has concealed or under-reported income, corrective action will be taken pursuant to the policies in Chapter 15, Program Integrity.

EIV Identity Verification

The EIV system verifies resident identities against Social Security Administration (SSA) records. These records are compared to Public and Indian Housing Information Center (PIC) data for a match on social security number, name, and date of birth.

PHAs are required to use EIV's *Identity Verification Report* on a monthly basis to improve the availability of income information in EIV [Notice PIH 2018-18].

When identity verification for a resident fails, a message will be displayed within the EIV system and no income information will be displayed.

PHA Policy

The PHA will identify residents whose identity verification has failed by reviewing EIV's *Identity Verification Report* on a monthly basis. The PHA will attempt to resolve PIC/SSA discrepancies by obtaining appropriate documentation from the tenant. When the PHA determines that discrepancies exist as a result of PHA errors, such as spelling errors or incorrect birth dates, it will correct the errors promptly.

Upfront Income Verification Using Non-HUD Systems (Optional)

In addition to mandatory use of the EIV system, HUD encourages PHAs to utilize other upfront verification sources.

PHA Policy

The PHA will inform all applicants and residents of its use of the following UIV resources during the admission and reexamination process:

HUD's EIV system

[Insert any additional UIV sources used by the PHA]

7-I.D. THIRD-PARTY WRITTEN AND ORAL VERIFICATION

HUD's current verification hierarchy defines two types of written third-party verification. The more preferable form, "written third-party verification," consists of an original document generated by a third-party source, which may be received directly from a third-party source or provided to the PHA by the family. If written third-party verification is not available, the PHA must attempt to obtain a "written third-party verification form." This is a standardized form used to collect information from a third party.

Written Third-Party Verification [Notice PIH 2018-18]

Written third-party verification documents must be original and authentic and may be supplied by the family or received from a third-party source.

Examples of acceptable tenant-provided documents include, but are not limited to: pay stubs, payroll summary reports, employer notice or letters of hire and termination, SSA benefit verification letters, bank statements, child support payment stubs, welfare benefit letters and/or printouts, and unemployment monetary benefit notices.

The PHA is required to obtain, at minimum, two current and consecutive pay stubs for determining annual income from wages.

The PHA may reject documentation provided by the family if the document is not an original, if the document appears to be forged, or if the document is altered, mutilated, or illegible.

PHA Policy

Third-party documents provided by the family must be dated within 60 days of the PHA request date.

If the PHA determines that third-party documents provided by the family are not acceptable, the PHA will explain the reason to the family and request additional documentation.

As verification of earned income, the PHA will require the family to provide the two most current, consecutive pay stubs.

Written Third-Party Verification Form

When upfront verification is not available and the family is unable to provide written third-party documents, the PHA must request a written third-party verification form. HUD's position is that this traditional third-party verification method presents administrative burdens and risks which may be reduced through the use of family-provided third-party documents.

PHAs may mail, fax, or email third-party written verification form requests to third-party sources.

PHA Policy

The PHA will send third-party verification forms directly to the third party.

Third-party verification forms will be sent when third-party verification documents are unavailable or are rejected by the PHA.

Oral Third-Party Verification [Notice PIH 2018-18]

For third-party oral verification, PHAs contact sources, identified by UIV techniques or by the family, by telephone or in person.

Oral third-party verification is mandatory if neither form of written third-party verification is available.

Third-party oral verification may be used when requests for written third-party verification forms have not been returned within a reasonable time—e.g., 10 business days.

PHAs should document in the file the date and time of the telephone call or visit, the name of the person contacted, the telephone number, as well as the information confirmed.

PHA Policy

In collecting third-party oral verification, PHA staff will record in the family's file the name and title of the person contacted, the date and time of the conversation (or attempt), the telephone number used, and the facts provided.

When any source responds verbally to the initial written request for verification the PHA will accept the verbal response as oral verification but will also request that the source complete and return any verification forms that were provided.

When Third-Party Verification is Not Required [Notice PIH 2018-18]

Third-party verification may not be available in all situations. HUD has acknowledged that it may not be cost-effective or reasonable to obtain third-party verification of income, assets, or expenses when these items would have a minimal impact on the family's total tenant payment.

PHA Policy

If the family cannot provide original documents, the PHA will pay the service charge required to obtain third-party verification, unless it is not cost effective in which case a self-certification will be acceptable as the only means of verification. The cost of verification will not be passed on to the family.

The cost of postage and envelopes to obtain third-party verification of income, assets, and expenses is not an unreasonable cost [VG, p. 18].

Primary Documents

Third-party verification is not required when legal documents are the primary source, such as a birth certificate or other legal documentation of birth.

Imputed Assets

The PHA may accept a self-certification from the family as verification of assets disposed of for less than fair market value [HCV GB, p. 5-28].

PHA Policy

The PHA will accept a self-certification from a family as verification of assets disposed of for less than fair market value [HCV GB, p. 5-28].

Value of Assets and Asset Income [24 CFR 960.259]

For families with net assets totaling \$5,000 or less, the PHA may accept the family's declaration of asset value and anticipated asset income. However, the PHA is required to obtain third-party verification of all assets regardless of the amount during the intake process, whenever a family member is added, and at least every three years thereafter.

PHA Policy

For families with net assets totaling \$5,000 or less, the PHA will accept the family's self-certification of the value of family assets and anticipated asset income when applicable. The family's declaration must show each asset and the amount of income expected from that asset. All family members 18 years of age and older must sign the family's declaration.

The PHA will use third-party documentation for assets as part of the intake process, whenever a family member is added to verify the individual's assets, and every three years thereafter.

PART II: VERIFYING FAMILY INFORMATION

7-II.A. VERIFICATION OF LEGAL IDENTITY

PHA Policy

The PHA will require families to furnish verification of legal identity for each household member.

Verification of Legal Identity for Adults	Verification of Legal Identity for Children
Certificate of birth, naturalization papers	Certificate of birth
Church issued baptismal certificate	Adoption papers
Current, valid driver's license or Department of Motor Vehicle identification card	Custody agreement Health and Human Services ID
U.S. military discharge (DD 214)	Certified school records
Current U.S. passport	
Current employer identification card	

If a document submitted by a family is illegible for any reason or otherwise questionable, more than one of these documents may be required.

If none of these documents can be provided and at the PHA's discretion, a third party who knows the person may attest to the person's identity. The certification must be provided in a format acceptable to the PHA and be signed in the presence of a PHA representative or PHA notary public.

Legal identity will be verified for all applicants at the time of eligibility determination and in cases where the PHA has reason to doubt the identity of a person representing him or herself to be a tenant or a member of a tenant family.

7-II.B. SOCIAL SECURITY NUMBERS [24 CFR 5.216 and Notice PIH 2018-24]

The family must provide documentation of a valid social security number (SSN) for each member of the household, with the exception of individuals who do not contend eligible immigration status. Exemptions also include, existing residents who were at least 62 years of age as of January 31, 2010, and had not previously disclosed an SSN.

The PHA must accept the following documentation as acceptable evidence of the social security number:

- An original SSN card issued by the Social Security Administration (SSA)
- An original SSA-issued document, which contains the name and SSN of the individual
- An original document issued by a federal, state, or local government agency, which contains the name and SSN of the individual

The PHA may only reject documentation of an SSN provided by an applicant or resident if the document is not an original document, if the original document has been altered, mutilated, is illegible, or if the document appears to be forged.

PHA Policy

The PHA will explain to the applicant or resident the reasons the document is not acceptable and request that the individual obtain and submit acceptable documentation of the SSN to the PHA within 90 days.

If an applicant family includes a child under 6 years of age who joined the household within the 6 months prior to the date of program admission, an otherwise eligible family may be admitted and must provide documentation of the child's SSN within 90 days. A 90-day extension will be granted if the PHA determines that the resident's failure to comply was due to unforeseen circumstances and was outside of the resident's control.

PHA Policy

The PHA will grant one additional 90-day extension if needed for reasons beyond the applicant's control, such as delayed processing of the SSN application by the SSA, natural disaster, fire, death in the family, or other emergency.

When a resident requests to add a new household member who is at least 6 years of age, or who is under the age of 6 and has an SSN, the resident must provide the complete and accurate SSN assigned to each new member at the time of reexamination or recertification, in addition to the documentation required to verify it. The PHA may not add the new household member until such documentation is provided.

Chapter 8

LEASING AND INSPECTIONS

[24 CFR 5, Subpart G; 24 CFR 966, Subpart A]

INTRODUCTION

Public housing leases are the contractual basis of the legal relationship between the PHA and the tenant. All units must be occupied pursuant to a dwelling lease agreement that complies with HUD regulations.

HUD regulations require the PHA to inspect each dwelling unit prior to move-in, at move-out, and annually during the period of occupancy. In addition, the PHA may conduct additional inspections in accordance with PHA policy.

This chapter is divided into two parts as follows:

<u>Part I: Leasing</u>. This part describes pre-leasing activities and the PHA's policies pertaining to lease execution, lease modification, and payments under the lease.

Part II: Inspections. This part describes the PHA's policies for inspecting dwelling units.

PART I: LEASING

8-I.A. OVERVIEW

An eligible family may occupy a public housing dwelling unit under the terms of a lease. The lease must meet all regulatory requirements, and must also comply with applicable state and local laws and codes.

The term of the lease must be for a period of 12 months. The lease must be renewed automatically for another 12-month term, except that the PHA may not renew the lease if the family has violated the community service requirement [24 CFR 966.4(a)(2)].

PHAs must adopt smoke-free policies, which must be implemented no later than July 30, 2018. The policy is attached as Exhibit 8-1.

PHAs must include in the ACOP residential minimum heating standards policies [Notice PIH 2018-19]. The policy is included in Part I of this chapter.

Part I of this chapter contains regulatory information on leasing, where applicable, as well as the PHA's leasing policies.

8-I.B. LEASE ORIENTATION

PHA Policy

After unit acceptance but prior to occupancy, a PHA representative will conduct a lease orientation with the family. The head of household or spouse is required to attend.

Orientation Agenda

PHA Policy

When families attend the lease orientation, they will be provided with:

A copy of the lease

A copy of the PHA's grievance procedure

A copy of the house rules

A copy of the PHA's schedule of maintenance charges

A copy of "Is Fraud Worth It?" (form HUD-1141-OIG), which explains the types of actions a family must avoid and the penalties for program abuse

A copy of "What You Should Know about EIV," a guide to the Enterprise Income Verification (EIV) system published by HUD as an attachment to Notice PIH 2017-12

A copy of the form HUD-5380, VAWA Notice of Occupancy Rights

A copy of form HUD-5382, Certification of Domestic Violence, Dating Violence, Sexual Assault, or Stalking

A copy of the PHA's smoke free policy

A notice that includes the procedures for requesting relief and the PHA's criteria for granting requests for relief for excess utility surcharges

Topics to be discussed and explained to all families include:

Applicable deposits and all other charges

Review and explanation of lease provisions

Unit maintenance requests and work orders

The PHA's interim reporting requirements

Review and explanation of occupancy forms

Community service requirements

Family choice of rent

VAWA protections

Smoke-free policies

8-I.G. MINIMUM HEATING STANDARDS [Notice PIH 2018-19]

PHAs in states, territories, or localities with existing minimum heating standards must use their respective local standards for public housing dwelling units. For PHAs where state or local minimum heating standards do not exist, PHAs must use the HUD-prescribed heating standards specified in Notice PIH 2018-19.

PHA Policy

The PHA is located in an area where state or local residential heating standards exist and will utilize those standards for public housing units. Therefore, the PHA's minimum heating standards are as follows:

Minimum temperature:

If the PHA controls the temperature, the minimum temperature in each unit must be at least [insert degrees Fahrenheit]. If the resident controls the temperature, the heating equipment must have the capability of heating to at least [insert degrees Fahrenheit].

Minimum temperature capability:

[If PHAs are allowed flexibility maintaining the indoor temperature when the outdoor temperature approaches the design day temperature, insert criteria for when flexibility applies and the minimum temperature threshold that the indoor temperature should never fall below. Note, the design day temperature refers to the lowest expected outdoor temperature a heating system was designed to accommodate.]

Measurement:

Temperature measurements must be taken according to the following methodology: [insert methodology for taking temperature measurements].

PART II: INSPECTIONS

8-II.A. OVERVIEW

HUD regulations require the PHA to inspect each dwelling unit prior to move-in, at move-out, and annually during occupancy. In addition, the PHA may require additional inspections, in accordance with PHA Policy. This part contains the PHA's policies governing inspections, notification of unit entry, and inspection results.

8-II.B. TYPES OF INSPECTIONS

Move-In Inspections [24 CFR 966.4(i)]

The lease must require the PHA and the family to inspect the dwelling unit prior to occupancy in order to determine the condition of the unit and equipment in the unit. A copy of the initial inspection, signed by the PHA and the tenant, must be provided to the tenant and retained in the resident file.

PHA Policy

Any adult family member may attend the initial inspection and sign the inspection form for the head of household.

Move-Out Inspections [24 CFR 966.4(i)]

The PHA must inspect the unit at the time the resident vacates the unit and must allow the resident to participate in the inspection if he or she wishes, unless the tenant vacates without notice to the PHA. The PHA must provide to the tenant a statement of any charges to be made for maintenance and damage beyond normal wear and tear.

The difference between the condition of the unit at move-in and move-out establishes the basis for any charges against the security deposit so long as the work needed exceeds that for normal wear and tear.

PHA Policy

When applicable, the PHA will provide the tenant with a statement of charges to be made for maintenance and damage beyond normal wear and tear, within 10 business days of conducting the move-out inspection.

Annual Inspections [24 CFR 5.705]

Section 6(f)(3) of the United States Housing Act of 1937 requires that PHAs inspect each public housing project annually to ensure that the project's units are maintained in decent, safe, and sanitary condition. The PHA shall continue using the Uniform Physical Condition Standards (UPCS) in 24 CFR 5, Subpart G, Physical Condition Standards and Inspection Requirements, to conduct annual project inspections. These standards address the inspection of the site area, building systems and components, and dwelling units.

PHA Policy

The PHA will inspect all occupied units annually using HUD's Uniform Physical Condition Standards (UPCS).

Quality Control Inspections

The purpose of quality control inspections is to assure that all defects were identified in the original inspection, and that repairs were completed at an acceptable level of craftsmanship and within an acceptable time frame

PHA Policy

Supervisory quality control inspections will be conducted in accordance with the PHA's maintenance plan.

Special Inspections

PHA Policy

PHA staff may conduct a special inspection for any of the following reasons:

Housekeeping

Unit condition

Suspected lease violation

Preventive maintenance

Routine maintenance

There is reasonable cause to believe an emergency exists

Other Inspections

PHA Policy

Building exteriors, grounds, common areas and systems will be inspected according to the PHA's maintenance plan.

8-II.C. NOTICE AND SCHEDULING OF INSPECTIONS

Notice of Entry

Non-emergency Entries [24 CFR 966.4(j)(1)]

The PHA may enter the unit, with reasonable advance notification to perform routine inspections and maintenance, make improvements and repairs, or to show the unit for re-leasing. A written statement specifying the purpose of the PHA entry delivered to the dwelling unit at least two days before such entry is considered reasonable advance notification.

PHA Policy

The PHA will notify the resident in writing at least 48 hours prior to any non-emergency inspection.

For regular annual inspections, the family will receive at least 2 weeks written notice of the inspection to allow the family to prepare the unit for the inspection.

Entry for repairs requested by the family will not require prior notice. Resident-requested repairs presume permission for the PHA to enter the unit.

Emergency Entries [24 CFR 966.4(j)(2)]

The PHA may enter the dwelling unit at any time without advance notice when there is reasonable cause to believe that an emergency exists. If no adult household member is present at the time of an emergency entry, the PHA must leave a written statement showing the date, time and purpose of the entry prior to leaving the dwelling unit.

Scheduling of Inspections

PHA Policy

Inspections will be conducted during business hours. If a family needs to reschedule an inspection, they must notify the PHA at least 24 hours prior to the scheduled inspection. The PHA will reschedule the inspection no more than once unless the resident has a verifiable good cause to delay the inspection. The PHA may request verification of such cause.

Attendance at Inspections

Residents are required to be present for move-in inspections [24 CFR 966.4(i)]. There is no such requirement for other types of inspections.

PHA Policy

Except at move-in inspections, the resident is not required to be present for the inspection. The resident may attend the inspection if he or she wishes.

If no one is at home, the inspector will enter the unit, conduct the inspection and leave a copy of the inspection report in the unit.

8-II.D. INSPECTION RESULTS

The PHA is obligated to maintain dwelling units and the project in decent, safe and sanitary condition and to make necessary repairs to dwelling units [24 CFR 966.4(e)].

Emergency Repairs [24 CFR 966.4(h)]

If the unit is damaged to the extent that conditions are created which are hazardous to the life, health, or safety of the occupants, the tenant must immediately notify the PHA of the damage, and the PHA must make repairs within a reasonable time frame.

If the damage was caused by a household member or guest, the PHA must charge the family for the reasonable cost of repairs. The PHA may also take lease enforcement action against the family.

If the PHA cannot make repairs quickly, the PHA must offer the family standard alternative accommodations. If the PHA can neither repair the defect within a reasonable time frame nor offer alternative housing, rent shall be abated in proportion to the seriousness of the damage and loss in value as a dwelling. Rent shall not be abated if the damage was caused by a household member or guest, or if the resident rejects the alternative accommodations.

PHA Policy

When conditions in the unit are hazardous to life, health, or safety, the PHA will make repairs or otherwise abate the situation within 24 hours.

Defects hazardous to life, health or safety include, but are not limited to, the following:

Any condition that jeopardizes the security of the unit

Major plumbing leaks or flooding, waterlogged ceiling or floor in imminent danger of falling

Natural or LP gas or fuel oil leaks

Any electrical problem or condition that could result in shock or fire

Absence of a working heating system when outside temperature is below 60 degrees Fahrenheit

Utilities not in service, including no running hot water

Conditions that present the imminent possibility of injury

Obstacles that prevent safe entrance or exit from the unit

Absence of a functioning toilet in the unit

Inoperable smoke detectors

Non-emergency Repairs

PHA Policy

The PHA will correct non-life threatening health and safety defects within 15 business days of the inspection date. If the PHA is unable to make repairs within that period due to circumstances beyond the PHA's control (e.g. required parts or services are not available, weather conditions, etc.) the PHA will notify the family of an estimated date of completion.

The family must allow the PHA access to the unit to make repairs.

Resident-Caused Damages

PHA Policy

Damages to the unit beyond wear and tear will be billed to the tenant in accordance with the policies in 8-I.F., Maintenance and Damage Charges.

Repeated or excessive damages to the unit beyond normal wear and tear will be considered a serious or repeated violation of the lease.

Housekeeping

PHA Policy

Residents whose housekeeping habits pose a non-emergency health or safety risk, encourage insect or rodent infestation, or cause damage to the unit are in violation of the lease. In these instances, the PHA will provide proper notice of a lease violation.

A reinspection will be conducted within 30 days to confirm that the resident has complied with the requirement to abate the problem. Failure to abate the problem or allow for a reinspection is considered a violation of the lease and may result in termination of tenancy in accordance with Chapter 13.

Notices of lease violation will also be issued to residents who purposely disengage the unit's smoke detector. Only one warning will be given. A second incidence will result in lease termination.

EXHIBIT 8-1: SMOKE-FREE POLICY

In accordance with HUD regulations, the Housing Authority has adopted these smoke-free policies. The policies are effective as of Board approval date.

Due to the increased risk of fire, increased maintenance costs, and the known health effects of secondhand smoke, smoking is prohibited in all living units and interior areas, including but not limited to hallways, rental and administrative offices, community centers, day care centers, laundry centers, and similar structures. Smoking is also prohibited in outdoor areas within 25 feet from public housing and administrative office buildings.

This policy applies to all employees, residents, household members, guests, and service persons. Residents are responsible for ensuring that household members and guests comply with this rule.

The term "smoking" means any inhaling, exhaling, burning, or carrying any lighted cigar, cigarette, pipe, or other prohibited tobacco product in any manner or any form. Prohibited tobacco products include water pipes or hookahs.

Violation of the smoke-free policy constitutes a violation of the terms of the public housing lease. Consequences of lease violations include termination of tenancy.

PHA POLICIES

Designated Smoking Areas (DSA)

Example policy 1: The PHA has established designated smoking areas at (INSERT LOCATION(S)). Residents using the designated smoking areas must extinguish all smoking materials and dispose of them safely in receptacles provided for that purpose.

Example policy 2: The PHA has not designated any smoking areas on the PHA's property. Residents may not discard smoking products on the property.

Electronic Nicotine Delivery Systems (ENDS)

Electronic nicotine delivery systems (ENDS) include e-cigarettes, nicotine inhalers, and vaping devices.

Example policy 1: Use of ENDS is permitted in public housing units but is prohibited in common areas and in outdoor areas within 25 feet from housing and administrative buildings. That is, use of ENDS is prohibited in all common and outdoor areas in which smoking is prohibited.

Example policy 2: Use of ENDS is permitted in public housing units only as a reasonable accommodation approved by the PHA that necessary for a person with disabilities. Use of ENDS is prohibited in common areas and in outdoor areas within 25 feet from housing and administrative buildings. That is, use of ENDS is prohibited in all areas in which smoking is prohibited.

Example policy 3: Use of ENDS is not permitted in public housing units, common areas, or in outdoor areas within 25 feet from housing and administrative buildings.

Effective Date

The PHA's effective date(s) of this smoke-free policy is/are as follows:

Example 1: The smoke-free policy will be effective for all residents, household members, employees, guests, and service persons on (INSERT DATE ON OR BEFORE JULY 30, 2018).

Example 2: The smoke-free policy will be effective for all employees and service persons on (INSERT DATE ON OR BEFORE JULY 30, 2018).

The smoke-free policy will take effect at the next annual lease renewal for each resident household. Residents must execute a smoke-free lease addendum as part of the annual lease renewal process. Regardless of the lease renewal date, all residents must be in compliance with the smoke-free policy no later than July 30, 2018.

Enforcement

The PHA must enforce smoke-free policies when a resident violates this policy. When enforcing the lease, the PHA will provide due process and allow residents to exercise their right to an informal settlement and formal hearing. The PHA will not evict a resident for a single incident of smoking in violation of this policy. As such, the PHA will implement a graduated enforcement framework that includes escalating warnings. Prior to pursuing eviction for violation of smoke-free policies, the PHA will take specific, progressive monitoring and enforcement actions, while at the same time educating tenants and providing smoking cessation information. The lease will identify the actions that constitute a policy violation, quantify the number of documented, verified violations that warrant enforcement action, state any disciplinary actions that will be taken for persistent non-responsiveness or repeated noncompliance, and state how many instances of noncompliance will constitute a violation. Tenancy termination and eviction will be pursued only as a last resort. The PHA may terminate tenancy at any time for violations of the lease and failure to otherwise fulfill household obligations if resident behavior disturbs other residents' peaceful enjoyment and is not conducive to maintaining the property in a decent, safe, and sanitary condition.

Example 1: Upon issuance of a written warning from the property manager and/or a documented complaint, the PHA will increase the frequency of unit inspections for a suspected policy violator. (INSERT PHA POLICY ON MORE FREQUENT INSPECTION HERE)

Example 2: The PHA will provide information and resources on smoking cessations, including: (INSERT A DESCRIPTION OF ANY INFORMATION THE PHA WILL PROVIDE)

Example 3: If the resident does not have any new violations for (INSERT PERIOD OF TIME), the resident will be considered to have a clear record, and no further enforcement action will be taken.

Example 4: Repeated violation of the smoke-free policy may rise to the level of other good cause for termination of tenancy. (INSERT PHA POLICY ON THE NUMBER OF DOCUMENTED VIOLATIONS THAT CONSTITUTE TERMINATION)

Reasonable Accommodation

While addiction to nicotine or smoking is not a disability, the PHA will provide reasonable accommodation to persons with disabilities who smoke that are in compliance with the requirements of this smoke-free policy.

PART II: REEXAMINATIONS FOR FAMILIES PAYING FLAT RENTS [24 CFR 960.257(2)]

9-II.A. OVERVIEW

HUD requires that the PHA offer all families the choice of paying income-based rent or flat rent at least annually. The PHA's policies for offering families a choice of rents are located in Chapter 6.

For families who choose flat rents, the PHA must conduct a reexamination of family composition at least annually, and must conduct a reexamination of family income at least once every 3 years [24 CFR 960.257(a)(2)]. The PHA is only required to provide the amount of income-based rent the family might pay in those years that the PHA conducts a full reexamination of income and family composition, or upon request of the family after the family submits updated income information [24 CFR 960.253(e)(2)].

As it does for families that pay income-based rent, the PHA must also review compliance with the community service requirement for families with nonexempt individuals.

This part contains the PHA's policies for conducting reexaminations of families who choose to pay flat rents.

9-II.B. FULL REEXAMINATION OF FAMILY INCOME AND COMPOSITION

Frequency of Reexamination

PHA Policy

For families paying flat rents, the PHA will conduct a full reexamination of family income and composition once every 3 years.

Reexamination Policies

PHA Policy

In conducting full reexaminations for families paying flat rents, the PHA will follow the policies used for the annual reexamination of families paying income-based rent as set forth in Sections 9-I.B through 9-I.D above.

9-II.C. REEXAMINATION OF FAMILY COMPOSITION ("ANNUAL UPDATE")

As noted above, full reexaminations are conducted every 3 years for families paying flat rents. In the years between full reexaminations, regulations require the PHA to conduct a reexamination of family composition ("annual update") [24 CFR 960.257(a)(2)].

The annual update process is similar to the annual reexamination process, except that the PHA does not collect information about the family's income and expenses, and the family's rent is not recalculated following an annual update.

Scheduling

The PHA must establish a policy to ensure that the reexamination of family composition for families choosing to pay the flat rent is completed at least annually [24 CFR 960.257(a)(2)].

PHA Policy

For families paying flat rents, annual updates will be conducted in each of the 2 years following the full reexamination.

In scheduling the annual update, the PHA will follow the policy used for scheduling the annual reexamination of families paying income-based rent as set forth in Section 9-I.B. above.

Conducting Annual Updates

The terms of the public housing lease require the family to furnish information necessary for the redetermination of rent and family composition [24 CFR 966.4(c)(2)].

PHA Policy

Generally, the family will not be required to attend an interview for an annual update. However, if the PHA determines that an interview is warranted, the family may be required to attend.

Notification of the annual update will be sent by first-class mail and will inform the family of the information and documentation that must be provided to the PHA. The family will have 10 business days to submit the required information to the PHA. If the family is unable to obtain the information or documents within the required time frame, the family may request an extension. The PHA will accept required documentation by mail, by email, by fax, or in person.

If the family's submission is incomplete, or the family does not submit the information in the required time frame, the PHA will send a second written notice to the family. The family will have 10 business days from the date of the second notice to provide the missing information or documentation to the PHA.

If the family does not provide the required documents or information within the required time frame (plus any extensions), the family will be in violation of their lease and may be terminated in accordance with the policies in Chapter 13.

9-III.D. PROCESSING THE INTERIM REEXAMINATION

Method of Reporting

PHA Policy

The family may notify the PHA of changes either orally or in writing. If the family provides oral notice, the PHA may also require the family to submit the changes in writing.

Generally, the family will not be required to attend an interview for an interim reexamination. However, if the PHA determines that an interview is warranted, the family may be required to attend.

Based on the type of change reported, the PHA will determine the documentation the family will be required to submit. The family must submit any required information or documents within 10 business days of receiving a request from the PHA. This time frame may be extended for good cause with PHA approval. The PHA will accept required documentation by mail, by email, by fax, or in person.

Effective Dates

The PHA must make the interim reexamination within a reasonable time after the family request [24 CFR 960.257(b)].

PHA Policy

If the tenant rent is to *increase*:

The increase generally will be effective on the first of the month following 30 days' notice to the family.

If a family fails to report a change within the required time frames, or fails to provide all required information within the required time frames, the increase will be applied retroactively, to the date it would have been effective had the information been provided on a timely basis. The family will be responsible for any underpaid rent and may be offered a repayment agreement in accordance with the policies in Chapter 16.

If the tenants rent is to *decrease*:

The decrease will be effective on the first day of the month following the month in which the change was reported. In cases where the change cannot be verified until after the date the change would have become effective, the change will be made retroactively.

PART II: TERMINATION BY PHA – MANDATORY

13-II.A. OVERVIEW

HUD requires mandatory termination of the lease for certain actions or inactions of the family. There are other actions or inactions of the family that constitute *grounds* for lease termination, but the lease termination is not mandatory. The PHA must establish policies for termination of the lease in these cases where termination is optional for the PHA.

For those tenant actions or failures to act where HUD requires termination, the PHA has no such option. In those cases, the family's lease must be terminated. This part describes situations in which HUD requires the PHA to terminate the lease.

13-II.B. FAILURE TO PROVIDE CONSENT [24 CFR 960.259(a) and (b)]

The PHA must terminate the lease if any family member fails to sign and submit any consent form s/he is required to sign for any reexamination. See Chapter 7 for a complete discussion of consent requirements.

13-II.C. FAILURE TO DOCUMENT CITIZENSHIP [24 CFR 5.514(c) and (d) and 24 CFR 960.259(a)]

The PHA must terminate the lease if (1) a family fails to submit required documentation within the required timeframe concerning any family member's citizenship or immigration status; (2) a family submits evidence of citizenship and eligible immigration status in a timely manner, but United States Citizenship and Immigration Services (USCIS) primary and secondary verification does not verify eligible immigration status of the family, resulting in no eligible family members; or (3) a family member, as determined by the PHA, has knowingly permitted another individual who is not eligible for assistance to reside (on a permanent basis) in the unit. For (3), such termination must be for a period of at least 24 months. This does not apply to ineligible noncitizens already in the household where the family's assistance has been prorated.

See Chapter 7 for a complete discussion of documentation requirements.

13-II.D. FAILURE TO DISCLOSE AND DOCUMENT SOCIAL SECURITY NUMBERS [24 CFR 5.218(c), 24 CFR 960.259(a)(3), Notice PIH 2018-24]

The PHA must terminate assistance if a participant family fails to disclose the complete and accurate social security numbers of each household member and the documentation necessary to verify each social security number.

However, if the family is otherwise eligible for continued program assistance, and the PHA determines that the family's failure to meet the SSN disclosure and documentation requirements was due to circumstances that could not have been foreseen and were outside of the family's control, the PHA may defer the family's termination and provide the opportunity to comply with the requirement within a period not to exceed 90 calendar days from the date the PHA determined the family to be noncompliant.

PHA Policy

The PHA will defer the family's termination and provide the family with the opportunity to comply with the requirement for a period of 90 calendar days for circumstances beyond the participant's control such as delayed processing of the SSN application by the SSA, natural disaster, fire, death in the family, or other emergency, if there is a reasonable likelihood that the participant will be able to disclose an SSN by the deadline.

See Chapter 7 for a complete discussion of documentation and certification requirements.

13-II.E. FAILURE TO ACCEPT THE PHA'S OFFER OF A LEASE REVISION [24 CFR 966.4(1)(2)(ii)(E)]

The PHA must terminate the lease if the family fails to accept the PHA's offer of a lease revision to an existing lease, provided the PHA has done the following:

- The revision is on a form adopted by the PHA in accordance with 24 CFR 966.3 pertaining to requirements for notice to tenants and resident organizations and their opportunity to present comments.
- The PHA has made written notice of the offer of the revision at least 60 calendar days before the lease revision is scheduled to take effect.
- The PHA has specified in the offer a reasonable time limit within that period for acceptance by the family.

See Chapter 8 for information pertaining to PHA policies for offering lease revisions.

If a family member has violated federal, state, or local law that imposes obligations in connection with the occupancy or use of the premises.

If a household member has engaged in or threatened violent or abusive behavior toward PHA personnel.

Abusive or violent behavior towards PHA personnel includes verbal as well as physical abuse or violence. Use of racial epithets, or other language, written or oral, that is customarily used to intimidate may be considered abusive or violent behavior.

Threatening refers to oral or written threats or physical gestures that communicate intent to abuse or commit violence.

In making its decision to terminate the lease, the PHA will consider alternatives as described in Section 13-III.D and other factors described in Sections 13-III.E and 13-III.F. Upon consideration of such alternatives and factors, the PHA may, on a case-by-case basis, choose not to terminate the lease.

Family Absence from Unit [24 CFR 982.551(i)]

It is reasonable that the family may be absent from the public housing unit for brief periods. However, the PHA needs a policy on how long the family may be absent from the unit. Absence in this context means that no member of the family is residing in the unit.

PHA Policy

The family must supply any information or certification requested by the PHA to verify that the family is living in the unit, or relating to family absence from the unit, including any PHA-requested information or certification on the purposes of family absences. The family must cooperate with the PHA for this purpose.

The family must promptly notify the PHA when all family members will be absent from the unit for an extended period. An extended period is defined as any period greater than 30 calendar days. In such a case promptly means within 10 business days of the start of the extended absence.

If a family is absent from the public housing unit for more than 180 consecutive days, and the family does not adequately verify that they are living in the unit, the PHA will terminate the lease for other good cause.

Abandonment of the unit. If the family appears to have vacated the unit without giving proper notice, the PHA will follow state and local landlord-tenant law pertaining to abandonment before taking possession of the unit. If necessary, the PHA will secure the unit immediately to prevent vandalism and other criminal activity.

Over-Income Families [24 CFR 960.261; FR Notice 7/26/18; Notice PIH 2019-11]

The Housing Opportunity Through Modernization Act (HOTMA) of 2016 placed an income limitation on public housing tenancies. The over-income requirement states that after a family's adjusted income has exceeded 120 percent of area median income (AMI) (or a different limitation established by the secretary) for two consecutive years, the PHA must either terminate the family's tenancy within six months of the determination, or charge the family a monthly rent that is the higher of the applicable fair market rent (FMR) or the amount of monthly subsidy for the unit, including amounts from the operating and capital funds, as determined by regulations.

Notice PIH 2019-11 also requires that PHAs publish over-income limits in their ACOP and update them no later than 60 days after HUD publishes new income limits each year. The over-income limit is calculated by multiplying the very low-income limit (VLI) by 2.4, as adjusted for family size.

PHAs also have discretion, under 24 CFR 960.261, to adopt policies allowing termination of tenancy for families whose income exceeds the limit for program eligibility. Such policies would exempt families participating in the Family Self-Sufficiency (FSS) program or currently receiving the earned income disallowance.

PHA Policy

At annual or interim reexamination, if a family's adjusted income exceeds the applicable over-income limit, the PHA will document the family file and begin tracking the family's over-income status.

If one year after the applicable annual or interim reexamination the family's income continues to exceed the applicable over-income limit, the PHA will notify the family in writing that their income has exceeded the over-income limit for one year, and that if the family continues to be over-income for 12 consecutive months, the family will be subject to the PHA's over-income policies.

If two years after the applicable annual or interim reexamination the family's income continues to exceed the applicable over-income limit, the PHA will charge the family a rent that is the higher of the applicable fair market rent (FMR) or the amount of monthly subsidy for the unit. The PHA will notify the family in writing of their new rent amount. The new rent amount will be effective 30 days after the PHA's written notice to the family.

If, at any time, an over-income family experiences a decrease in income, the family may request an interim redetermination of rent in accordance with PHA policy. If, as a result, the previously over-income family is now below the over-income limit, the family is no longer subject to over-income provisions as of the effective date of the recertification. The PHA will notify the family in writing that over-income policies no longer apply to them. If the family's income later exceeds the over-income limit again, the family is entitled to a new two-year grace period.

The PHA will begin tracking over-income families once these policies have been adopted, but no later than March 24, 2019.

The PHA will not evict or terminate the tenancies of families whose income exceeds the income limit for program eligibility as described at 24 CFR 960.261.

The PHA will rely on the following over-income limits. These numbers will be updated within 60 days of HUD publishing new income limits each year and will be effective for all annual and interim reexaminations once these policies have been adopted.

Family Size	1	2	3	4	5	6	7	8
Over- Income Limit								

For families larger than 8 persons, the over-income limit will be calculated by multiplying the applicable very-low income limit by 2.4.

13-III.D. ALTERNATIVES TO TERMINATION OF TENANCY

Exclusion of Culpable Household Member [24 CFR 966.4(l)(5)(vii)(C)]

As an alternative to termination of the lease for criminal activity or alcohol abuse HUD provides that the PHA may consider exclusion of the culpable household member. Such an alternative can be used for any other reason where such a solution appears viable in accordance with PHA policy.

Additionally, under the Violence against Women Reauthorization Act of 2013, the PHA may bifurcate a lease in order to terminate the tenancy of an individual who is a tenant or lawful occupant of a unit and engages in criminal activity directly related to domestic violence, dating violence, sexual assault, or stalking.

PHA Policy

The PHA will consider requiring the tenant to exclude a household member in order to continue to reside in the assisted unit, where that household member has participated in or been culpable for action or failure to act that warrants termination.

As a condition of the family's continued occupancy, the head of household must certify that the culpable household member has vacated the unit and will not be permitted to visit or to stay as a guest in the assisted unit. The family must present evidence of the former household member's current address upon PHA request.

Repayment of Family Debts

PHA Policy

If a family owes amounts to the PHA, as a condition of continued occupancy, the PHA will require the family to repay the full amount or to enter into a repayment agreement, within 30 days of receiving notice from the PHA of the amount owed. See Chapter 16 for policies on repayment agreements.

13-III.E. CRITERIA FOR DECIDING TO TERMINATE TENANCY

A PHA that has grounds to terminate a tenancy is not required to do so, except as explained in Part II of this chapter, and may consider all of the circumstances relevant to a particular case before making a decision.

Evidence [24 CFR 982.553(c)]

For criminal activity, HUD permits the PHA to terminate the lease if a *preponderance of the evidence* indicates that a household member has engaged in the activity, regardless of whether the household member has been arrested or convicted, and without satisfying the standard of proof used for a criminal conviction.

PHA Policy

The PHA will use the preponderance of the evidence as the standard for making all termination decisions.

Preponderance of the evidence is defined as evidence which is of greater weight or more convincing than the evidence which is offered in opposition to it; that is, evidence which as a whole shows that the fact sought to be proved is more probable than not.

Preponderance of the evidence may not be determined by the number of witnesses, but by the greater weight of all evidence.

Consideration of Circumstances [24 CFR 966.4(l)(5)(vii)(B)]

Although it is required that certain lease provisions exist for criminal activity and alcohol abuse, HUD provides that the PHA may consider all circumstances relevant to a particular case in order to determine whether or not to terminate the lease.

Such relevant circumstances can also be considered when terminating the lease for any other reason.

PHA Policy

The PHA will consider the following facts and circumstances before deciding whether to terminate the lease for any of the HUD required lease provisions or for any other reasons:

The seriousness of the offending action, especially with respect to how it would affect other residents' safety or property

The extent of participation or culpability of the leaseholder, or other household members, in the offending action, including whether the culpable member is a minor, a person with disabilities, or (as discussed further in section 13-III.F) a victim of domestic violence, dating violence, sexual assault, or stalking

The effects that the eviction will have on other family members who were not involved in the action or failure to act

The effect on the community of the termination, or of the PHA's failure to terminate the tenancy

The effect of the PHA's decision on the integrity of the public housing program

The demand for housing by eligible families who will adhere to lease responsibilities

The extent to which the leaseholder has shown personal responsibility and whether they have taken all reasonable steps to prevent or mitigate the offending action

The length of time since the violation occurred, including the age of the individual at the time of the conduct, as well as the family's recent history, and the likelihood of favorable conduct in the future

While a record of arrest(s) will not be used as the basis for termination, an arrest may, however, trigger an investigation to determine whether the participant actually engaged in disqualifying criminal activity. As part of its investigation, the PHA may obtain the police report associated with the arrest and consider the reported circumstances of the arrest. The PHA may also consider:

Any statements made by witnesses or the participant not included in the police report

Whether criminal charges were filed

Whether, if filed, criminal charges were abandoned, dismissed, not prosecuted, or ultimately resulted in an acquittal

Any other evidence relevant to determining whether or not the participant engaged in disqualifying activity

Evidence of criminal conduct will be considered if it indicates a demonstrable risk to safety and/or property.

In the case of program abuse, the dollar amount of the underpaid rent and whether or not a false certification was signed by the family

Consideration of Rehabilitation [24 CFR 966.4(l)(5)(vii)(D)]

HUD authorizes PHAs to take into consideration whether a household member who had used illegal drugs or abused alcohol and is no longer engaging in such use or abuse is participating in or has successfully completed a supervised drug or alcohol rehabilitation program.

PHA Policy

In determining whether to terminate the lease for illegal drug use or a pattern of illegal drug use, or for abuse or a pattern of abuse of alcohol, by a household member who is no longer engaging in such use or abuse, the PHA will consider whether such household member has successfully completed a supervised drug or alcohol rehabilitation program.

For this purpose the PHA will require the tenant to submit evidence of the household member's successful completion of a supervised drug or alcohol rehabilitation program.

Reasonable Accommodation [24 CFR 966.7]

If the family includes a person with disabilities, the PHA's decision to terminate the family's lease is subject to consideration of reasonable accommodation in accordance with 24 CFR Part 8.

PHA Policy

If a family indicates that the behavior of a family member with a disability is the reason for a proposed termination of lease, the PHA will determine whether the behavior is related to the disability. If so, upon the family's request, the PHA will determine whether alternative measures are appropriate as a reasonable accommodation. The PHA will only consider accommodations that can reasonably be expected to address the behavior that is the basis of the proposed lease termination. See Chapter 2 for a discussion of reasonable accommodation.

Nondiscrimination Limitation [24 CFR 966.4(1)(5)(vii)(F)]

The PHA's eviction actions must be consistent with fair housing and equal opportunity provisions of 24 CFR 5.105.

13-III.F. TERMINATIONS RELATED TO DOMESTIC VIOLENCE, DATING VIOLENCE, SEXUAL ASSAULT, OR STALKING

This section addresses the protections against termination of tenancy that the Violence against Women Act of 2013 (VAWA) provides for public housing residents who are victims of domestic violence, dating violence, sexual assault, or stalking. For general VAWA requirements and PHA policies pertaining to notification, documentation, and confidentiality, see section 16-VII of this ACOP, where definitions of key VAWA terms are also located.

VAWA Protections against Termination [24 CFR 5.2005(c)]

VAWA provides that no person may deny assistance, tenancy, or occupancy rights to public housing to a tenant on the basis or as a direct result of criminal activity directly relating to domestic violence, dating violence, sexual assault, or stalking that is engaged in by a member of the household of the tenant or any guest or other person under the control of the tenant, if the tenant or affiliated individual is the victim or threatened victim of such domestic violence, dating violence, sexual assault, or stalking [FR Notice 8/6/13].

VAWA further provides that incidents of actual or threatened domestic violence, dating violence, sexual assault, or stalking may not be construed either as serious or repeated violations of the lease by the victim or threatened victim of such violence or as good cause for terminating the tenancy or occupancy rights of the victim of such violence [24 CFR 5.2005(c)(1), FR Notice 8/6/13].

Limits on VAWA Protections [24 CFR 5.2005(d) and (e), FR Notice 8/6/13]

While VAWA prohibits a PHA from using domestic violence, dating violence, sexual assault, or stalking as the cause for a termination or eviction action against a public housing tenant who is the victim of the abuse, the protections it provides are not absolute. Specifically:

- VAWA does not limit a PHA's otherwise available authority to terminate assistance to or
 evict a victim for lease violations not premised on an act of domestic violence, dating
 violence, sexual assault, or stalking providing that the PHA does not subject the victim to a
 more demanding standard than the standard to which it holds other tenants.
- VAWA does not limit a PHA's authority to terminate the tenancy of any public housing tenant if the PHA can demonstrate an actual and imminent threat to other tenants or those employed at or providing service to the property if that tenant's tenancy is not terminated.

HUD regulations define *actual and imminent threat* to mean words, gestures, actions, or other indicators of a physical threat that (a) is real, (b) would occur within an immediate time frame, and (c) could result in death or serious bodily harm [24 CFR 5.2005(d)(2) and (e)]. In determining whether an individual would pose an actual and imminent threat, the factors to be considered include:

- The duration of the risk
- The nature and severity of the potential harm
- The likelihood that the potential harm will occur
- The length of time before the potential harm would occur [24 CFR 5.2005(e)]

In order to demonstrate an actual and imminent threat, the PHA must have objective evidence of words, gestures, actions, or other indicators. Even when a victim poses an actual and imminent threat, however, HUD regulations authorize a PHA to terminate the victim's assistance "only when there are no other actions that could be taken to reduce or eliminate the threat, including but not limited to transferring the victim to a different unit, barring the perpetrator from the property, contacting law enforcement to increase police presence or develop other plans to keep the property safe, or seeking other legal remedies to prevent the perpetrator from acting on a threat" [24 CFR 5.2005(d)(3)]. Additionally, HUD regulations state that restrictions "predicated on public safety cannot be based on stereotypes, but must be tailored to particularized concerns about individual residents" [24 CFR 5.2005(d)(3)].

PHA Policy

In determining whether a public housing tenant who is a victim of domestic violence, dating violence, sexual assault, or stalking is an actual and imminent threat to other tenants or those employed at or providing service to a property, the PHA will consider the following, and any other relevant, factors:

Whether the threat is toward an employee or tenant other than the victim of domestic violence, dating violence, sexual assault, or stalking

Whether the threat is a physical danger beyond a speculative threat

Whether the threat is likely to happen within an immediate time frame

Whether the threat to other tenants or employees can be eliminated in some other way, such as by helping the victim relocate to a confidential location, transferring the victim to another unit, or seeking a legal remedy to prevent the perpetrator from acting on the threat

If the tenant wishes to contest the PHA's determination that he or she is an actual and imminent threat to other tenants or employees, the tenant may do so as part of the grievance hearing or in a court proceeding.

Documentation of Abuse [24 CFR 5.2007]

PHA Policy

When an individual facing termination of tenancy for reasons related to domestic violence, dating violence, sexual assault, or stalking claims protection under VAWA, the PHA will request in writing that the individual provide documentation supporting the claim in accordance with the policies in section 16-VII.D of this ACOP.

The PHA reserves the right to waive the documentation requirement if it determines that a statement or other corroborating evidence from the individual will suffice. In such cases the PHA will document the waiver in the individual's file.

Terminating or Evicting a Perpetrator of Domestic Violence

Although VAWA provides protection from termination for victims of domestic violence, it does not provide such protection for perpetrators. In fact, VAWA gives the PHA the explicit authority to bifurcate a lease, or remove a household member from a lease, "in order to evict, remove, or terminate assistance to any individual who is a tenant or lawful occupant of the housing and who engages in criminal activity directly relating to domestic violence, dating violence, sexual assault, or stalking against an affiliated individual or other individual, without evicting, removing, terminating assistance to, or otherwise penalizing a victim of such criminal activity who is also a tenant or lawful occupant of the housing" [FR Notice 8/6/13]. Moreover, HUD regulations impose on the PHA the obligation to consider lease bifurcation in any circumstances involving domestic violence, dating violence, or stalking [24 CFR 966.4(e)(9)].

Specific lease language affirming the PHA's authority to bifurcate a lease is not necessary, and the authority supersedes any local, state, or federal law to the contrary. However, if the PHA chooses to exercise its authority to bifurcate a lease, it must follow any procedures prescribed by HUD or by applicable local, state, or federal law for eviction, lease termination, or termination of assistance. This means that the PHA must follow the same rules when terminating or evicting an individual as it would when terminating or evicting an entire family [FR Notice 3/16/07]. However, perpetrators should be given no more than 30 days' notice of termination in most cases [Notice PIH 2017-08].

PHA Policy

The PHA will bifurcate a family's lease and terminate the tenancy of a family member if the PHA determines that the family member has committed criminal acts of physical violence against other family members or others. This action will not affect the tenancy or program assistance of the remaining, nonculpable family members.

In making its decision, the PHA will consider all credible evidence, including, but not limited to, a signed certification (form HUD-5382) or other documentation of abuse submitted to the PHA by the victim in accordance with this section and section 16-VII.D. The PHA will also consider the factors in section 13.III.E. Upon such consideration, the PHA may, on a case-by-case basis, choose not to bifurcate the lease and terminate the tenancy of the culpable family member.

If the PHA does bifurcate the lease and terminate the tenancy of the culpable family member, it will do so in accordance with the lease, applicable law, and the policies in this ACOP. If the person removed from the lease was the only tenant eligible to receive assistance, the PHA must provide any remaining tenant a chance to establish eligibility for the unit. If the remaining tenant cannot do so, the PHA must provide the tenant reasonable time to find new housing or to establish eligibility for another housing program covered by VAWA 2013.

PART IV: NOTIFICATION REQUIREMENTS, EVICTION PROCEDURES AND RECORD KEEPING

13-IV.A. OVERVIEW

HUD regulations specify the requirements for the notice that must be provided prior to lease termination. This part discusses those requirements and the specific requirements that precede and follow termination for certain criminal activities which are addressed in the regulations. This part also discusses specific requirements pertaining to the actual eviction of families and record keeping.

13-IV.B. CONDUCTING CRIMINAL RECORDS CHECKS [24 CFR 5.903(e)(ii) and 24 CFR 960.259]

HUD authorizes PHAs to conduct criminal records checks on public housing residents for lease enforcement and eviction. PHA policy determines when the PHA will conduct such checks.

PHA Policy

The PHA will conduct criminal records checks when it has come to the attention of the PHA, either from local law enforcement or by other means, that an individual has engaged in the destruction of property, engaged in violent activity against another person, or has interfered with the right to peaceful enjoyment of the premises of other residents. Such checks will also include sex offender registration information. In order to obtain such information, all adult household members must sign consent forms for release of criminal conviction and sex offender registration records on an annual basis.

The PHA may not pass along to the tenant the costs of a criminal records check.

13-IV.C. DISCLOSURE OF CRIMINAL RECORDS TO FAMILY [24 CFR 5.903(f), 24 CFR 5.905(d) and 24 CFR 966.4(l)(5)(iv)]

In conducting criminal records checks, if the PHA uses the authority of 24 CFR 5.903 and 5.905 to obtain such information, certain protections must be afforded the tenant before any adverse action is taken. In such cases if the PHA obtains criminal records information from a state or local agency showing that a household member has been convicted of a crime, or is subject to a sex offender registration requirement, relevant to lease enforcement or eviction, the PHA must notify the household of the proposed action and must provide the subject of the record and the tenant a copy of such information, and an opportunity to dispute the accuracy and relevance of the information before an eviction or lease enforcement action is taken.

PHA Policy

In all cases where criminal record or sex offender registration information would result in lease enforcement or eviction, the PHA will notify the household in writing of the proposed adverse action and will provide the subject of the record and the tenant a copy of such information, and an opportunity to dispute the accuracy and relevance of the information before an eviction or lease enforcement action is taken.

The family will be given 10 business days from the date of the PHA notice, to dispute the accuracy and relevance of the information. If the family does not contact the PHA to dispute the information within that 10 business day period, the PHA will proceed with the termination action.

Should the tenant not exercise their right to dispute prior to any adverse action, the tenant still has the right to dispute in the grievance hearing or court trial.

13-IV.D. LEASE TERMINATION NOTICE [24 CFR 966.4(1)(3)]

Form, Delivery, and Content of the Notice

Notices of lease termination must be in writing. The notice must state the specific grounds for termination, the date the termination will take place, the resident's right to reply to the termination notice, and their right to examine PHA documents directly relevant to the termination or eviction. If the PHA does not make the documents available for examination upon request by the tenant, the PHA may not proceed with the eviction [24 CFR 996.4(m)].

When the PHA is required to offer the resident an opportunity for a grievance hearing, the notice must also inform the resident of their right to request a hearing in accordance with the PHA's grievance procedure. In these cases, the tenancy shall not terminate until the time for the tenant to request a grievance hearing has expired and the grievance procedure has been completed.

When the PHA is not required to offer the resident an opportunity for a grievance hearing because HUD has made a due process determination and the lease termination is for criminal activity that threatens health, safety or right to peaceful enjoyment or for drug-related criminal activity, the notice of lease termination must state that the tenant is not entitled to a grievance hearing on the termination. It must specify the judicial eviction procedure to be used by the PHA for eviction of the tenant, and state that HUD has determined that the eviction procedure provides the opportunity for a hearing in court that contains the basic elements of due process as defined in HUD regulations. The notice must also state whether the eviction is for a criminal activity that threatens the health, safety, or right to peaceful enjoyment of the premises of other residents or employees of the PHA, or for a drug-related criminal activity on or off the premises.

PHA Policy

The PHA will attempt to deliver notices of lease termination directly to the tenant or an adult member of the household. If such attempt fails, the notice will be sent by first-class mail the same day.

All notices of lease termination will include a copy of the forms HUD-5382 and HUD-5380 to accompany the termination notice. Any tenant who claims that the cause for termination involves domestic violence, dating violence, sexual assault, or stalking of which the tenant or affiliated individual of the tenant is the victim will be given the opportunity to provide documentation in accordance with the policies in sections 13-III.F and 16-VII.D.

Timing of the Notice [24 CFR 966.4(l)(3)(i)]

The PHA must give written notice of lease termination of:

- 14 calendar days in the case of failure to pay rent
- A reasonable period of time considering the seriousness of the situation (but not to exceed 30 calendar days)

If the health or safety of other residents, PHA employees, or persons residing in the immediate vicinity of the premises is threatened

If any member of the household has engaged in any drug-related criminal activity or violent criminal activity

If any member of the household has been convicted of a felony

• 30 calendar days in any other case, except that if a state or local law allows a shorter notice period, such shorter period shall apply

PHA Policy

The PHA will give written notice of 14 calendar days for nonpayment of rent. For all other lease terminations the PHA will give 30 days written notice or, if state or local law allows less than 30 days, such shorter notice will be given.

The Notice to Vacate that may be required under state or local law may be combined with or run concurrently with the notice of lease termination.

Notice of Nonrenewal Due to Community Service Noncompliance [24 CFR 966.4(I)(2)(ii)(D), 24 CFR 960.603(b) and 24 CFR 960.607(b)]

When the PHA finds that a family is in noncompliance with the community service requirement, the tenant and any other noncompliant resident must be notified in writing of this determination. Notices of noncompliance will be issued in accordance with the requirements and policies in Section 11-I.E.

PHA Policy

If after receiving a notice of initial noncompliance the family does not request a grievance hearing, or does not take either corrective action required by the notice within the required timeframe, a termination notice will be issued in accordance with the policies above.

If a family agreed to cure initial noncompliance by signing an agreement, and is still in noncompliance after being provided the 12-month opportunity to cure, the family will be issued a notice of continued noncompliance. The notice of continued noncompliance will be sent in accordance with the policies in Section 11-I.E. and will also serve as the notice of termination of tenancy.

Notice of Termination Based on Citizenship Status [24 CFR 5.514 (c) and (d)]

In cases where termination of tenancy is based on citizenship status, HUD requires the notice of termination to contain additional information. In addition to advising the family of the reasons their assistance is being terminated, the notice must also advise the family of any of the following that apply: the family's eligibility for proration of assistance, the criteria and procedures for obtaining relief under the provisions for preservation of families, the family's right to request an appeal to the USCIS of the results of secondary verification of immigration status and to submit additional documentation or a written explanation in support of the appeal, and the family's right to request an informal hearing with the PHA either upon completion of the USCIS appeal or in lieu of the USCIS appeal. Please see Chapter 14 for the PHA's informal hearing procedures.

13-IV.E. EVICTION [24 CFR 966.4(1)(4) and 966.4(m)]

Eviction notice means a notice to vacate, or a complaint or other initial pleading used under state or local law to commence an eviction action. The PHA may only evict the tenant from the unit by instituting a court action, unless the law of the jurisdiction permits eviction by administrative action, after a due process administrative hearing, and without a court determination of the rights and liabilities of the parties.

PHA Policy

When a family does not vacate the unit after receipt of a termination notice, by the deadline given in the notice, the PHA will follow state and local landlord-tenant law in filing an eviction action with the local court that has jurisdiction in such cases.

If the eviction action is finalized in court and the family remains in occupancy beyond the deadline to vacate given by the court, the PHA will seek the assistance of the court to remove the family from the premises as per state and local law.

The PHA may not proceed with an eviction action if the PHA has not made available the documents to be used in the case against the family, and has not afforded the family the opportunity to examine and copy such documents in accordance with the provisions of 24 CFR 966.4(1)(3) and (m).

13-IV.F. NOTIFICATION TO POST OFFICE [24CFR 966.4(1)(5)(iii)(B)]

When the PHA evicts an individual or family for criminal activity, including drug-related criminal activity, the PHA must notify the local post office serving the dwelling unit that the individual or family is no longer residing in the unit.

13-IV.G. RECORD KEEPING

For more information concerning general record keeping, see Chapter 16.

PHA Policy

A written record of every termination and/or eviction will be maintained by the PHA at the development where the family was residing, and will contain the following information:

Name of resident, number and identification of unit occupied

Date of the notice of lease termination and any other notices required by state or local law; these notices may be on the same form and will run concurrently

Specific reason(s) for the notices, citing the lease section or provision that was violated, and other facts pertinent to the issuing of the notices described in detail (other than any criminal history reports obtained solely through the authorization provided in 24 CFR 5.903 and 5.905)

Date and method of notifying the resident

Summaries of any conferences held with the resident including dates, names of conference participants, and conclusions

Chapter 14

GRIEVANCES AND APPEALS

INTRODUCTION

This chapter discusses grievances and appeals pertaining to PHA actions or failures to act that adversely affect public housing applicants or residents. The policies are discussed in the following three parts:

<u>Part I: Informal Hearings for Public Housing Applicants</u>. This part outlines the requirements and procedures for informal hearings for public housing applicants.

<u>Part II: Informal Hearings with Regard to Noncitizens</u>. This part discusses informal hearings regarding citizenship status and where they differ from the requirements for general applicant and tenant grievances.

<u>Part III: Grievance Procedures for Public Housing Residents</u>. This part outlines the requirements and procedures for handling grievances for public housing residents.

Note that this chapter is not the PHA's grievance procedure. The grievance procedure is a document separate from the ACOP. This chapter of the ACOP provides the policies that drive the grievance procedure. A sample grievance procedure is provided as Exhibit 14-1. However, please note that the procedure provided is only a sample and is designed to match up with the default policies in the model ACOP. As such, the PHA would need to modify accordingly should any alternative policy decisions be adopted.

PART I: INFORMAL HEARINGS FOR PUBLIC HOUSING APPLICANTS

14-I.A. OVERVIEW

When the PHA makes a decision that has a negative impact on an applicant family, the family is often entitled to appeal the decision. For applicants, the appeal takes the form of an informal hearing. HUD regulations do not provide a structure for or requirements regarding informal hearings for applicants (except with regard to citizenship status, to be covered in Part II). This part discusses the PHA policies necessary to respond to applicant appeals through the informal hearing process.

14-I.B. INFORMAL HEARING PROCESS [24 CFR 960.208(a) and PH Occ GB, p. 58]

Informal hearings are provided for public housing applicants. An applicant is someone who has applied for admission to the public housing program, but is not yet a tenant in the program. Informal hearings are intended to provide a means for an applicant to dispute a determination of ineligibility for admission to a project [24 CFR 960.208(a)]. Applicants to public housing are not entitled to the same hearing process afforded tenants under the PHA grievance procedures [24 CFR 966.53(a) and PH Occ GB, p. 58].

Informal hearings provide applicants the opportunity to review the reasons for denial of admission and to present evidence to refute the grounds for denial.

Use of Informal Hearing Process

While the PHA must offer the opportunity of an informal hearing to applicants who have been determined as ineligible for admission, the PHA could make the informal hearing process available to applicants who wish to dispute other PHA actions that adversely affect them.

PHA Policy

The PHA will only offer informal hearings to applicants for the purpose of disputing denials of admission.

Notice of Denial [24 CFR 960.208(a)]

The PHA must give an applicant prompt notice of a decision denying eligibility for admission. The notice must contain a brief statement of the reasons for the PHA decision, and must also state that the applicant may request an informal hearing to dispute the decision. The notice must describe how to obtain the informal hearing.

When denying eligibility for admission, the PHA must provide the family a notice of VAWA rights (form HUD-5380) as well as the HUD VAWA self-certification form (form HUD-5382) in accordance with the Violence against Women Reauthorization Act of 2013, and as outlined in 16-VII.C. The notice and self-certification form must accompany the written notification of the denial of eligibility determination.

Prior to notification of denial based on information obtained from criminal or sex offender registration records, the family, in some cases, must be given the opportunity to dispute the information in those records which would be the basis of the denial. See Section 3-III.G for details concerning this requirement.

PART III: GRIEVANCE PROCEDURES FOR PUBLIC HOUSING RESIDENTS

14-III.A. REQUIREMENTS [24 CFR 966.52]

PHAs must have a grievance procedure in place through which residents of public housing are provided an opportunity to grieve any PHA action or failure to act involving the lease or PHA policies which adversely affect their rights, duties, welfare, or status. The PHA must not only meet the minimal procedural due process requirements provided under the regulations, but must also meet any additional requirements imposed by local, state or federal law.

The PHA grievance procedure must be included in, or incorporated by reference in, the lease.

PHA Policy

The PHA grievance procedure will be incorporated by reference in the tenant lease.

The PHA must provide at least 30 days notice to tenants and resident organizations setting forth proposed changes in the PHA grievance procedure, and provide an opportunity to present written comments. Comments submitted must be considered by the PHA before adoption of any changes to the grievance procedure by the PHA.

PHA Policy

Residents and resident organizations will have 30 calendar days from the date they are notified by the PHA of any proposed changes in the PHA grievance procedure, to submit written comments to the PHA.

The PHA must furnish a copy of the grievance procedure to each tenant and to resident organizations.

14-III.B. DEFINITIONS [24 CFR 966.53; 24 CFR 966.51(a)(2)(i)]

There are several terms used by HUD with regard to public housing grievance procedures, which take on specific meanings different from their common usage. These terms are as follows:

- **Grievance** any dispute which a tenant may have with respect to PHA action or failure to act in accordance with the individual tenant's lease or PHA regulations which adversely affect the individual tenant's rights, duties, welfare or status
- **Complainant** any tenant whose grievance is presented to the PHA or at the project management office
- **Due Process Determination** a determination by HUD that law of the jurisdiction requires that the tenant must be given the opportunity for a hearing in court which provides the basic elements of due process before eviction from the dwelling unit
- **Expedited Grievance** a procedure established by the PHA for any grievance or termination that involves:
 - Any criminal activity that threatens the health, safety, or right to peaceful enjoyment or the PHA's public housing premises by other residents or employees of the PHA; or
 - Any drug-related criminal activity on or off the premises
- **Elements of Due Process** an eviction action or a termination of tenancy in a state or local court in which the following procedural safeguards are required:
 - Adequate notice to the tenant of the grounds for terminating the tenancy and for eviction
 - Right of the tenant to be represented by counsel
 - Opportunity for the tenant to refute the evidence presented by the PHA including the right to confront and cross-examine witnesses and to present any affirmative legal or equitable defense which the tenant may have
 - A decision on the merits
- **Hearing Officer** an impartial person or selected by the PHA, other than the person who made or approved the decision under review, or a subordinate of that person. The individual or individuals do not need legal training.
- **Tenant** the adult person (or persons) (other than a live-in aide)
 - Who resides in the unit, and who executed the lease with the PHA as lessee of the dwelling unit, or, if no such person now resides in the unit,
 - Who resides in the unit, and who is the remaining head of household of the tenant family residing in the dwelling unit
- **Resident Organization** includes a resident management corporation

14-III.C. APPLICABILITY [24 CFR 966.51]

Grievances could potentially address most aspects of a PHA's operation. However, there are some situations for which the grievance procedure is not applicable.

The grievance procedure is applicable only to individual tenant issues relating to the PHA. It is not applicable to disputes between tenants not involving the PHA. Class grievances are not subject to the grievance procedure and the grievance procedure is not to be used as a forum for initiating or negotiating policy changes of the PHA.

If HUD has issued a due process determination, a PHA may exclude from the PHA grievance procedure any grievance concerning a termination of tenancy or eviction that involves:

- Any criminal activity that threatens the health, safety or right to peaceful enjoyment of the premises of other residents or employees of the PHA;
- Any violent or drug-related criminal activity on or off such premises; or
- Any criminal activity that resulted in felony conviction of a household member

In states without due process determinations, PHAs must grant opportunity for grievance hearings for all lease terminations, regardless of cause, with the following exception: PHAs may use expedited grievance procedures for the excluded categories listed above. These expedited grievance procedures are described in Section 14-III.E. below.

If HUD has issued a due process determination, the PHA may evict through the state/local judicial eviction procedures. In this case, the PHA is not required to provide the opportunity for a hearing under the PHA's grievance procedure as described above.

PHA Policy

The PHA is located in a HUD-declared due process state. Therefore, the PHA will not offer grievance hearings for lease terminations involving criminal activity that threatens the health, safety, or right to peaceful enjoyment of the premises of other residents or employees of the PHA, for violent or drug-related criminal activity on or off the premises, or for any criminal activity that resulted in felony conviction of a household member.

See Chapter 13 for related policies on the content of termination notices.

14-III.D. INFORMAL SETTLEMENT OF GRIEVANCE [24 CFR 966.54]

HUD regulations state that any grievance must be personally presented, either orally or in writing, to the PHA office or to the office of the housing development in which the complainant resides so that the grievance may be discussed informally and settled without a hearing.

PHA Policy

The PHA will accept requests for an informal settlement of a grievance either orally or in writing, to the PHA office within 10 business days of the grievable event. Within 10 business days of receipt of the request the PHA will arrange a meeting with the tenant at a mutually agreeable time and confirm such meeting in writing to the tenant.

If a tenant fails to attend the scheduled meeting without prior notice, the PHA will reschedule the appointment only if the tenant can show good cause for failing to appear, or if it is needed as a reasonable accommodation for a person with disabilities.

Good cause is defined as an unavoidable conflict which seriously affects the health, safety or welfare of the family.

HUD regulations require that a summary of such discussion will be prepared within a reasonable time and one copy will be given to the tenant and one retained in the PHA's tenant file.

The summary must specify the names of the participants, dates of meeting, the nature of the proposed disposition of the complaint and the specific reasons therefore, and will specify the procedures by which a hearing may be obtained if the complainant is not satisfied.

PHA Policy

The PHA will prepare a summary of the informal settlement within five business days; one copy to be given to the tenant and one copy to be retained in the PHA's tenant file.

For PHAs who have the option to establish an expedited grievance procedure, and who exercise this option, the informal settlement of grievances is not applicable to those grievances for which the expedited grievance procedure applies.

14-III.E. PROCEDURES TO OBTAIN A HEARING

Requests for Hearing and Failure to Request

PHA Policy

The resident must submit a written request for a grievance hearing to the PHA within five business days of the tenant's receipt of the summary of the informal settlement.

If the complainant does not request a hearing, the PHA's disposition of the grievance under the informal settlement process will become final. However, failure to request a hearing does not constitute a waiver by the complainant of the right to contest the PHA's action in disposing of the complaint in an appropriate judicial proceeding.

Scheduling of Hearings [24 CFR 966.56(a)]

If the complainant has complied with all requirements for requesting a hearing as described above, a hearing must be scheduled by the hearing officer promptly for a time and place reasonably convenient to both the complainant and the PHA. A written notification specifying the time, place and the procedures governing the hearing must be delivered to the complainant and the appropriate PHA official.

PHA Policy

Within 10 business days of receiving a written request for a hearing, the hearing officer will schedule and send written notice of the hearing to both the complainant and the PHA.

The PHA may wish to permit the tenant to request to reschedule a hearing for good cause.

PHA Policy

The tenant may request to reschedule a hearing for good cause, or if it is needed as a reasonable accommodation for a person with disabilities. Good cause is defined as an unavoidable conflict which seriously affects the health, safety, or welfare of the family. Requests to reschedule a hearing must be made orally or in writing prior to the hearing date. At its discretion, the PHA may request documentation of the "good cause" prior to rescheduling the hearing.

Expedited Grievance Procedure [24 CFR 966.52(a)]

The PHA may establish an expedited grievance procedure for any grievance concerning a termination of tenancy or eviction that involves:

- Any criminal activity that threatens the health, safety, or right to peaceful enjoyment of the premises by other residents or employees of the PHA;
- Any drug-related criminal activity on or near such premises; or
- Any criminal activity that resulted in felony conviction of a household member.

In such expedited grievances, the informal settlement of grievances as discussed in 14-III.D is not applicable.

The PHA may adopt special procedures concerning expedited hearings, including provisions for expedited notice or scheduling, or provisions for expedited decision on the grievance.

PHA Policy

The PHA will not offer expedited grievance procedures.

14-III.F. SELECTION OF HEARING OFFICER [24 CFR 966.53(e)]

The grievance hearing must be conducted by an impartial person or persons appointed by the PHA, other than the person who made or approved the PHA action under review, or a subordinate of such person. The PHA must describe their policies for selection of a hearing officer in their lease.

PHA Policy

PHA grievance hearings will be conducted by a single hearing officer and not a panel.

The PHA will appoint a staff member who was not involved in the decision under appeal. If a designated staff member (such as the program manager) was involved in the decision, or is a subordinate of such person, an alternate hearing officer will be selected.

The PHA may select designated staff members who were not involved in the decision under appeal in certain circumstances, such as appeals involving discrimination claims or denials of requests for reasonable accommodations.

PHAs must describe their policies for selection of a hearing officer in their lease forms. Changes to the public housing lease are subject to a 30-day comment period [24 CFR 966.4].

14-III.G. PROCEDURES GOVERNING THE HEARING [24 CFR 966.56]

Rights of Complainant [24 CFR 966.56(b)]

The complainant will be afforded a fair hearing. This includes:

• The opportunity to examine before the grievance hearing any PHA documents, including records and regulations that are directly relevant to the hearing. The tenant must be allowed to copy any such document at the tenant's expense. If the PHA does not make the document available for examination upon request by the complainant, the PHA may not rely on such document at the grievance hearing.

PHA Policy

The tenant will be allowed to copy any documents related to the hearing at a cost of \$.25 per page. The family must request discovery of PHA documents no later than 12:00 p.m. on the business day prior to the hearing.

• The right to be represented by counsel or other person chosen to represent the tenant, and to have such person make statements on the tenant's behalf.

PHA Policy

Hearings may be attended by the following applicable persons:

A PHA representative(s) and any witnesses for the PHA

The tenant and any witnesses for the tenant

The tenant's counsel or other representative

Any other person approved by the PHA as a reasonable accommodation for a person with a disability

- The right to a private hearing unless the complainant requests a public hearing.
- The right to present evidence and arguments in support of the tenant's complaint, to controvert evidence relied on by the PHA or project management, and to confront and cross-examine all witnesses upon whose testimony or information the PHA or project management relies.
- A decision based solely and exclusively upon the facts presented at the hearing.

Failure to Appear [24 CFR 966.56(c)]

If the complainant or the PHA fails to appear at a scheduled hearing, the hearing officer may make a determination to postpone the hearing for no more than five business days or may make a determination that the party has waived his/her right to a hearing. Both the complainant and the PHA must be notified of the determination by the hearing officer: Provided, That a determination that the complainant has waived his/her right to a hearing will not constitute a waiver of any right the complainant may have to contest the PHA's disposition of the grievance in an appropriate judicial proceeding.

There may be times when a complainant does not appear due to unforeseen circumstances which are out of their control and are no fault of their own.

PHA Policy

If the tenant does not appear at the scheduled time of the hearing, the hearing officer will wait up to 30 minutes. If the tenant appears within 30 minutes of the scheduled time, the hearing will be held. If the tenant does not arrive within 30 minutes of the scheduled time, they will be considered to have failed to appear.

If the tenant fails to appear and was unable to reschedule the hearing in advance, the tenant must contact the PHA within 24 hours of the scheduled hearing date, excluding weekends and holidays. The hearing officer will reschedule the hearing only if the tenant can show good cause for the failure to appear, or it is needed as a reasonable accommodation for a person with disabilities.

"Good cause" is defined as an unavoidable conflict which seriously affects the health, safety, or welfare of the family.

General Procedures [24 CFR 966.56(d), (e)]

At the hearing, the complainant must first make a showing of an entitlement to the relief sought and thereafter the PHA must sustain the burden of justifying the PHA action or failure to act against which the complaint is directed [24 CFR 966.56(d)].

The hearing is conducted informally by the hearing officer. The PHA and the tenant must be given the opportunity to present oral or documentary evidence pertinent to the facts and issues raised by the complaint, and to question any witnesses.

PHA Policy

Any evidence to be considered by the hearing officer must be presented at the time of the hearing. There are four categories of evidence.

Oral evidence: the testimony of witnesses

Documentary evidence: a writing which is relevant to the case, for example, a letter written to the PHA. Writings include all forms of recorded communication or representation, including letters, emails, words, pictures, sounds, videotapes or symbols or combinations thereof.

Demonstrative evidence: Evidence created specifically for the hearing and presented as an illustrative aid to assist the hearing officer, such as a model, a chart or other diagram.

Real evidence: A tangible item relating directly to the case.

Hearsay Evidence is evidence of a statement that was made other than by a witness while testifying at the hearing and that is offered to prove the truth of the matter. Even though evidence, including hearsay, is generally admissible, hearsay evidence alone cannot be used as the sole basis for the hearing officer's decision.

If the PHA fails to comply with the discovery requirements (providing the tenant with the opportunity to examine PHA documents prior to the grievance hearing), the hearing officer will refuse to admit such evidence.

Other than the failure of the PHA to comply with discovery requirements, the hearing officer has the authority to overrule any objections to evidence.

The complainant or the PHA may arrange, in advance and at the expense of the party making the arrangement, for a transcript of the hearing. Any interested party may purchase a copy of such transcript [24 CFR 966.56(e)].

PHA Policy

If the complainant would like the PHA to record the proceedings by audiotape, the request must be made to the PHA by 12:00 p.m. on the business day prior to the hearing.

The PHA will consider that an audio tape recording of the proceedings is a transcript.

Accommodations of Persons with Disabilities [24 CFR 966.56(f)]

The PHA must provide reasonable accommodation for persons with disabilities to participate in the hearing. Reasonable accommodation may include qualified sign language interpreters, readers, accessible locations, or attendants.

If the tenant is visually impaired, any notice to the tenant which is required in the grievance process must be in an accessible format.

See Chapter 2 for a thorough discussion of the PHA's responsibilities pertaining to reasonable accommodation.

Limited English Proficiency (24 CFR 966.56(g)

The PHA must comply with HUD's LEP Final Rule in providing language services throughout the grievance process.

14-III.H. DECISION OF THE HEARING OFFICER [24 CFR 966.57]

The hearing officer must issue a written decision, stating the reasons for the decision, within a reasonable time after the hearing. Factual determinations relating to the individual circumstances of the family must be based on a preponderance of evidence presented at the hearing. A copy of the decision must be sent to the complainant and the PHA. The PHA must retain a copy of the decision in the tenant's folder. A log of all hearing officer decisions must also be maintained by the PHA and made available for inspection by a prospective complainant, his/her representative, or the hearing officer [24 CFR 966.57(a)].

PHA Policy

In rendering a decision, the hearing officer will consider the following matters:

PHA Notice to the Family: The hearing officer will determine if the reasons for the PHA's decision are factually stated in the notice.

Discovery: The hearing officer will determine if the family was given the opportunity to examine any relevant documents in accordance with PHA policy.

PHA Evidence to Support the PHA Decision: The evidence consists of the facts presented. Evidence is not conclusion and it is not argument. The hearing officer will evaluate the facts to determine if they support the PHA's conclusion.

Validity of Grounds for Termination of Tenancy (when applicable): The hearing officer will determine if the termination of tenancy is for one of the grounds specified in the HUD regulations and PHA policies. If the grounds for termination are not specified in the regulations or in compliance with PHA policies, then the decision of the PHA will be overturned.

The hearing officer will issue a written decision to the family and the PHA no later than 10 business days after the hearing. The report will contain the following information:

Hearing information:

Name of the complainant

Date, time and place of the hearing

Name of the hearing officer

Name of the PHA representative(s)

Name of family representative (if any)

Names of witnesses (if any)

Background: A brief, impartial statement of the reason for the hearing and the date(s) on which the informal settlement was held, who held it, and a summary of the results of the informal settlement. Also includes the date the complainant requested the grievance hearing.

Summary of the Evidence: The hearing officer will summarize the testimony of each witness and identify any documents that a witness produced in support of his/her testimony and that are admitted into evidence.

Findings of Fact: The hearing officer will include all findings of fact, based on a preponderance of the evidence. *Preponderance of the evidence* is defined as evidence which is of greater weight or more convincing than the evidence which is offered in opposition to it; that is, evidence which as a whole shows that the fact sought to be proved is more probable than not. Preponderance of the evidence may not be determined by the number of witnesses, but by the greater weight of all evidence.

Conclusions: The hearing officer will render a conclusion derived from the facts that were found to be true by a preponderance of the evidence. The conclusion will result in a determination of whether these facts uphold the PHA's decision.

Order: The hearing report will include a statement of whether the PHA's decision is upheld or overturned. If it is overturned, the hearing officer will instruct the PHA to change the decision in accordance with the hearing officer's determination. In the case of termination of tenancy, the hearing officer will instruct the PHA to restore the family's status.

Procedures for Further Hearing

PHA Policy

The hearing officer may ask the family for additional information and/or might adjourn the hearing in order to reconvene at a later date, before reaching a decision. If the family misses an appointment or deadline ordered by the hearing officer, the action of the PHA will take effect and another hearing will not be granted.

Final Decision [24 CFR 966.57(b)]

The decision of the hearing officer is binding on the PHA which must take the action, or refrain from taking the action cited in the decision unless the PHA Board of Commissioners determines within a reasonable time, and notifies the complainant that:

- The grievance does not concern PHA action or failure to act in accordance with or involving the complainant's lease on PHA policies which adversely affect the complainant's rights, duties, welfare, or status; or
- The decision of the hearing officer is contrary to federal, state, or local law, HUD regulations or requirements of the annual contributions contract between HUD and the PHA

PHA Policy

When the PHA considers the decision of the hearing officer to be invalid due to the reasons stated above, it will present the matter to the PHA Board of Commissioners within 10 business days of the date of the hearing officer's decision. The Board has 30 calendar days to consider the decision. If the Board decides to reverse the hearing officer's decision, it must notify the complainant within 10 business days of this decision.

A decision by the hearing officer or Board of Commissioners in favor of the PHA or which denies the relief requested by the complainant in whole or in part must not constitute a waiver of any rights, nor effect in any manner whatever, any rights the complainant may have to a subsequent trial or judicial review in court [24 CFR 966.57(c)].

EXHIBIT 14-1: SAMPLE GRIEVANCE PROCEDURE

Note: The sample procedure provided below is a sample only and is designed to match up with the default policies in the model ACOP. If your PHA has made policy decisions that do not reflect the default policies in the ACOP, you would need to ensure that the procedure matches those policy decisions.

I. Definitions applicable to the grievance procedure [24 CFR 966.53]

- A. Grievance: Any dispute a tenant may have with respect to PHA action or failure to act in accordance with the individual tenant's lease or PHA regulations that adversely affects the individual tenant's rights, duties, welfare, or status.
- B. Complainant: Any tenant (as defined below) whose grievance is presented to the PHA or at the project management office in accordance with the requirements presented in this procedure.
- C. Elements of due process: An eviction action or a termination of tenancy in a state or local court in which the following procedural safeguards are required:
 - i. Adequate notice to the tenant of the grounds for terminating the tenancy and for eviction
 - ii. Right of the tenant to be represented by counsel
 - iii. Opportunity for the tenant to refute the evidence presented by the PHA, including the right to confront and cross-examine witnesses and to present any affirmative legal or equitable defense that the tenant may have
 - iv. A decision on the merits of the case
- D. Hearing officer: An impartial person or persons selected by the PHA other than the person who made or approved the decision under review, or a subordinate of that person. Such individuals do not need legal training.
- E. Tenant: The adult person (or persons other than a live-in aide) who resides in the unit and who executed the lease with the PHA as lessee of the dwelling unit, or if no such person now resides in the unit, who resides in the unit and who is the remaining head of the household of the tenant family residing in the dwelling unit.
- F. Resident organization: An organization of residents, which also may include a resident management corporation.

II. Applicability of this grievance procedure [24 CFR 966.51]

In accordance with the applicable federal regulations (24 CFR 966.50), this grievance procedure is applicable to all individual grievances (as defined in Section I above) between the tenant and the PHA with the following exception of disputes between tenants not involving the PHA or class grievances. The grievance procedure is not intended as a forum for initiating or negotiating policy changes between a group or groups of tenants and the PHA's Board of Commissioners [24 CFR 966.51(b)].

This grievance procedure is incorporated by reference in all tenant dwelling leases and will be furnished to each tenant and all resident organizations [24 CFR 966.52 (b) and (d)].

Any changes proposed in this grievance procedure must provide for at least 30 days' notice to tenants and resident organizations, setting forth the proposed changes and providing an opportunity to present written comments. Comments will be considered by the PHA before any revisions are made to the grievance procedure [24 CFR 966.52(c)].

III. Informal settlement of a grievance [24 CFR 966.54]

Any grievance must be personally presented, either orally or in writing, to the PHA's central office or the management office of the development in which the complainant resides within 10 days after the grievable event.

Grievances related to complaints about operations matters that are received by the PHA's central office will be referred to the person responsible for the management of the development in which the complainant resides. Grievances involving complaints related to discrimination, harassment, or disability rights will be referred to the Civil Rights Administrator or Director of Operations.

As soon as the grievance is received, it will be reviewed by the management office of the development or the Civil Rights Administrator (if applicable) to be certain that neither of the exclusions in paragraphs II.A or II.B above applies to the grievance. Should one of the exclusions apply, the complainant will be notified in writing that the matter raised is not subject to the PHA's grievance procedure with the reason specified.

If neither of the exclusions cited above apply, the complainant will be contacted to arrange a mutually convenient time **within 10 business days** to meet so the grievance may be discussed informally and settled without a hearing. At the informal settlement, the complainant will present the grievance and the person in charge of the management office or the Civil Rights Administrator will attempt to settles the grievance to the satisfaction of both parties.

Within five business days following the informal discussion, the PHA will prepare and either hand deliver, mail, or email to the tenant a summary of the discussion that must specify the names of the participants, the dates of meeting, the nature of the proposed disposition of the complaint, and the specific reasons therefore, and will specify the procedures by which a formal hearing under this procedure may be obtained if the complainant is not satisfied. A copy of this summary will also be placed in the tenant's file.

IV. Formal grievance hearing

If the complainant is not satisfied with the settlement arrived at in the informal settlement, the complainant must submit a written request for a hearing to the management office of the development where the tenant resides **no later than five business days after the summary of the informal hearing is received.**

The written request must specify:

- The reasons for the grievance; and
- The action of relief sought from the PHA

Within 10 days of receiving the written request for a hearing, the hearing officer will schedule and sent written notice of hearing to both the complainant and the PHA.

V. Selecting the hearing officer

A grievance hearing will be conducted by a single impartial person appointed by the PHA as described below:

- A. The hearing officer will be appointed directly by the executive director.
- B. The hearing officer will be a staff member who did not make or approve the decision under review and who is not a subordinate of such persons. If the designated staff member (such as the program manager) was involved in the decision or is a subordinate of such person, an alternate hearing officer will be selected.
- C. The PHA may select designated staff members who were not involved in the decision under review in certain circumstances, such as those involving discrimination claims or denials of requests for reasonable accommodations.
- D. The PHA's method for selecting a hearing officer will be inserted into the lease.

VI. Scheduling hearings [24 CFR 966.56(a)]

When a complainant submits a timely request for a grievance hearing, the PHA will immediately appoint an impartial hearing office to schedule the hearing within the following 10 business days.

Once the hearing officer has scheduled the hearing, the hearing officer will send written notice of the hearing to both the complainant and the PHA. Notice to the complainant will be in writing, either personally delivered to the complainant, or sent by mail or email, return receipt requested.

The written notice will specify the time, place, and procedures governing the hearing.

The tenant may request to reschedule a hearing on a one-time basis. Should the complainant need to reschedule a second time, he or she may do so for good cause, or if needed as a reasonable accommodation for a person with disabilities. *Good cause* is defined as an unavoidable conflict which seriously affects the health, safety, or welfare of the family. Requests to reschedule a hearing must be made orally or in writing prior to the hearing date.

VII. Procedures governing the hearing [24 CFR 966.56]

The hearing will be held before a hearing officer as described above in Section V. The complainant will be afforded a fair hearing, which will include:

- A. The opportunity to examine before the hearing any PHA documents, including records and regulations, that are directly relevant to the hearing.
 - The tenant is allowed to copy any such document at the tenant's expense. If the PHA does not make the document available for examination upon request by the complainant, the PHA may no rely on such document at the grievance hearing.
- B. The right to be represented by counsel or other person chosen as the tenant's representative and to have such person make statements on the tenant's behalf.
- C. The right to a private hearing unless the complainant requests a public hearing.
- D. The right to present evidence and arguments in support of the tenant's complaint, to refute evidence relied on by the PHA or project management, and to confront and cross-examine all witnesses upon whose testimony or information the PHA or project management relies.
- E. A decision based solely and exclusively upon the fact presented at the hearing [24 CFR 966.56(b)].

The hearing is conducted informally by the hearing officer. The PHA and the tenant must be given the opportunity to present oral or documentary evidence pertinent to the facts and issues raised by the complaint, and to question any witnesses.

The complainant or the PHA may arrange in advance for a transcript of the hearing at the expense of the party making the arrangement. Any interested party may purchase a copy of the transcript [24 CFR 966.56(e)].

The PHA must provide reasonable accommodation for persons with disabilities to participated in the hearing. Reasonable accommodation may include qualified sign language interpreters, readers, accessible locations, or attendants. If the tenant is visually impaired, any notice to the tenant that is required under this procedure must be in an accessible format [24 CFR 966.56(f)].

The PHA must comply with HUD's requirements regarding limited English proficiency as specified in "Final Guidance to Federal Financial Assistance Recipients Regarding Title VI Prohibition against National Origin Discrimination Affecting Limited English Proficient Persons," issued January 22, 2007, and available at:

http://portal.hud.gov/hudportal/HUD?src=/program_offices/fair_housing_equal_opp/promotingfh/lep-faq.

VIII. Failure to appear at the hearing

If the complainant or PHA fails to appear at the hearing, the hearing officer may make a determination to postpone the hearing or make a determination that the complainant has waived his or her right to a hearing.

Both the complainant and the PHA must be notified of the determination by the hearing officer. A determination that the complainant has waived his or her right to a hearing will not constitute a waiver of any right the complainant may have to contest the PHA's disposition of the grievance in an appropriate judicial setting [24 CFR 966.56(c)].

IX. Decision of the hearing officer [24 CFR 966.57]

The hearing officer will prepare a written decision together with the reasons for the decision **within 10 business days** after the hearing. A copy of the decision will be sent to the complainant and the PHA.

The PHA will retain a copy of the decision in the tenant's file.

The hearing officer may ask the family for additional information and/or might adjourn the hearing in order to reconvene at a later date before reaching a decision. If the family misses a deadline ordered by the hearing officer, the hearing officer will make a decision based on the evidence presented.

The decision of the hearing officer will be binding on the PHA unless the PHA's Board of Commissioners determines within a reasonable time and notifies the complainant of its determination that:

- A. The grievance does not concern PHA action or failure to act in accordance with or involving the complainant's lease or PHA regulations, which adversely affect the complainant's rights, duties, welfare, or status; or
- B. The decision of the hearing officer is contrary to applicable federal, state, or local law, HUD regulations, or requirements of the annual contributions contract (ACC) between HUD and the PHA.

When the PHA considers the decision of the hearing officer to be invalid due to either of the reasons stated above, it will present the matter to the PHA Board of Commissioners within 10 business days of the date of the hearing officer's decision. The Board has 30 calendar days to consider the decision. If the Board decides to reverse the hearing officer's decision, it must notify the complainant within 10 business days of this decision.

A decision by the hearing officer or Board of Commissioners in favor of the PHA or which denies the relief requested by the complainant in whole or in part will not constitute a waiver of nor affect in any way the rights of the complainant to a trial or judicial review in any court proceedings, which may be brought in the matter later [24 CFR 966.57].

Prohibited Actions

An applicant or resident in the public housing program must not knowingly:

- Make a false statement to the PHA [Title 18 U.S.C. Section 1001].
- Provide incomplete or false information to the PHA [24 CFR 960.259(a)(4)].
- Commit fraud, or make false statements in connection with an application for assistance or with reexamination of income [24 CFR 966.4(1)(2)(iii)(C)].

PHA Policy

Any of the following will be considered evidence of family program abuse:

Offering bribes or illegal gratuities to the PHA Board of Commissioners, employees, contractors, or other PHA representatives

Offering payments or other incentives to a third party as an inducement for the third party to make false or misleading statements to the PHA on the family's behalf

Use of a false name or the use of falsified, forged, or altered documents

Intentional misreporting of family information or circumstances (e.g., misreporting of income or family composition)

Omitted facts that were obviously known by a family member (e.g., not reporting employment income)

Admission of program abuse by an adult family member

The PHA may determine other actions to be program abuse based upon a preponderance of the evidence, as defined earlier in this chapter.

Penalties for Program Abuse

In the case of program abuse caused by a family the PHA may, at its discretion, impose any of the following remedies.

- The PHA may require the family to repay any amounts owed to the program (see 15-II.B., Family Reimbursement to PHA).
- The PHA may require, as a condition of receiving or continuing assistance, that a culpable family member not reside in the unit. See policies in Chapter 3 (for applicants) and Chapter 13 (for residents).
- The PHA may deny admission or terminate the family's lease following the policies set forth in Chapter 3 and Chapter 13 respectively.
- The PHA may refer the family for state or federal criminal prosecution as described in section 15-II.D.

15-II.C. PHA-CAUSED ERRORS OR PROGRAM ABUSE

The responsibilities and expectations of PHA staff with respect to normal program administration are discussed throughout the ACOP. This section specifically addresses actions of a PHA staff member that are considered errors or program abuse related to the public housing program. Additional standards of conduct may be provided in the PHA personnel policy.

PHA-caused incorrect rental determinations include (1) failing to correctly apply public housing rules regarding family composition, income, assets, and expenses, and (2) errors in calculation.

Repayment to the PHA

The family is not required to repay an underpayment of rent if the error or program abuse is caused by PHA staff.

PHA Reimbursement to Family

PHA Policy

The PHA will reimburse a family for any family overpayment of rent, regardless of whether the overpayment was the result of staff-caused error or staff program abuse.

Prohibited Activities

PHA Policy

Any of the following will be considered evidence of program abuse by PHA staff:

Failing to comply with any public housing program requirements for personal gain

Failing to comply with any public housing program requirements as a result of a conflict of interest relationship with any applicant or resident

Seeking or accepting anything of material value from applicants, residents, vendors, contractors, or other persons who provide services or materials to the PHA

Disclosing confidential or proprietary information to outside parties

Gaining profit as a r15esult of insider knowledge of PHA activities, policies, or practices

Misappropriating or misusing public housing funds

Destroying, concealing, removing, or inappropriately using any records related to the public housing program

Committing any other corrupt or criminal act in connection with any federal housing program

Committing sexual harassment or other harassment based on race, color, religion, national origin, familial status, disability, sexual orientation, or gender identity, either quid pro quo (supervisory harassment) or hostile environment

Allowing sexual harassment or other harassment based on race, color, religion, national origin, familial status, disability, sexual orientation, or gender identity, either quid pro quo (supervisory harassment) or hostile environment, where the PHA knew or should have known such harassment was occurring

Retaliating against any applicant, resident, or staff reporting sexual harassment or other harassment based on race, color, religion, national origin, familial status, disability, sexual orientation, or gender identity, either quid pro quo (supervisory harassment) or hostile environment

15-II.D. CRIMINAL PROSECUTION

PHA Policy

When the PHA determines that program abuse by a family or PHA staff member has occurred and the amount of underpaid rent meets or exceeds the threshold for prosecution under local or state law, the PHA will refer the matter to the appropriate entity for prosecution. When the amount of underpaid rent meets or exceeds the federal threshold, the case will also be referred to the HUD Office of Inspector General (OIG).

Other criminal violations related to the public housing program will be referred to the appropriate local, state, or federal entity.

15-II.E. FRAUD AND PROGRAM ABUSE RECOVERIES

PHAs who enter into a repayment agreement with a family to collect rent owed, initiate litigation against the family to recover rent owed, or begin eviction proceedings against a family may retain 100 percent of program funds that the PHA recovers [Notice PIH 2007-27 (HA)].

If the PHA does none of the above, all amounts that constitute an underpayment of rent must be returned to HUD.

The family must be afforded the opportunity for a hearing through the PHA's grievance process.

PART II: ESTABLISHING FLAT RENTS

16-II.A. OVERVIEW

Flat rents are designed to encourage self-sufficiency and to avoid creating disincentives for continued residency by families who are attempting to become economically self-sufficient.

Flat rents are also used to prorate assistance for a mixed family. A mixed family is one whose members include those with citizenship or eligible immigration status, and those without citizenship or eligible immigrations status [24 CFR 5.504].

This part discusses how the PHA establishes and updates flat rents. Policies related to the use of flat rents, family choice of rent, flat rent hardships, and proration of rent for a mixed family are discussed in Chapter 6.

16-II.B. FLAT RENTS [24 CFR 960.253(b) and Notice PIH 2017-23]

Establishing Flat Rents

The 2015 Appropriations Act requires that flat rents must be set at no less than 80 percent of the applicable fair market rent (FMR). Alternatively, the PHA may set flat rents at no less than 80 percent of the applicable small area FMR(SAFMR) for metropolitan areas, or 80 percent of the applicable unadjusted rents for nonmetropolitan areas.

For areas where HUD has not determined a SAFMR or an unadjusted rent, PHAs must set flat rents at no less than 80 percent of the FMR or apply for an exception flat rent.

The 2015 Appropriations Act permits PHAs to request an exception flat rent that is lower than either 80 percent of the FMR or SAFMR/unadjusted rent if the PHA can demonstrate that these FMRs do not reflect the market value of a particular property or unit.

In order to demonstrate the need for an exception flat rent, PHAs are required to submit a market analysis methodology that demonstrates the value of the unit. The PHA must use HUD's rent reasonableness methodology to determine flat rents. In determining flat rents, PHAs must consider the following:

- Location
- Quality
- Unit size
- Unit type
- Age of the unit
- Amenities at the property and in immediate neighborhood
- Housing services provided
- Maintenance provided by the PHA
- Utilities provided by the PHA and/or landlord for (comparable units in the market study)

The PHA must provide a corresponding key explaining the calculations used for determining the valuation for each factor. HUD published a Flat Rent Market Analysis tool on August 22, 2018, which includes a rent adjustment guide, a market rent comparison guide, and a rent adjustment worksheet to aide PHAs in requesting exception flat rents.

PHAs must receive written HUD approval before implementing exception flat rents. PHAs with a previously approved flat rent exception request may submit a written request to extend the approved flat rents for up to two additional years, provided local market conditions remain unchanged. Detailed information on how to request exception flat rents can be found in Notice PIH 2017-23.

PHAs are now required to apply a utility allowance to flat rents as necessary. Flat rents set at 80 percent of the FMR must be reduced by the amount of the unit's utility allowance, if any.

Review of Flat Rents

No later than 90 days after the effective date of the new annual FMRs/SAFMRs/unadjusted rent, PHAs must implement new flat rents as necessary based changes to the FMR/SAFMR/unadjusted rent or request an exception.

If the FMR falls from year to year, the PHA may, but is not required to, lower the flat rent to 80 percent of the current FMR/SAFMR/unadjusted rent.

PHA Policy

If the FMR/SAFMR/unadjusted rent is lower than the previous year, the PHA will reduce flat rents to 80 percent of the current FMR/SAFMR.

Posting of Flat Rents

PHA Policy

The PHA will publicly post the schedule of flat rents in a conspicuous manner in the applicable PHA or project office.

Documentation of Flat Rents [24 CFR 960.253(b)(5)]

The PHA must maintain records that document the method used to determine flat rents, and that show how flat rents were determined by the PHA in accordance with this method.